.....

SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM 10-Q

Quarterly Report Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

For the Quarterly Period Ended June 30, 1997

Commission File Number 1-12139

W. R. GRACE & CO.

Delaware 65-0654331

(State of Incorporation) (I.R.S. Employer Identification No.)

One Town Center Road Boca Raton, Florida 33486-1010 (561) 362-2000

Indicate by check mark whether the registrant (including its predecessor) (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months and (2) has been subject to such filing requirements for the past 90 days.

Yes X No

73,410,165 shares of Common Stock, \$.01 par value, were outstanding at August 1, 1997.

W. R. GRACE & CO. AND SUBSIDIARIES

Table of Contents

		Page No.
Part I.	Financial Information	
Item 1.	Financial Statements	
	Consolidated Statement of Operations	I - 1
	Consolidated Statement of Cash Flows	I - 2
	Consolidated Balance Sheet	I - 3
	Notes to Consolidated Financial Statements	I - 4 to I - 10
Item 2.	Management's Discussion and Analysis of Results of Operations and Financial Condition	I - 11 to I - 19
Part II.	Other Information	
Item 4. Item 5.	Legal Proceedings Submission of Matters to a Vote of Security Holders Other Information Exhibits and Reports on Form 8-K	II - 1 II - 1 II - 3 II - 3

As used in this Report, the term "Company" refers to W. R. Grace & Co., and the term "Grace" refers to the Company and/or one or more of its subsidiaries.

PART 1. FINANCIAL INFORMATION

ITEM 1. FINANCIAL STATEMENTS

W. R. GRACE & CO. AND SUBSIDIARIES Three CONSOLIDATED STATEMENT OF OPERATIONS (UNAUDITED)		e 30,	Six Month June	30,
In millions (except per share amounts)	1997	1996	1997	1996
Sales and revenues	\$ 842.5	\$ 920.3	\$1,627.6	\$1,781.9
Other income	18.9	14.3	23.6	18.1
TOTAL	861.4	934.6	1,651.2	1,800.0
Cost of goods sold and operating expenses	513.2	549.2	993.2	1,061.8
Selling, general and administrative expenses	158.0	201.0	304.6	394.9
Depreciation and amortization	49.2	45.9	97.7	90.8
Interest expense and related financing costs	20.3	18.3	39.4	36.7
Research and development expenses	21.7	26.2	43.6	52.5
Restructuring costs	12.4	53.7		53.7
Gain on sales of businesses	(103.1)	(326.4)		(326.4)
TOTAL	671.7	567.9	1,387.8	1,364.0
INCOME FROM CONTINUING OPERATIONS BEFORE	400 7	000 7	000 4	400.0
INCOME TAXES	189.7		263.4	436.0
Provision for income taxes	72.3	131.7	99.6	157.4
INCOME FROM CONTINUING OPERATIONS	117.4	235.0	163.8	278.6
Income from discontinued operations		98.9		118.9
NET INCOME	\$ 117.4	\$ 333.9	\$ 163.8	\$ 397.5
	======	======	======	======
Earnings per share: Continuing operations	\$ 1.61	\$ 2.43	\$ 2.21	\$ 2.86
Net income	\$ 1.61			
Fully diluted earnings per share:				
Continuing operations	\$ 1.56			
Net income	\$ 1.56	\$ 3.39	\$ 2.15	\$ 4.00
Dividends declared per common share	\$.145	\$.125	\$.27	\$.25
Weighted average shares outstanding	72.8	96.6	74.0	97.3

The Notes to Consolidated Financial Statements are integral parts of these statements.

ONSOLIDATED STATEMENT OF CASH FLOWS (UNADDITED)		/	
n millions	1997	1996	
PERATING ACTIVITIES ncome from continuing operations before income taxes	\$263.4	\$ 436.0	
econciliation to cash provided by operating activities:	Ψ200.4	Ψ 430.0	
Depreciation and amortization	97.7	90.8	
Provision relating to restructuring costs	12.4	53.7	
Gain on sales of businesses	(103.1)	(326.4)	
currency exchange:	(72.2)	(70.4)	
Increase in notes and accounts receivable, net (Increase)/decrease in inventories	(72.3) (13.1)	(78.4) 33.3	
Proceeds from asbestos-related insurance settlements	42.5	99.7	
Payments made for asbestos-related litigation	42.5	99.1	
settlements, judgments and defense costs	(49.9)	(73.1)	
Decrease in accounts payable	(36.3)	(20.3)	
Other	(47.2)	(98.2)	
NET PRETAX CASH PROVIDED BY OPERATING ACTIVITIES			
OF CONTINUING OPERATIONS	94.1	117.1	
t pretax cash (used for)/provided by operating activities	0411	117.1	
of discontinued operations	(25.3)	41.9	
NET DRETAY CACH DROVIDED BY OBERATING ACTIVITIES		450.0	
NET PRETAX CASH PROVIDED BY OPERATING ACTIVITIES come taxes paid	68.8 (28.0)	159.0 (40.5)	
come taxes para	(20.0)	(40.3)	
NET CASH PROVIDED BY OPERATING ACTIVITIES	40.8	118.5	
VESTING ACTIVITIES			
pital expenditures	(113.6)	(234.5)	
t proceeds from divestmentssinesses acquired in purchase transactions, net of	666.3	697.1	
cash acquired and debt assumed	(15.9)		
t investing activities of discontinued operations	(70.7)	(97.6)	
her	14.9	14.7	
NET CASH PROVIDED BY INVESTING ACTIVITIES	481.0 	379.7	
NANCING ACTIVITIES			
vidends paidpayments of borrowings having original maturities in	(19.8)	(24.5)	
excess of three monthscrease in borrowings having original maturities in	(72.8)	(50.8)	
excess of three monthst repayments of borrowings having original maturities of	(2.9)		
three months or less	(137.0)	(6.7)	
ock options exercised	24.3	53.2	
t financing activities of discontinued operations		(37.1)	
rchase of treasury stock	(335.9)	(398.2)	
NET CASH USED FOR FINANCING ACTIVITIES	(544.1)	(464.1)	
fect of exchange rate changes on cash and cash equivalents	(2.8)		
(DECREASE)/INCREASE IN CASH AND CASH EQUIVALENTS	d (05.4)		
	\$ (25.1)	\$ 33.1	

The Notes to Consolidated Financial Statements are integral parts of these statements.

- ------

W. R. GRACE & CO. AND SUBSIDIARIES CONSOLIDATED BALANCE SHEET (UNAUDITED)

n millions (except par value)	1997	December 31, 1996
SSETS		
JRRENT ASSETS		
ash and cash equivalents	\$ 43.2	\$ 68.3
		831.4
otes and accounts receivable, net		
nventories		376.1
et assets of discontinued operations	56.8	297.4
eferred income taxes	121.5	183.9
ther current assets	26.8	17.8
TOTAL CURRENT ASSETS	1,259.4	1,774.9
	,	_,
roperties and equipment, net of accumulated depreciation an		
amortization of \$1,478.9 (December 31, 1996 - \$1,436.6) bodwill, less accumulated amortization of \$14.3	1,814.6	1,871.3
(December 31, 1996 - \$18.6)	38.1	40.6
sbestos-related insurance receivable		296.3
eferred income taxes		309.2
ther assets	614.3	653.5
TOTAL ASSETS	\$4,312.3 ======	\$4,945.8 ======
IABILITIES AND SHAREHOLDERS' EQUITY		
URRENT LIABILITIES		
nort-term debt	\$ 106.5	\$ 315.2
counts payable	218.8	274.7
ncome taxes	122.6	123.3
ther current liabilities	697.0	773.9
THE CUITERIC TEMPETERS	097.0	113.9
TOTAL CURRENT LIABILITIES		1,487.1
	,	, -
ong-term debt	1,060.9	1,073.0
eferred income taxes	•	43.5
oncurrent liability for asbestos-related litigation		859.1
ther liabilities		850.7
.ner itubitities		
TOTAL LIABILITIES		4,313.4
OMMITMENTS AND CONTINGENCIES		
HAREHOLDERS' EQUITY		
ommon stock issued, par value \$.01	.8	.8
aid in capital	556.7	524.1
etained earnings	316.6	172.6
umulative translation adjustments		
eferred compensation trust, at market: 70,250 common shares		(04.0)
reasury stock, at cost: 6,220,583 common shares	(3.9)	- -
(December 31, 1996 - 10,000)	(331.9)	(.5)
TOTAL SHAREHOLDERS' EQUITY		632.4
TOTAL LIABILITIES AND SHAREHOLDERS' EQUITY	Ф4 212 2	
INTAL LIABILITIES AND SHAKEHULDERS, FOULTY	\$4,312.3 ======	\$4,945.8 ======

The Notes to Consolidated Financial Statements are integral parts of these statements.

1. BASIS OF PRESENTATION

The interim consolidated financial statements in this Report are unaudited and should be read in conjunction with the consolidated financial statements in the Company's Annual Report on Form 10-K for the year ended December 31, 1996 (1996 Form 10-K). The interim consolidated financial statements reflect all adjustments that, in the opinion of management, are necessary for a fair presentation of the results of the interim periods presented; all such adjustments were of a normal recurring nature. Certain amounts in the prior periods' consolidated financial statements have been reclassified to conform to the current periods' basis of presentation and as required with respect to discontinued operations.

The results of operations for the three- and six-month interim periods ended June 30, 1997 are not necessarily indicative of the results of operations for the fiscal year ending December 31, 1997.

2. ASBESTOS AND RELATED INSURANCE LITIGATION

Grace is a defendant in property damage and personal injury lawsuits relating to previously sold asbestos-containing products and anticipates that it will be named as a defendant in additional asbestos-related lawsuits in the future. Grace was a defendant in approximately 43,300 asbestos-related lawsuits at June 30, 1997 (25 involving claims for property damage and the remainder involving approximately 104,500 claims for personal injury), compared to approximately 41,500 lawsuits at December 31, 1996 (31 involving claims for property damage and the remainder involving approximately 91,500 claims for personal injury).

Property Damage Litigation

Through June 30, 1997, 137 asbestos property damage cases were dismissed without payment of any damages or settlement amounts; judgments were entered in favor of Grace in nine cases (excluding cases settled following appeals of judgments in favor of Grace); judgments were entered in favor of the plaintiffs in seven cases (none of which is on appeal) for a total of \$60.3; and 190 property damage cases were settled for a total of \$454.0. Property damage case activity for the six months ended June 30, 1997 was as follows:

Cases outstanding, December 31, 1996	31
Settlements	(4)
Dismissals	(2)
Cases outstanding, June 30, 1997	25
	===

Personal Injury Litigation

Through June 30, 1997, approximately 12,500 asbestos personal injury lawsuits involving approximately 29,000 claims were dismissed without payment of any damages or settlement amounts (primarily on the basis that Grace products were not involved), and approximately 32,800 lawsuits involving approximately 69,700 claims were disposed of for a total of \$204.0. Personal injury claim activity for the six months ended June 30, 1997 was as follows:

Claims outstanding, December 31, 1996	91,511 14,859
Judgments	(3)
Claims outstanding, June 30, 1997	=======

Asbestos-Related Liability

Based upon and subject to the factors discussed in Note 2 to the consolidated financial statements in the 1996 Form 10-K, Grace estimates that its probable liability with respect to the defense and disposition of asbestos property damage and personal injury cases and claims was as follows at June 30, 1997 and December 31, 1996:

	June 30, 1997(1)	December 31, 1996(1)
Current liability for asbestos-related litigation(2) Noncurrent liability for asbestos-related litigation	\$135.0 812.7	\$135.0 859.1
Total asbestos-related liability(3)	\$947.7 =====	\$994.1 =====

- (1) Reflects property damage and personal injury cases and claims pending at June 30, 1997 and December 31, 1996, respectively, as well as personal injury claims expected to be filed through 2001.
- (2) Included in "Other current liabilities" in the Consolidated Balance Sheet.
- (3) Excludes one property damage case as to which liability is not yet estimable because Grace has not yet been able to obtain sufficient information through discovery proceedings.

Asbestos-Related Insurance Receivable

Grace previously purchased insurance policies with respect to its asbestos-related lawsuits and claims. The following table displays the activity in Grace's notes receivable from insurance carriers and asbestos-related insurance receivable during the six months ended June 30, 1907.

NOTES RECEIVABLE Notes receivable from insurance carriers at December 31, 1996, \$ 48.5 (17.4)Amortization, net..... 1.3 Notes receivable from insurance carriers at June 30, 1997, net of discount of \$6.1(2)..... \$ 32.4 INSURANCE RECEIVABLE Asbestos-related insurance receivable at December 31, 1996(3)..... \$ 331.3 Proceeds from asbestos-related insurance settlements..... (25.1)Asbestos-related insurance receivable at June 30, 1997(3)...... \$ 306.2 Total amounts due from insurance carriers..... \$ 338.6

- (1) Classified in the December 31, 1996 Consolidated Balance Sheet as \$17.2 in "Notes and accounts receivable, net" and \$31.3 in "Other assets."
- (2) Classified in the June 30, 1997 Consolidated Balance Sheet as \$15.0 in "Notes and accounts receivable, net" and \$17.4 in "Other assets."
- (3) \$35.0 of the asbestos-related insurance receivable is classified in "Notes and accounts receivable, net" in the December 31, 1996 and June 30, 1997 Consolidated Balance Sheets.

Insurance Litigation

Grace's ultimate exposure with respect to its asbestos-related cases and claims will depend on the extent to which its insurance will cover damages for which it may be held liable, amounts paid in settlement and litigation costs. In Grace's opinion, it is probable that recoveries from its insurance carriers (including amounts reflected in the receivable discussed above), along with other funds, will be available to satisfy the property damage and personal injury cases and claims pending at June 30, 1997, as well as personal injury claims expected to be filed in the foreseeable future. Consequently, Grace believes that the resolution of its asbestos-related litigation will not have a material adverse effect on its consolidated financial position.

For additional information, see Note 2 to the consolidated financial statements in the 1996 Form 10-K.

3. ACQUISITIONS AND DIVESTMENTS

Acquisitions

In April 1997, Grace purchased all of the shares of capital stock of Schurpack, Inc. (Schurpack), a manufacturer of flexible food packaging located in St. Joseph, Missouri. Schurpack, with 1996 sales of approximately \$20.0, is a leading manufacturer of plastic laminate packaging materials for the institutional and retail cook-in market segment. Schurpack also co-extrudes and converts film for food and non-food applications.

Divestments

In May 1997, Grace completed the sale of its specialty polymers business to National Starch and Chemical Company for \$148.0, subject to adjustment. The sales and revenues of this business for the three and six months ended June 30, 1997 were \$6.8 and \$24.8, respectively (\$17.6 and \$36.1 for the three and six months ended June 30, 1996, respectively); its financial position and results of operations were not significant to Grace. The sale of this business resulted in a pretax gain of \$103.1, and an after-tax gain of \$63.0 (\$.86 per common share), in continuing operations.

In June 1996, Grace sold its water treatment and process chemicals business (Dearborn). The sales and revenues of this business for the three and six months ended June 30, 1996 were \$103.6 and \$201.2, respectively; its financial position and results of operations were not significant to Grace. The sale of this business and the biopesticides business (sold in the second quarter of 1996) resulted in a pretax gain of \$326.4, and an after-tax gain of \$210.1 (\$2.18 per common share), in continuing operations.

4. DISCONTINUED OPERATIONS

In February 1997, Grace sold its cocoa business to Archer-Daniels-Midland Company for total proceeds of \$470.0 (inclusive of debt assumed by the buyer), subject to adjustment. The pretax and after-tax effects of the divestment were consistent with prior estimates and have been charged against previously established reserves.

In the fourth quarter of 1996, Grace classified its thermal and emission control systems business (TEC Systems) as a discontinued operation. Grace classified its health care business as a discontinued operation in the second quarter of 1995 and disposed of that business in 1996. Results of these discontinued operations that were not charged against previously established reserves, and the gain on the May 1996 sale of Grace's transgenic plant business, were as follows:

	Three Months Ended June 30, 1996	
HEALTH CARE Sales and revenues	\$565.5	\$1,105.2
Income from operations before taxes(1) Income tax provision	\$ 38.9 17.5	\$ 77.1 33.7
Income from discontinued health care operations	s \$ 21.4	\$ 43.4
TEC SYSTEMS Sales and revenues	\$ 28.5	\$ 52.9
Loss from operations before taxes	\$ (3.1) (1.2)	\$ (6.4) (2.5)
Loss from discontinued TEC Systems operations	\$ (1.9)	\$ (3.9)
Total operating results	\$ 19.5	\$ 39.5
Gain on sale of transgenic plant business Provision for income taxes on sale of transgenic	129.0	129.0
plant business	49.6	49.6
Total income from discontinued operations	\$ 98.9	\$ 118.9
	=====	======

(1) Reflects allocated interest expense of \$24.4 and \$51.2 for the three and six months ended June 30, 1996, respectively, based on the ratio of the net assets of the health care business compared to Grace's total capital.

For the three and six months ended June 30, 1997, the operating results of TEC Systems, the cocoa business and other discontinued operations have been charged against previously established reserves and, therefore, are not reflected in the Consolidated Statement of Operations.

The net assets of discontinued operations at June 30, 1997 primarily reflect TEC Systems. The components of the net assets of Grace's discontinued operations (excluding intercompany assets) at June 30, 1997 are as follows:

	June 30,
Current assets	\$42.4 15.2 12.1 6.3 \$76.0
Current liabilities	\$17.8 1.4
Net assets	\$19.2 \$56.8 =====

In June 1997, Grace entered into an agreement to sell TEC Systems to Sequa Corporation.

For additional information, see Note 6 to the consolidated financial statements in the 1996 Form 10-K.

5. RESTRUCTURING COSTS

As discussed in Note 4 to the consolidated financial statements in the 1996 Form 10-K, Grace began implementing a worldwide program in 1995 focused on streamlining processes and reducing general and administrative expenses, factory administration costs and noncore corporate research and development expenses. As previously reported, Grace has continued to implement additional cost reductions and efficiency improvements beyond those initiated in 1995, as its businesses have further evaluated and reengineered their operations. As a result of these evaluations, in the second quarters of 1997 and 1996, Grace recorded pretax charges of \$12.4 (\$8.0 after-tax) and \$53.7 (\$32.4 after-tax), respectively, principally related to the restructuring of its packaging business. The 1997 charge primarily relates to the restructuring of the packaging business from a worldwide group of independent regional units into an integrated global organization, and is primarily comprised of employee termination benefits. The 1996 charge primarily related to the restructuring of Grace's European packaging business and consisted of costs related to employee termination benefits and lease termination costs.

6. INVENTORIES

The components of Grace's inventories are as follows:

J	une 30, 1997	Dec	ember 31, 1996
\$	101.8 70.7 184.5 65.0	\$	100.9 67.6 179.0 73.4
	(46.0)		(44.8)
\$	376.0	\$	376.1
	\$	\$ 101.8 70.7 184.5 65.0 (46.0)	\$ 101.8 \$ 70.7 184.5 65.0 (46.0)

7. OTHER ASSETS

The components of Grace's other assets are as follows:

	J	une 30, 1997		ember 31, 1996	
Prepaid pension costs	\$	279.0	\$	275.1	
(December 31, 1996 - \$42.7)		126.8		152.9	
Deferred charges		105.2		102.4	
Other		103.3		123.1	
	\$	614.3	\$	653.5	
	==:	======	===	======	

8. SHAREHOLDERS' EQUITY

During the first half of 1997, the Company substantially completed the share repurchase program initiated in 1996 by acquiring 6,306,300 additional shares of its common stock for \$335.9, or an average price of \$53.26 per share. For additional information, see Note 13 to the consolidated financial statements in the 1996 Form 10-K.

In 1997, Grace established a trust to fund certain deferred employee incentive compensation and nonemployee director compensation and benefits. The shares held in the trust are valued at the closing market price at the end of each reporting period.

9. EARNINGS PER SHARE

In calculating primary earnings per share, the dilutive effect of common stock equivalents was not material (as defined by Accounting Principles Board Opinion No. 15, "Earnings Per Share") for the periods presented. Accordingly, earnings per share presented in the Consolidated Statement of Operations are computed on the basis of the weighted average number of shares outstanding.

In the first quarter of 1997, the Financial Accounting Standards Board issued Statement of Financial Accounting Standards (SFAS) No. 128, "Earnings Per Share," which establishes new standards for computing and presenting earnings per share effective December 31, 1997. At December 31, 1997, all prior periods will be restated to reflect the new basic and diluted earnings per share amounts required by SFAS No. 128. The pro forma effect of the adoption of SFAS No. 128 on earnings per share is not material for the periods presented.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF RESULTS OF OPERATIONS AND FINANCIAL CONDITION

REVIEW OF OPERATIONS

OVERVIEW

Grace is one of the world's leading packaging and specialty chemicals companies. Grace's core businesses are packaging, catalysts and other silica-based products (Grace Davison), and construction products.

Sales and revenues decreased 8.5% for the second quarter of 1997 compared to the second quarter of 1996 and decreased 8.7% for the first half of 1997 compared to the first half of 1996. Excluding divested businesses, sales and revenues for the 1997 second quarter increased 4.6%, and sales and revenues for the first six months of 1997 increased 3.8%, over the comparable periods of 1996. Pretax income from continuing operations was \$189.7 million for the second quarter of 1997, a 48.3% decrease compared to the 1996 second quarter, and was \$263.4 million for the first six months of 1997, a 39.6% decrease compared to the 1996 first half. As noted in the table below, pretax income from continuing operations for the three and six months ended June 30, 1997 and 1996 was affected by various special items. Excluding these special items, Grace's pretax operating income was \$118.5 million for the 1997 second quarter, an increase of 12.6% over the 1996 quarter, and \$209.8 million for the six months ended June 30, 1997, an increase of 9.0% over the same period in 1996. Pretax operating results for the three and six months ended June 30, 1996 have been restated to reflect the classification of certain businesses as discontinued operations.

TAX OPERATING RESULTS - CONTINUING OPERATIONS Three Months Ended June 30,		7 00			
	1997	1996	1997	1996	
Sales and revenues, excluding divested businesses Sales and revenues of divested businesses(1)	\$835.7	\$798.8 121.5	\$1,602.8 24.8	\$1,543.9 238.0	
Sales and revenues		\$920.3	\$1,627.6 ======	\$1,781.9	
Operating income, excluding special items and divested businesses		3.2	\$ 206.0	4.1	
Operating income before special items			\$ 209.8		
Gain on sales of businesses				(53.7)	
Operating income from continuing operations Other income/(expense):	\$209.2		\$ 300.5		
Interest expense and related financing costs Other income, net	.8	7.1	2.3		
Income from continuing operations	\$189.7 =====	\$366.7	\$ 263.4	\$ 436.0	

(1) Primarily reflects Grace's specialty polymers business, divested in May 1997, and Grace's water treatment and process chemicals business, divested in June 1996.

The following discussion includes projections and/or other "forward-looking" information. Grace is subject to risks and other uncertainties that could cause its actual results to differ materially from any such projections or that could cause other forward-looking information to prove incorrect. For a discussion of such risks and uncertainties, see "Introduction and Overview -- Projections and Other Forward-Looking Information" in Item 1 of the 1996 Form 10-K.

SALES AND REVENUES

SALES AND REVENUES (excluding divested businesses) (In millions)	1997	1996	30, % Change 1997 VS. 1996
Packaging	\$531.8	\$493.8	7.7%
Grace Davison	175.5	190.2	(7.7)
Construction Products	127.8	114.2	11.9
Other	.6	.6	
SALES AND REVENUES	\$835.7	\$798.8	4.6%
	======	======	

SALES AND REVENUES (excluding divested businesses) (In millions)	1997	ded June 30, 1996	% Change 1997 VS. 1996
Packaging	\$1,020.6 350.1 230.8 1.3 \$1,602.8	\$ 968.4 375.8 198.5 1.2 \$1,543.9	5.4% (6.8) 16.3 8.3

As noted above, sales and revenues (excluding divested businesses) increased 4.6% and 3.8% for the three-month and six-month periods ended June 30, 1997, respectively, over the same periods in 1996. Excluding unfavorable currency translation variances estimated at 3.4% and 3.1%, sales and revenues in the 1997 three- and six-month periods increased by an estimated 8.0% and 6.9%, respectively, over the 1996 periods. The following is a discussion of the sales and revenues of Grace's product lines.

PACKAGING

Sales and revenues of \$531.8 million for the second quarter and \$1,020.6 million for the first six months of 1997 increased 7.7% and 5.4%, respectively, versus the comparable 1996 periods. Excluding unfavorable currency translation variances estimated at 3.8% and 3.3%, sales and revenues in the 1997 three- and six-month periods increased by an estimated 11.5% and 8.7%, respectively, over the 1996 periods. Packaging sales volume in 1997 has been positively affected by the July 1996 acquisition of Cypress Packaging, Inc., a leading supplier of plastic packaging materials for the retail pre-cut produce market, the August 1996 acquisition of Bayem S.A. de C.V. (Bayem), a Mexican producer of can coatings and closure sealants for the rigid container industry, and the April 1997 acquisition of Schurpack, a U.S. manufacturer of plastic laminate packaging materials for the institutional and retail cook-in market segment. These acquisitions accounted for approximately 28% and 35% of the overall sales increases for the 1997 second quarter and first half, respectively. In addition to these acquisitions, Packaging experienced sales growth within its product groups, as described below (in all cases excluding the effect of currency translation).

Bag sales increased across all geographic regions for the second quarter and first half of 1997 over the second quarter and first half of 1996. Volumes increased in North America as a result of sales of fresh red meat bags to new customers and the continued penetration of TBG(TM) boneguard packaging products into the fresh beef segment. In addition, sales increased as a result of price increases in North America that went into effect during the second quarter of 1997; these price increases also resulted in volume increases, as customers made additional purchases prior to the price increases. These increases, however, were moderately offset by continued softness in the North American pork market, reflecting reduced slaughter rates stemming from 1996 livestock reductions caused by higher prices for corn and other feeds. This market is expected to improve over time as corn prices further stabilize and livestock numbers return to more normal levels. European bag sales increased over the second quarter of 1996, as fresh red meat sales (primarily in the U.K. and France) have partially recovered from consumer fears associated with publicity surrounding bovine spongiform encephalopathy - commonly referred to as "mad cow disease." Strong sales in the cheese and processed (smoked and cured) meat segments in northeastern Europe (primarily Poland and Russia) continued in the second quarter. The increase in cheese sales was primarily due to substantial growth in Russian domestic sales due to an improving economy, which in turn drove large exports of cheese from Poland to the Russian market; the processed meat increase was primarily due to the modernization of food distribution infrastructures in northeastern Europe. Volumes in Latin America increased due to the growing acceptance of boxed beef packaging and increased exports of fresh red meat to Europe as a result of the lingering effects of mad cow disease. Bag sales volumes in Asia Pacific increased as a result of strong demand in the Australian and New Zealand beef and lamb markets. These increases were partially offset by sales volume decreases in Japan due to a continuing slow economy and a decline in beef consumption as consumers become more health conscious.

Second quarter and first half 1997 laminate sales increased 14.0% and 8.3%, respectively, as compared to the 1996 periods, primarily due to increases in North America and Latin America. In North America, volumes increased as a result of increased customer acceptance of packaging liquid products in flexible packaging and the Schurpack acquisition discussed above. Sales growth in Latin America resulted from increased demand for cook-in and food service packaging for the processed and prepared foods segment.

Film sales for the second quarter and first half of 1997 increased in all regions as compared to the second quarter and first half of 1996. In North America, film sales increased as a result of new product introductions for industrial and consumer goods applications. Volumes were also favorably impacted by increased purchases prior to a price increase that took effect in May 1997. In Europe and Latin America, film sales increased primarily due to strong volumes, resulting from a strengthening of distribution channels, including greater use of distributors, which enabled wider access to market opportunities.

Container sealants and coatings sales increased for the second quarter and first half of 1997 compared to the same periods in 1996. In Latin America, volumes increased as a result of improved market penetration of can coating products, primarily due to the acquisition of Bayem discussed above. These increases were partially offset by unfavorable variances within Europe and Asia Pacific. European sales decreased primarily due to an unseasonably cool and rainy spring, which resulted in decreased European beverage consumption. Additionally, two European customers reverted to manufacturing their own closure sealants in the second quarter of 1997. The Asia Pacific sales decrease resulted primarily from the depletion of customers' excess inventory in China and the increased penetration of alternative forms of packaging, such as plastic and glass.

GRACE DAVISON

Sales of catalysts and other silica-based products for the second quarter and first six months of 1997 declined 7.7% and 6.8%, respectively, compared to the same periods in 1996, due primarily to the depressed fluid cracking catalyst market, as discussed below. The effect of a strengthening dollar caused sales to decline by an estimated 4.0% and 3.7% for the three- and six-month periods ended June 30, 1997, respectively, compared to the 1996 periods.

Fluid cracking catalyst sales were negatively impacted by both price and volume reductions for the second quarter and first half of 1997 compared to the comparable periods in 1996. Pricing pressures, which began in the third quarter of 1996, appear to have lessened in the first half of 1997 in North America, but continue in Europe and Asia Pacific. Volume reductions were primarily caused by a large number of refinery turnarounds (temporary shutdowns of customers' refineries for maintenance, repair and upgrades) in the first quarter of 1997, a large nonrecurring order to the Middle East in the second quarter of 1996, the refining of "sweeter" crude oil, which requires less fluid cracking catalysts, and the loss of two customers in Asia Pacific during the first half of 1997.

Silica/adsorbent sales were higher for the second quarter and first half of 1997 compared to the second quarter and first half of 1996. These increases were primarily due to increased volumes in North America from sales of new products in the graphic arts market, and in Asia Pacific as a result of the new silica plant in Kuantan, Malaysia. This new plant began operation in the fourth quarter of 1996, and increased production is expected over the next two quarters.

Polyolefin catalyst sales increased for both the three and six months ended June 30, 1997 compared to the same periods of 1996. Volumes increased primarily due to the addition of two new customers. Total 1997 sales of polyolefin catalysts are expected to increase compared to 1996 fueled by a strong worldwide resin industry.

CONSTRUCTION PRODUCTS

Construction Products had record sales of \$127.8 million for the 1997 second quarter and \$230.8 million for the first six months of 1997, increasing 11.9% and 16.3%, respectively, over the comparable 1996 periods, driven by volume increases. Higher volumes of cement additives and concrete admixtures accounted for approximately half of the increase, with increased volumes of fire protection and waterproofing products contributing significantly to the remainder of the growth. New value-added products, primarily water-reducing and anti-corrosion concrete admixtures and waterproofing products, continued to show strong sales growth as a result of greater market acceptance. Water-reducing concrete admixtures enhance concrete strength and surface texture and provide greater plasticity during pouring. Anti-corrosion concrete admixtures significantly increase the life of concrete structures by retarding the oxidation of steel reinforcement bars.

Sales grew in all geographic regions for the second quarter and first six months of 1997. North American sales increased by 13.0% and 20.3% in the 1997 second quarter and six months, respectively, over the 1996 periods, contributing approximately 70% and 77% of the overall sales increase in the 1997 second quarter and first six months, respectively. Second quarter construction activity in North America remained strong after a mild winter in the northeastern U.S. allowed for more construction activity than normal in the first quarter, especially when compared to the unusually harsh winter experienced in the 1996 first quarter. Market share gains and market penetration of new value-added products, discussed above, also drove North American sales growth. Sales in Europe grew faster in the second quarter than in the first quarter, recovering from weather-related construction delays at the beginning of the year and benefiting from an improving economy in the United Kingdom. Geographic expansion into Asia Pacific and Latin America, where infrastructure building activity helped to drive sales, progressed favorably.

OPERATING RESULTS

OPERATING INCOME (In millions) (excluding special items and divested businesses)	1997	1996	30, % Change 1997 VS. 1996
Packaging	. \$ 78.6	\$ 64.1	22.6%
Grace Davison	. 21.3	24.2	(12.0)
Construction Products	. 15.3	11.9	28.6
Other	. 2.1	1.8	16.7
OPERATING INCOME	. \$117.3	\$102.0	15.0%
	=====	=====	

Six Months Ended June 30, OPERATING INCOME (In millions) % Change (excluding special items and divested businesses) 1997 VS. 1996 1997 1996 -----\$145.8 \$124.1 17.5% Packaging..... Grace Davison..... 39.0 52.3 (25.4)Construction Products..... 19.1 8.3 130.1 Other..... 2.1 3.6 (41.7)OPERATING INCOME..... \$206.0 \$188.3 9.4%

Cost management programs initially implemented in 1995 continued to favorably impact pretax operating income across all geographic regions and product lines. Grace has implemented, and expects to further implement, additional cost reductions and efficiency improvements, as it further evaluates and reengineers its manufacturing productivity and operations. The following is a discussion of the operating results of Grace's product lines.

PACKAGING

Packaging pretax operating income increased 22.6% and 17.5% for the second quarter and first half of 1997, respectively, compared to the second quarter and first half of 1996. The favorable variances were primarily due to the volume and price increases discussed above, as well as ongoing cost containment efforts, such as lower factory administration and selling expenses. However, these improvements in operating income were partially offset by increased expenses, including an increase in raw material prices for resins in North America, higher expenses (primarily depreciation and amortization expenses) associated with the new packaging plant in Kuantan, Malaysia that began operations in the fourth quarter of 1996, and an increase in research and development expenses as a result of the continued emphasis on new product development.

GRACE DAVISON

Grace Davison operating income for the second quarter and first six months of 1997 declined 12.0% and 25.4%, respectively, compared to the same periods in 1996. A weak fluid cracking catalyst market resulted in both volume and price decreases, and foreign exchange translation unfavorably impacted earnings. There were also nonrecurring items contributing to the decrease in operating income between the first half of 1997 and 1996. In particular, the large number of refinery turnarounds (discussed above) decreased sales, and harsh winter weather at Grace Davison's Lake Charles, Louisiana facility increased repair and maintenance costs. These charges were offset by manufacturing efficiencies and ongoing cost reduction efforts. Despite the decline in operating income, Grace Davison has maintained its market share, and the introductions of new manufacturing process technologies and new products have improved margins throughout the year.

CONSTRUCTION PRODUCTS

Construction Products pretax operating income increased 28.6% and 130.1% for the second quarter and first six months of 1997, respectively, over the 1996 comparable periods. The favorable results were primarily due to the record sales levels achieved in 1997, discussed above. Production rate improvements from enhancements to manufacturing processes, along with additional cost containment efforts, also contributed to earnings growth.

OTHER ITEMS

INTEREST EXPENSE AND RELATED FINANCING COSTS

Interest expense and related financing costs for continuing operations of \$20.3 million and \$39.4 million for the second quarter and first half of 1997, respectively, increased 10.9% and 7.4%, respectively, compared to the 1996 periods. Including amounts allocated to discontinued operations, interest expense and related financing costs decreased 52.5% and 55.2% during the second quarter and first half of 1997, respectively, compared to the second quarter and first half of 1996. The decrease was primarily due to lower average debt levels (as a result of debt repayments made using the proceeds from the September 1996 separation of Grace's principal health care business and other divestments) and lower average effective interest rates.

See "Financial Condition: Liquidity and Capital Resources" below for further information on borrowings.

RESEARCH AND DEVELOPMENT EXPENSES

Research and development (R&D) spending decreased 17.2% and 17.0% for the second quarter and first half of 1997, respectively, compared to the second quarter and first half of 1996. The decrease reflects the continued positive impact of cost management initiatives implemented during 1996 and 1995, primarily the elimination of Grace's corporate research organization, the transfer of core R&D activities to the product lines and the termination of R&D activities not related to Grace's core packaging and specialty chemicals businesses. The decrease is also attributable to the elimination of R&D spending related to Grace's water treatment and process chemicals business, which was divested in June 1996. As a result of these initiatives, Grace has been able to increase R&D spending for its core businesses while reducing total R&D expenses. Grace's R&D spending will continue to be directed toward its core businesses.

RESTRUCTURING COSTS

As discussed in Note 4 to the consolidated financial statements in the 1996 Form 10-K, Grace began implementing a worldwide program in 1995 focused on streamlining processes and reducing general and administrative expenses, factory administration costs and noncore corporate research and development expenses. As previously reported, Grace has continued to implement additional cost reductions and efficiency improvements beyond those initiated in 1995, as its businesses have further evaluated and reengineered their operations. As a result of these evaluations, in the second quarters of 1997 and 1996, Grace recorded pretax charges of \$12.4 million (\$8.0 million after-tax) and \$53.7 million (\$32.4 million after-tax), respectively, principally related to the restructuring of its packaging business. The 1997 charge primarily relates to the restructuring of the packaging business from a worldwide group of independent regional units into an integrated global organization; the 1996 charge primarily related to the restructuring of Grace's European packaging operations. The 1997 restructuring is expected to spur sales growth, improve manufacturing productivity, reduce costs, shorten product development time and enhance customer service for the packaging business. The 1997 charge is primarily comprised of employee termination benefits, and the 1996 charge consisted of costs related to employee termination benefits and lease termination costs.

INCOME TAXES

Grace's effective tax rates were 38.1% and 37.8%, respectively, for the second quarter and first six months of 1997, compared with 35.9% and 36.1%, respectively, for the second quarter and first six months of 1996. The variance was largely due to higher state and foreign taxes on the gain on Grace's sale of the specialty polymers business. Excluding special items, the effective tax rates were 37.0% for the second quarter and first six months of 1997 and 39.0% and 38.2%, respectively, for the second quarter and first six months of 1996, primarily due to lower taxes on foreign operations in the 1997 periods.

DISCONTINUED OPERATIONS

During 1996, Grace completed the separation of National Medical Care, Inc. and sold its separations science business (Amicon). These health care businesses had been classified as discontinued operations in 1995. Second quarter and first half 1996 income from discontinued operations of \$98.9 million and \$118.9 million, respectively, included net income of \$21.4 million (\$38.9 million pretax) and \$43.4 million (\$77.1 million pretax), respectively, from health care operations. In 1996, Grace classified TEC Systems as a discontinued operation. Second quarter and first half 1996 income from discontinued operations included a loss of \$1.9 million (\$3.1 million pretax) and \$3.9 million (\$6.4 million pretax), respectively, from TEC Systems. In May 1996, Grace completed the sale of the transgenic plant business for \$150.0 million in cash, resulting in a pretax gain of \$129.0 million, and an after-tax gain of \$79.4 million, in discontinued operations.

FINANCIAL CONDITION

LIQUIDITY AND CAPITAL RESOURCES

Grace's continuing operating activities provided net pretax cash of \$94.1 million in the first half of 1997, as compared to \$117.1 million in the first half of 1996. Improved operating income cash flow was offset by higher payments made for asbestos-related litigation settlements, judgments and defense costs, net of amounts received from settlements with certain insurance carriers in connection with asbestos-related litigation. Net pretax cash provided by operating activities of discontinued operations in the first half of 1997 decreased by \$67.2 million compared to the first half of 1996, primarily due to the disposition of Grace's health care and cocoa businesses, which provided cash in the first six months of 1996. After giving effect to the payment of income taxes, Grace's net cash provided by operating activities was \$40.8 million in the first half of 1997, versus \$118.5 million in the first half of 1996.

Investing activities provided \$481.0 million of cash in the first half of 1997, largely reflecting net cash proceeds of \$450.7 million from divestments (primarily the sale of Grace's specialty polymers and cocoa businesses) and the receipt of \$215.6 million in January 1997 on the 1996 divestments of Dearborn and Amicon. Grace made capital expenditures of \$113.6 million in the six months ended June 30, 1997, primarily related to the Packaging and Grace Davison businesses. Total Grace capital expenditures for 1997 are not expected to exceed \$300.0 million and will all be directed towards core businesses.

Net cash used for financing activities in the first half of 1997 was \$544.1 million, primarily reflecting reductions in debt, the repurchase of stock as discussed below, and the payment of dividends, partially offset by proceeds from the exercise of employee stock options. Total debt was \$1,167.4 million at June 30, 1997, a decrease of \$220.8 million from December 31, 1996.

During the first half of 1997, the Company substantially completed the share repurchase program initiated in 1996 by acquiring 6,306,300 additional shares of its common stock for \$335.9 million, or an average price of \$53.26 per share.

Grace is targeting a ratio of debt to earnings before interest, taxes, depreciation and amortization (EBITDA) of 1.6 to 2.0. The debt/EBITDA ratio was 1.8 at June 30, 1997.

At June 30, 1997, Grace had committed borrowing facilities totaling \$1.0 billion, consisting of \$650.0 million under a 364-day facility expiring in May 1998 (extendible for successive 364-day periods at the discretion of Grace and the lenders) and \$350.0 million under a long-term facility expiring in May 2002. As of June 30, 1997, \$606.7 million was available under these facilities.

ASBESTOS-RELATED MATTERS

In the first half of 1997, Grace paid \$7.4 million for the defense and disposition of asbestos-related property damage and personal injury litigation, net of amounts received under settlements with certain insurance carriers. Although the amounts to be paid in 1997 with respect to asbestos-related claims (after giving effect to payments to be received from insurance carriers) cannot be precisely estimated, Grace expects that it will be required to expend approximately \$75-\$100 million (pretax) in 1997 to defend against and dispose of such claims (after giving effect to anticipated insurance recoveries). The amounts with respect to the probable cost of defending against and disposing of asbestos-related claims and probable recoveries from insurance carriers represent estimates and are on an undiscounted basis; the outcomes of such claims cannot be predicted with certainty.

In May 1997, the Texas legislature adopted legislation that had the effect of making it more difficult for out-of-state residents to file claims in Texas state courts. The legislation is retroactive to January 1, 1997. The effect of this legislation on Grace's ultimate exposure with respect to its asbestos-related cases and claims cannot be predicted with certainty.

See Note 2 to the interim consolidated financial statements in this Report for further information concerning asbestos-related lawsuits and claims.

ENVIRONMENTAL MATTERS

There were no significant developments relating to environmental liabilities in the first half of 1997. For additional information relating to environmental liabilities, see Note 11 to the consolidated financial statements in the 1996 Form 10-K.

PART II - OTHER INFORMATION

Item 1. Legal Proceedings.

Note 2 to the Consolidated Financial Statements in Part I of this Report is incorporated herein by reference.

Item 4. Submission of Matters to a Vote of Security Holders.

The Company's 1997 Annual Meeting of Shareholders ("Annual Meeting") was held on May 9, 1997. At the Annual Meeting, the Company's shareholders (a) elected four Class II Directors for a term expiring in 2000; (b) ratified the selection of Price Waterhouse LLP as independent certified public accountants of the Company and its consolidated subsidiaries for 1997; (c) approved the Company's 1996 Stock Incentive Plan; (d) approved the Company's Long-Term Incentive Program; (e) approved the Company's Annual Incentive Compensation Program; and (f) approved the Company's 1997 Stock Plan for Nonemployee Directors.

The results of voting at the Annual Meeting appear on the following page:

\/	\cap	ΓE	C	*

MATTER	FOR	AGAINST**	ABSTENTIONS
Election of Directors**			
J. F. Akers C. Cheng V. A. Kamsky J. E. Phipps	65,313,700 64,751,489 56,879,213 56,876,922	419,217 981,428 8,853,704 8,855,995	- 0 - - 0 - - 0 - - 0 -
Selection of Independent Accountants	64,952,184	502,715	278,018
Approval of 1996 Stock Incentive Plan	39,958,099	25,308,108	466,709
Approval of Long-Term Incentive Program	61,720,386	3,504,974	507,556
Approval of Annual Incentive Compensation Program	62,745,711	2,414,648	572,555
Approval of 1997 Stock Plan for Nonemployee Directors	62,213,024	2,962,113	557,680

There were no broker non-votes on any matter.
With respect to the election of directors, the form of proxy permitted shareholders to check boxes indicating votes either "for" or "withheld;" votes relating to directors designated above as "against" are votes cast as "withheld."

Item 5. Other Information.

On June 23, 1997, Grace announced that it had entered into an agreement to sell its TEC Systems unit to Sequa Corporation. Grace TEC Systems is a leading global supplier of high-performance air flotation dryers and other web handling equipment for the printing, industrial coating and paper industries.

Item 6. Exhibits and Reports on Form 8-K.

- (a) Exhibits. The following are being filed as exhibits to this Report:
 - -- 364-Day Credit Agreement, dated as of May 16, 1997, among W. R. Grace & Co.-Conn., W. R. Grace & Co., the several banks parties thereto, NationsBank, N.A. (South), as documentation agent, and The Chase Manhattan Bank, as administrative agent for such banks
 - -- Credit Agreement, dated as of May 16, 1997, among W. R. Grace & Co.-Conn., W. R. Grace & Co., the several banks parties thereto, and The Chase Manhattan Bank, as administrative agent for such banks
 - -- weighted average number of shares and earnings used in per share computations
 - -- computation of ratio of earnings to fixed charges and combined fixed charges and preferred stock dividends
 - -- financial data schedule

(b) Reports on Form 8-K. The Company filed the following Reports on Form 8-K during the second quarter and to date during the third quarter of 1997:

Date of Filing May 1, 1997 August 5, 1997

Disclosure(s)
----------Announcement of 1997 first quarter results
Announcement of 1997 second quarter results

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

W. R. GRACE & CO. (Registrant)

Date: August 12, 1997

By /s/ Kathleen A. Browne

Kathleen A. Browne Vice President and Controller (Principal Accounting Officer)

W. R. GRACE & CO.

QUARTERLY REPORT ON FORM 10-Q FOR THE QUARTER ENDED JUNE 30, 1997

EXHIBIT INDEX

EXHIBIT NO.	DESCRIPTION
10.1	364-Day Credit Agreement, dated as of May 16, 1997, among W. R. Grace & CoConn., W. R. Grace & Co., the several banks parties thereto, NationsBank, N.A. (South), as documentation agent, and The Chase Manhattan Bank, as administrative agent for such banks
10.2	Credit Agreement, dated as of May 16, 1997, among W. R. Grace & CoConn., W. R. Grace & Co., the several banks parties thereto, and The Chase Manhattan Bank, as administrative agent for such banks
11	Weighted average number of shares and earnings used in per share computations
12	Computation of ratio of earnings to fixed charges and combined fixed charges and preferred stock dividends
27	Financial Data Schedule

364-DAY CREDIT AGREEMENT

DATED AS OF MAY 16, 1997

AMONG

W. R. GRACE & CO.-CONN., W. R. GRACE & CO.,

The Banks Party Hereto,

NATIONSBANK, N.A. (SOUTH), as Documentation Agent,

and

THE CHASE MANHATTAN BANK, AS ADMINISTRATIVE AGENT

TABLE OF CONTENTS

		Page
SECTION	1. 1.1 1.2	DEFINITIONS
SECTION	2. 2.1 2.2	AMOUNT AND TERMS OF COMMITMENTS
	2.3	Procedure for Revolving Credit Borrowing 16 Extension of Termination Date
SECTION	3. 3.1 3.2	BILATERAL OPTION LOANS
	3.3 3.4	Determination of Dollar Equivalents
SECTION	4. 4.1 4.2 4.3 4.4 4.5	BID LOANS
SECTION	5. 5.1 5.2 5.3	LOAN FACILITY COMMON PROVISIONS
	5.4 5.5 5.6 5.7 5.8 5.9 5.10 5.11 5.12 5.13	Prepayments
SECTION	6. 6.1 6.2	REPRESENTATIONS AND WARRANTIES
	6.3 6.4 6.5 6.6 6.7 6.8	Obligations

i

	6.9 6.10 6.11 6.12 6.13 6.14 6.15	Certain Federal Regulations	37 37 37 38 38
SECTION	7. 7.1 7.2		38 38 39
SECTION	8.1 8.2 8.3 8.4	Financial Statements. Certificates; Other Information. Payment of Obligations. Conduct of Business and Maintenance of Existence. Insurance.	41 42
	8.6 8.7 8.8		42 42 43
SECTION	9. 9.1 9.2 9.3 9.4	Financial Condition Covenants Limitation on Liens Limitation on Fundamental Changes Limitation on Asset Transfers to Foreign Subsidiaries	46
SECTION	10.	EVENTS OF DEFAULT	47
SECTION	11. 11.1 11.2 11.3 11.4 11.5		50 51
	11.7 11.8	Other Banks	
	11.9	CapacitySuccessor Administrative Agent	
SECTION	12. 12.1 12.2 12.3	GUARANTEES	53
	12.4	or Indemnity	
	12.5 12.6 12.7	Obligations	56 56

Page

SECTION	13.1 13.2 13.3	MISCELLANEOUS	57 59
	13.4	Survival of Representations and Warranties	60
	13.5	Payment of Expenses and Taxes	60
	13.6	Successors and Assigns; Participations;	
		Purchasing Banks	61
	13.7	Adjustments; Set-off	64
	13.8	Counterparts	65
	13.9	Severability	65
	13.10	Integration	65
	13.11	GOVERNING LAW	65
	13.12	Submission to Jurisdiction; Waivers	65
	13.13	Acknowledgments	66
	13.14	WAIVERS OF JURY TRIAL	67
	13.15	Additional Borrowers	67

SCHEDULES

Schedule I	Commitments; Lending Offices and Addresses for
Schedule II	Principal Subsidiaries
EXHIBITS	

Exhibit A Exhibit B Exhibit C	Form of Revolving Credit Note Form of Bid Loan Note Form of Bid Loan Confirmation
Exhibit D Exhibit E	Form of Bid Loan Offer Form of Bid Loan Request
Exhibit F-1	Form of Opinion of Counsel to the Company and the
Exhibit F-2 Exhibit G Exhibit H Exhibit I	Parent Form of Opinion of Simpson Thacher & Bartlett Form of Officer's Certificate Form of Commitment Transfer Supplement Form of Notice of Additional Borrower

364-DAY CREDIT AGREEMENT, dated as of May 16, 1997, among W. R. GRACE & CO.-CONN., a Connecticut corporation (the "Company"), W. R. GRACE & CO., a Delaware corporation and sole shareholder of the Company (the "Parent"), the several banks from time to time parties to this Agreement (the "Banks"), NATIONSBANK, N.A. (SOUTH), a national association, as documentation agent (in such capacity, the "Documentation Agent"), and THE CHASE MANHATTAN BANK, a New York banking corporation, as administrative agent for the Banks hereunder (in such capacity, the "Administrative Agent").

The parties hereto hereby agree as follows:

SECTION 1. DEFINITIONS

 ${\tt 1.1}$ Defined Terms. As used in this Agreement, the following terms shall have the following meanings:

"ABR Loans": Loans the rate of interest applicable to which is based upon the Alternate Base Rate.

"Additional Bank": as defined in subsection 2.4(b).

"Affiliate": as to any Person, (a) any other Person (other than a Subsidiary) which, directly or indirectly, is in control of, is controlled by, or is under common control with, such Person or (b) any Person who is a director, officer, shareholder or partner (i) of such Person, (ii) of any Subsidiary of such Person or (iii) of any Person described in the preceding clause (a). For purposes of this definition, "control" of a Person means the power, directly or indirectly, either to (i) vote 10% or more of the securities having ordinary voting power for the election of directors of such Person or (ii) direct or cause the direction of the management and policies of such Person whether by contract or otherwise.

"Aggregate Outstanding Bilateral Option Loans": at any time, (i) the aggregate outstanding principal amount of all Dollar Bilateral Loans and (ii) the aggregate Dollar Equivalents at such time with respect to all outstanding Alternative Currency Bilateral Loans.

"Agreement": this 364-Day Credit Agreement, as the same may be amended, supplemented or otherwise modified from time to time.

"Alternate Base Rate": for any day, a rate per annum (rounded upwards, if necessary, to the next 1/16 of 1%) equal to the greater of (a) the Prime Rate in effect on such day and (b) the Federal Funds Effective Rate in effect on such day plus 1/2 of 1%. "Prime Rate" shall mean the rate of interest per annum publicly announced from time to time by the Administrative Agent as its prime rate in

effect at its principal office in New York City. "Federal Funds Effective Rate" shall mean, for any day, the weighted average of the rates on overnight federal funds transactions with members of the Federal Reserve System arranged by federal funds brokers, as published on the next succeeding Business Day by the Federal Reserve Bank of New York, or, if such rate is not so published for any day which is a Business Day, the average of the quotations for the day of such transactions received by the Administrative Agent from three federal funds brokers of recognized standing selected by it. If for any reason the Administrative Agent shall have determined (which determination shall be conclusive absent manifest error) that it is unable to ascertain the Federal Funds Effective Rate, for any reason, including, the inability or failure of the Administrative Agent to obtain sufficient quotations in accordance with the terms hereof, the Alternate Base Rate shall be determined without regard to clause (b) of the first sentence of this definition until the circumstances giving rise to such inability no longer exist. Any change in the Alternate Base Rate due to a change in the Prime Rate or the Federal Funds Effective Rate shall be effective on the effective date of such change in the Prime Rate or the Federal Funds Effective Rate, respectively.

"Alternative Currency": any currency other than Dollars which is freely transferable and convertible into Dollars.

"Alternative Currency Bilateral Loan": a Loan made by a Bank to any Borrower in an Alternative Currency pursuant to Section 3.

"Applicable Margin": for any day on which the long term senior unenhanced, unsecured debt of the Company is rated by both S&P and Moody's, the rate per annum under the caption "Margin" (a "Margin Rate") set forth below opposite the S&P and Moody's ratings applicable to such debt on such day (or, if such ratings are set opposite two different Margin Rates, then the Applicable Margin shall be the lower of said two Margin Rates):

Margin	S&P	Moody's
. 450%	BB+ or lower	Ba1 or lower
. 300%	BBB-	ВааЗ
. 240%	BBB	Baa2
.210%	BBB+	Baa1
.195%	A- or higher	A3 or higher

provided that if on any day the long term senior unenhanced, unsecured debt of the Company is rated by only one of either S&P or Moody's, the Applicable Margin will be determined based on the rating by such rating agency, and provided, further, that if on any day the long term senior unenhanced, unsecured debt of the Company is rated by neither S&P nor Moody's, the Applicable Margin will be determined based on the rating of such debt by Duff & Phelps, Fitch or another nationally recognized statistical rating organization agreed to by and among the Company, the Administrative Agent and the Majority Banks (each, a "Substitute Rating Agency") and will be the Margin Rate set forth above opposite the S&P and Moody's ratings comparable to such Substitute Rating Agency's rating of such debt on such date, and provided, further, that if on any day the long term senior unenhanced, unsecured debt of the Company is rated by none of S&P, Moody's or any Substitute Rating Agency, the Company, the Administrative Agent and the Banks will negotiate in good faith to determine an alternative basis for calculating the Applicable Margin consistent with the table set forth above and, if agreement on such alternative basis is not reached within 30 days, the Applicable Margin will be calculated on an alternative basis determined by the Administrative Agent and the Banks in their reasonable discretion consistent with the table above, and until such alternative basis is determined the Applicable Margin will be the Applicable Margin last determined as provided in the table above.

"Available Commitment": as to any Bank at any time, an amount equal to the excess, if any, of (a) the amount of such Bank's Commitment over (b) the Loan Outstandings of such Bank at such time.

"Bid Loan": each Bid Loan made pursuant to Section 4.

"Bid Loan Banks": Banks which have outstanding Bid Loans or which are making Bid Loans.

"Bid Loan Confirmation": each confirmation by the Borrower of its acceptance of Bid Loan Offers, which Bid Loan Confirmation shall be substantially in the form of Exhibit C.

"Bid Loan Note": as defined in subsection 4.5(c); collectively, the "Bid Loan Notes".

"Bid Loan Offer": each offer by a Bank to make Bid Loans pursuant to a Bid Loan Request, which Bid Loan Offer shall contain the information specified in Exhibit D.

"Bid Loan Request": each request by a Borrower for Banks to submit bids to make Bid Loans at a fixed rate, which shall contain the information in respect of such requested Bid Loans specified in Exhibit E and shall be delivered to the Administrative Agent. "Bilateral Option Loan": a Loan made by a Bank to a Borrower pursuant to Section 3. Bilateral Option Loans may be either Dollar Bilateral Loans or Alternative Currency Bilateral Loans.

"Bilateral Option Loan Report": as defined in subsection 3.2.

"Board": The Board of Governors of the Federal Reserve System of the United States of America or any successor thereto.

"Borrower": the Company and any Subsidiary of the Company with respect to which a Notice of Additional Borrower has been given and all conditions precedent to the effectiveness thereof have been satisfied.

"Borrowing Date": any Business Day specified in a notice pursuant to subsection 2.3 and 4.2, as a date on which a Borrower requests the Banks to make Loans hereunder, or any date that a Bilateral Option Loan is made in accordance with subsection 3.1.

"Business Day": a day other than a Saturday, Sunday or other day on which commercial banks in New York City are authorized or required by law to close.

"Capitalized Lease": any lease of property, real or personal, the obligations of the lessee in respect of which are required to be capitalized in accordance with GAAP.

"Change of Control Date": (i) the first day on which the Company determines that any Person or group of related Persons has direct or indirect beneficial ownership of 30% or more of the outstanding capital stock of the Parent having ordinary voting power (other than stock having such power only by reason of the happening of a contingency) for the election of a majority of the board of directors of the Parent or (ii) the first day on which any Person or group of related Persons shall acquire all or substantially all of the assets of the Parent.

"Chase": The Chase Manhattan Bank.

"Closing Date": the first date on which the conditions set forth in subsection 7.1 have been satisfied or waived.

"Code": the Internal Revenue Code of 1986, as amended from time to time. $\,$

"Commitment": as to any Bank, the obligation of such Bank to make Revolving Credit Loans hereunder to the Borrowers in an aggregate principal amount at any one time outstanding not to exceed the amount set forth opposite such Bank's name on Schedule I under the heading "Commitment".

"Commitment Percentage": as to any Bank at any time, the percentage of the aggregate Commitments then constituted by such Bank's Commitment.

"Commitment Period": the period from and including the date hereof to but not including the Termination Date or such earlier date on which the Commitments shall terminate as provided herein.

"Commonly Controlled Entity": an entity, whether or not incorporated, which is under common control with the Company within the meaning of Section 4001(a)(14) of ERISA or is part of a group which includes the Company and which is treated as a single employer under subsection (b) or (c) of Section 414 of the Code.

"Consolidated Debt": at a particular date, with respect to the Parent and its Subsidiaries, and without duplication, the sum of the amounts set forth on a consolidated balance sheet of the Parent and its Subsidiaries prepared as of such date in accordance with GAAP opposite the captions (1) "Long-term debt" (or the equivalent caption) and (2) "Short-term debt" (or the equivalent caption) but always to include all indebtedness for borrowed money of the Parent and its Subsidiaries in accordance with GAAP.

"Consolidated Interest Expense": for any period, with respect to the Parent and its Subsidiaries, the amount which, in conformity with GAAP, would be set forth opposite the caption "Interest expense and related financing costs" (or the equivalent caption) on a consolidated statement of operations of the Parent and its Subsidiaries for such period.

"Continuing Banks": as defined in subsection 2.4(b).

"Contractual Obligation": as to any Person, any provision of any security issued by such Person or of any agreement, instrument or other undertaking to which such Person is a party or by which it or any of its property is bound.

"Default": any of the events specified in Section 10, whether or not any requirement for the giving of notice, the lapse of time, or both, has been satisfied.

"Dollar Bilateral Loan": a Bilateral Option Loan denominated in Dollars.

"Dollar Equivalent": on any date of determination by the Administrative Agent pursuant to subsection 3.2(b) or 3.2(c), as applicable, in respect of any Alternative Currency Bilateral Loan the amount of Dollars obtained by converting the outstanding amount of currency of such Alterative Currency Bilateral Loan, as specified in the then most recent Bilateral Option Loan Report, into Dollars at the spot rate for the purchase of Dollars with such currency as quoted by the Administrative Agent at its principal foreign exchange trading operations office in New York City on such date.

"Dollars" and "\$": dollars in lawful currency of the United States of America. $\,$

"Domestic Indebtedness": any Indebtedness of the Parent and any Domestic Subsidiary.

"Domestic Subsidiary": any Subsidiary of the Parent other than a Foreign Subsidiary.

"Duff & Phelps": Duff & Phelps, Inc.

"EBIT": for any period, with respect to the Parent and its Subsidiaries, (a) all amounts which would be set forth opposite the caption "Income from continuing operations before income taxes" (or the equivalent caption) on a consolidated statement of operations of the Parent and its Subsidiaries prepared in accordance with GAAP for such period plus (b) non-cash pre-tax charges arising from: (1) asset disposals (excluding the retirement of property, plant and equipment in the ordinary course of business) by the Parent and its Subsidiaries, (2) the implementation or modified application of financial accounting standards applicable to the Parent and its Subsidiaries, and (3) other special non-recurring transactions (including charges relating to Restructuring Activities, discontinued operations and asbestos related litigation and claims) (to the extent that such amounts have been deducted in determining the amount set forth opposite the caption "Income from continuing operations" (or the equivalent caption) for such period), plus (c) Consolidated Interest Expense for such period, plus (d) any payments received in such period in respect of non-cash pre-tax gains referred to in clause (e) of this definition, minus (e) non-cash pre-tax gains arising from: (1) asset disposals (excluding the retirement of property, plant and equipment in the ordinary course of business) by the Parent and its Subsidiaries, (2) the implementation or modified application of financial accounting standards applicable to the Parent and its Subsidiaries, and (3) other special non-recurring transactions (including charges relating to Restructuring Activities, discontinued operations and asbestos related litigation and claims) (to the extent that such amounts have been added in determining the amount set forth opposite the caption "Income from continuing operations" (or the equivalent caption) for such period), minus (f) any payments made in such period in respect of non-cash pre-tax charges referred to in clause (b) of this definition.

"EBITDA": for any period, EBIT plus all amounts which would be set forth opposite the caption "Depreciation and amortization" (or the equivalent caption) on a consolidated statement of operations of the Parent and its Subsidiaries prepared in accordance with GAAP for such period.

"Environmental Laws": any and all federal, state, local or municipal laws, rules, orders, regulations, statutes, ordinances, codes, decrees or requirements of any Governmental Authority regulating, relating to or imposing liability or standards of conduct concerning environmental protection matters, including without limitation, Hazardous Materials, as now or may at any time hereafter be in effect.

"ERISA": the Employee Retirement Income Security Act of 1974, as amended from time to time.

"Eurocurrency Reserve Requirements": for any day as applied to a Eurodollar Loan, the aggregate (without duplication) of the rates (expressed as a decimal fraction) of any reserve requirements in effect on such day (including, without limitation, basic, supplemental, marginal and emergency reserves under any regulations of the Board of Governors of the Federal Reserve System or other Governmental Authority having jurisdiction with respect thereto) dealing with reserve requirements prescribed for eurocurrency funding (currently referred to as "Eurocurrency Liabilities" in Regulation D of such Board) maintained by a member bank of such System.

"Eurodollar Loans": Loans the rate of interest applicable to which is based upon the Eurodollar Rate.

"Eurodollar Rate": with respect to each day during each Interest Period pertaining to a Eurodollar Loan, the rate per annum equal to the rate at which Chase is offered Dollar deposits at or about 10:00 A.M., New York City time, two Business Days prior to the beginning of such Interest Period in the interbank eurodollar market where the eurodollar and foreign currency and exchange operations in respect of its Eurodollar Loans are then being conducted for delivery on the first day of such Interest Period for the number of days comprised therein and in an amount comparable to the amount of its Eurodollar Loan to be outstanding during such Interest Period.

"Eurodollar Tranche": the collective reference to Eurodollar Loans, the Interest Periods with respect to all of which begin on the same date and end on the same later date (whether or not such Loans shall originally have been made on the same day).

"Event of Default": any of the events specified in Section 10, provided that any requirement for the giving of notice, the lapse of time, or both, has been satisfied. "Excluded Subsidiaries": GN Holdings, Inc., a Delaware corporation; CCHP, Inc., a Delaware corporation; Assignment America, Inc., a Delaware corporation; Cross Country Staffing, a Delaware general partnership, and each of their respective Subsidiaries.

"Existing Termination Date": as defined in subsection 2.4(a).

"FASB 5": Statement of Financial Accounting Standards No. 5, Accounting for Contingencies, of the Financial Accounting Standards Board, as the same may be from time to time supplemented, amended or interpreted by such Board.

"Fitch": Fitch Investors Service Inc.

"Foreign Subsidiary": any Subsidiary of the Parent (i) that is organized under the laws of any jurisdiction other than any state (including the District of Columbia), territory or possession of the United States of America (a "foreign jurisdiction"), or (ii) more than 50 percent of the book value of the assets of which (as of the end of the most recent fiscal period for which financial statements are required to have been provided pursuant to subsection 8.1(a) or (b)) are located in one or more foreign jurisdictions, or (iii) more than 50 percent of the Net Sales and Revenues of which (for the most recent fiscal year for which financial statements are required to have been provided pursuant to subsection 8.1(a)) were from sales made and/or services provided in one or more foreign jurisdictions, or (iv) more than 50 percent of the book value of the assets of which (as of the end of the most recent fiscal period for which financial statements are required to have been provided pursuant to subsection 8.1(a) or (b)) consists of equity interests in and/or Indebtedness of one or more Subsidiaries that are "Foreign Subsidiaries" within clauses (i), (ii), (iii) or (iv) of this definition.

"Foreign Subsidiary Indebtedness": any Indebtedness of any Foreign Subsidiary.

"GAAP": generally accepted accounting principles in the United States of America in effect from time to time.

"Governmental Authority": any nation or government, any state or other political subdivision thereof and any entity exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government.

"Hazardous Materials": any hazardous materials, hazardous wastes, hazardous constituents, hazardous or toxic substances, petroleum products (including crude oil or any fraction thereof), defined or regulated as such in or under any Environmental Law.

"Indebtedness": of any Person at any date, (a) all indebtedness of such Person for borrowed money or for the deferred purchase price of property or services (other than current trade liabilities incurred in the ordinary course of business and payable in accordance with customary practices) or which is evidenced by a note, bond, debenture or similar instrument, (b) all obligations of such Person under Capitalized Leases, and (c) without duplication, all "loss contingencies" of such Person of the types described in paragraph 12 of FASB 5, whether or not disclosed or required to be disclosed on the financial statements or footnotes thereto of such Person pursuant to GAAP.

"Insolvency": with respect to any Multiemployer Plan, the condition that such Plan is insolvent within the meaning of Section 4245 of ERISA.

"Insolvent": pertaining to a condition of Insolvency.

"Interest Payment Date": (a) as to any ABR Loan, the fifteenth day of each March, June, September and December to occur while such Loan is outstanding and, if different, the Termination Date, and (b) as to any Eurodollar Loan having an Interest Period of three months or less, the last day of such Interest Period, and (c) as to any Eurodollar Loan having an Interest Period longer than three months, if any, as agreed by the Borrower of such Loan and the Banks.

"Interest Period": with respect to any Eurodollar Loan: $% \left(1\right) =\left(1\right) \left(1$

(i) initially, the period commencing on the borrowing or conversion date, as the case may be, with respect to such Eurodollar Loan and ending one or two weeks, or one, two, three or six months thereafter, or such other period as may be requested by the Borrower and agreed to by the Banks making such Loan, as selected by the Borrower of such Loan in its notice of borrowing or notice of conversion, as the case may be, given with respect thereto; and

(ii) thereafter, each period commencing on the last day of the next preceding Interest Period applicable to such Eurodollar Loan and ending one or two weeks, or one, two, three or six months thereafter, or such other period as may be requested by the Borrower and agreed to by the Banks making such Loan, as selected by such Borrower by irrevocable notice to the Administrative Agent and the Banks which made such Eurodollar Loan not less than two Business Days prior to the last day of the then current Interest Period with respect thereto;

provided that, all of the foregoing provisions relating to Interest Periods are subject to the following:

- (1) if any Interest Period pertaining to a Eurodollar Loan would otherwise end on a day that is not a Business Day, such Interest Period shall be extended to the next succeeding Business Day unless the result of such extension would be to carry such Interest Period into another calendar month in which event such Interest Period shall end on the immediately preceding Business Day;
- $\mbox{\footnotemark}$ (2) any Interest Period that would otherwise extend beyond the Termination Date shall end on the Termination Date; and
- (3) any Interest Period pertaining to a Eurodollar Loan that begins on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall end on the last Business Day of a calendar month.

"Lien": any mortgage, pledge, hypothecation, assignment as security, security deposit arrangement, encumbrance, lien (statutory or other), conditional sale or other title retention agreement or other similar arrangement.

"Loan": any loan made by any Bank pursuant to this Agreement.

"Loan Documents": this Agreement, the Notes and the Notices of Additional Borrower.

"Loan Outstandings": as to any Bank at any time, the sum of (a) the aggregate principal amount of all Revolving Credit Loans made by such Bank then outstanding, and (b) such Bank's Commitment Percentage multiplied by the aggregate principal amount of all Bid Loans then outstanding.

"Loan Parties": the collective reference to the Company, the other Borrowers and the Parent.

"Majority Banks": at any time, Banks the Commitment Percentages of which aggregate (or, if at such time all of the Commitments shall have been terminated, Banks the Commitment Percentages of which immediately prior to such termination aggregated) at least 51%.

"Material Adverse Effect": a material adverse effect on (a) the business, operations, properties, or condition (financial or otherwise) of the Parent and its Subsidiaries taken as a whole, (b) the ability of the Company, or any Borrower or any other Loan Party to perform their respective obligations hereunder and under the other Loan Documents to which such Person is a party, or (c) the validity or

enforceability of the Loan Documents or the rights or remedies of the Administrative Agent or the Banks hereunder or thereunder.

"Moody's": Moody's Investors Services, Inc.

"Multiemployer Plan": a Plan which is a multiemployer plan as defined in Section 4001(a)(3) of ERISA.

"Net Sales and Revenues": with respect to any Person for any period, all sales and operating revenues of such Person during such period computed in accordance with GAAP after deducting therefrom sales returns, discounts and allowances.

"Notes": the collective reference to the Revolving Credit Notes and the Bid Loan Notes, if any.

"Notice of Additional Borrower": as defined in subsection 13.15(a).

"Obligations": the unpaid principal of and interest on (including, without limitation, interest accruing after the maturity of the Loans and interest accruing after the filing of any petition in bankruptcy, or the commencement of any insolvency, reorganization or like proceeding, relating to any of the Loan Parties or any of the Borrowers, whether or not a claim for post-filing or post-petition interest is allowed in such proceeding) the Loans and the Notes, if any, and all other obligations and liabilities of any of the Loan Parties or the Borrowers to the Administrative Agent or to the Banks, whether direct or indirect, absolute or contingent, due or to become due, or now existing or hereafter incurred, which may arise under, out of, or in connection with, this Agreement, the Notes, any other Loan Document and any other document made, delivered or given in connection herewith or therewith, whether on account of principal, interest, reimbursement obligations, fees, indemnities, costs, expenses (including, without limitation, all fees and disbursements of counsel to the Administrative Agent or to the Banks that are required to be paid by the Loan Parties and/or the Borrowers pursuant to the terms of this Agreement) or any other obligation hereunder or thereunder.

"Parent Guarantee": as defined in subsection 12.1.

"Participant": as defined in subsection 13.6(b).

"Payment Sharing Notice": a written notice from the Company or any Bank informing the Administrative Agent that an Event of Default has occurred and is continuing and directing the Administrative Agent to allocate payments thereafter received from the Borrower in accordance with subsection 5.9(c).

"PBGC": the Pension Benefit Guaranty Corporation established pursuant to Subtitle A of Title IV of ERISA.

"Person": an individual, partnership, corporation, business trust, joint stock company, trust, unincorporated association, joint venture, Governmental Authority or other entity of whatever nature.

"Plan": at a particular time, any employee benefit plan which is covered by ERISA and in respect of which the Company or a Commonly Controlled Entity is (or, if such plan were terminated at such time, would under Section 4069 of ERISA be deemed to be) an "employer" as defined in Section 3(5) of ERISA.

"Prepayment Date": as defined in subsection 5.3(b).

"Principal Subsidiary": (a) any Borrower and (b) any other Subsidiary if it shall have Total Assets at the end of the most recent fiscal year for which financial statements are required to have been furnished pursuant to subsection 8.1(a) in excess of \$75,000,000 or have had during such year Net Sales and Revenues in excess of \$75,000,000.

"Purchasing Banks": as defined in subsection 13.6(c).

"Register": as defined in subsection 13.6(d).

"Regulation U": Regulation U of the Board.

"Regulation X": Regulation X of the Board.

"Reorganization": with respect to any Multiemployer Plan, the condition that such plan is in reorganization within the meaning of Section 4241 of ERISA.

"Reportable Event": any of the events set forth in Section 4043(c) of ERISA, other than those events as to which the thirty day notice requirement is waived under PBGC Reg. ss. 4043.

"Requested Bank": as defined in subsection 3.1(a).

"Requested Termination Date": as defined in subsection 2.4(a).

"Requirement of Law": as to any Person, the Certificate of Incorporation and By-Laws or other organizational or governing documents of such Person, and any law, treaty, rule or regulation or determination of an arbitrator or a court or other Governmental Authority, in each case applicable to or binding upon such Person or any of its property or to which such Person or any of its property is subject.

"Responsible Officer": the chief executive officer, the president, the chief financial officer or the treasurer, assistant treasurer or controller of, respectively, the Parent and the Company.

"Restructuring Activities": all reductions in carrying value of assets or investments and provisions for the termination and/or relocation of operations and employees.

"Revolving Credit Loans": as defined in subsection 2.1(a).

"Revolving Credit Notes": as defined in subsection 2.2.

"SEC": the Securities and Exchange Commission, and any successor or analogous federal Governmental Authority.

"Single Employer Plan": any Plan which is covered by Title IV of ERISA, but which is not a Multiemployer Plan.

"S&P": Standard & Poor's Ratings Group.

"Subsidiary": as to any Person, a corporation, partnership or other entity which is required to be consolidated with such Person in accordance with GAAP; provided, that any such corporation, partnership or other entity which is controlled by a receiver or trustee under any bankruptcy, insolvency or similar law shall continue to be a "Subsidiary" of such Person for purposes of this Agreement. Unless otherwise qualified, all references to a "Subsidiary" or to "Subsidiaries" in this Agreement shall refer to a Subsidiary or Subsidiaries of the Parent.

"Substitute Rating Agency": as defined in the definition of "Applicable Margin".

"Terminating Banks": as defined in subsection 2.4(b).

"Termination Date": May 14, 1998 or such later date to which the Termination Date may be extended pursuant to subsection 2.4.

"Total Assets": with respect to any Person at any time, the total of all assets appearing on the asset side of the balance sheet of such Person prepared in accordance with GAAP as of such time.

"Transferee": as defined in subsection 13.6(f).

"Type": as to any Loan, its nature as an ABR Loan or a Eurodollar Loan. $\,$

 ${\hbox{1.2 Other Definitional Provisions.}} \quad \hbox{(a) Unless} \\ {\hbox{otherwise specified therein, all terms defined in this Agreement}} \\$

shall have the defined meanings when used in the Notes, if any, or any certificate or other document made or delivered pursuant hereto.

- (b) As used herein and in the Notes, if any, and any certificate or other document made or delivered pursuant hereto, accounting terms relating to the Parent and its Subsidiaries not defined in subsection 1.1, to the extent not defined, shall have the respective meanings given to them under GAAP.
- (c) The words "hereof", "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement, and Section, subsection, Schedule and Exhibit references are to this Agreement, unless otherwise specified.
- (d) The meanings given to terms defined herein shall be equally applicable to both the singular and plural forms of such terms.

SECTION 2. AMOUNT AND TERMS OF COMMITMENTS

- 2.1 Commitments. (a) Subject to the terms and conditions hereof, each Bank severally agrees to make revolving credit loans ("Revolving Credit Loans") to any Borrower from time to time during the Commitment Period in an aggregate principal amount at any one time outstanding not to exceed the amount of such Bank's Commitment, provided that no Bank shall make any Revolving Credit Loan if, after giving effect to such Loan, the aggregate Loan Outstandings of all of the Banks plus the Aggregate Outstanding Bilateral Option Loans would exceed the aggregate Commitments.
- (b) The Revolving Credit Loans may from time to time be (i) Eurodollar Loans, (ii) ABR Loans or (iii) a combination thereof, as determined by the Borrower thereof and notified to the Administrative Agent in accordance with subsections 2.3 and 5.5, provided that no Revolving Credit Loan shall be made as a Eurodollar Loan maturing after the Termination Date.
- 2.2 Obligations of Borrowers; Revolving Credit Notes. (a) Each Borrower agrees that each Revolving Credit Loan made by each Bank to such Borrower pursuant hereto shall constitute the promise and obligation of such Borrower to pay to the Administrative Agent, on behalf of such Bank, at the office of the Administrative Agent specified in subsection 13.2, in lawful money of the United States of America and in immediately available funds the aggregate unpaid principal amount of all Revolving Credit Loans made by such Bank to such Borrower pursuant to subsection 2.1, which amounts shall be due and payable (whether at maturity or by acceleration) as set forth in this Agreement and, in any event, on the Termination Date.

(b) Each Borrower agrees that each Bank and the Administrative Agent are authorized to record (i) the date, amount and Type of each Revolving Credit Loan made by such Bank to such Borrower pursuant to subsection 2.1, (ii) the date of each continuation thereof pursuant to subsection 5.5(b), (iii) the date of each conversion of all or a portion thereof to another Type pursuant to subsection 5.5(a), (iv) the date and amount of each payment or prepayment of principal of each such Revolving Credit Loan and (v) in the case of each such Revolving Credit Loan which is a Eurodollar Loan, the length of each Interest Period and the Eurodollar Rate with respect thereto, in the books and records of such Bank or the $\,$ Administrative Agent, as the case may be, and in such manner as is reasonable and customary for such Bank or the Administrative Agent, as the case may be, and a certificate of an officer of such Bank or the Administrative Agent, as the case may be, setting forth in reasonable detail the information so recorded, shall constitute prima facie evidence of the accuracy of the information so recorded; provided that the failure to make any such recording shall not in any way affect the obligations of such Borrower hereunder.

(c) Each Borrower agrees that, upon the request to the Administrative Agent by any Bank at any time, the Revolving Credit Loans made by such Bank to such Borrower shall be evidenced by a promissory note of such Borrower, substantially in the form of Exhibit A with appropriate insertions as to Borrower, payee, date and principal amount (a "Revolving Credit Note"), payable to the order of such Bank and in a principal amount equal to the lesser of (a) the amount of the initial Commitment of such Bank and (b) the aggregate unpaid principal amount of all Revolving Credit Loans made by such Bank to such Borrower. Upon the request to the Administrative Agent by any such Bank at any time, such Borrower shall execute and deliver to such Bank a Revolving Credit Note conforming to the requirements hereof and executed by a duly authorized officer of such Borrower. Each Bank is hereby authorized to record the date, Type and amount of each Revolving Credit Loan made by such Bank to such Borrower, each continuation thereof, each conversion of all or a portion thereof to another Type, the date and amount of each payment or prepayment of principal thereof and, in the case of Eurodollar Loans, the length of each Interest Period and the Eurodollar Rate with respect thereto, on the schedule annexed to and constituting a part of its Revolving Credit Note and any such recordation shall constitute prima facie evidence of the accuracy of the information so recorded; provided that the failure to make any such recording shall not in any way affect the obligations of such Borrower hereunder or thereunder. Each Revolving Credit Note shall (x) be dated the Closing Date, (y) be stated to mature on the Termination Date and (z) provide for the payment of interest in accordance with subsection 5.1.

2.3 Procedure for Revolving Credit Borrowing. Any Borrower may borrow under the Commitments from all Banks during the Commitment Period on any Business Day, provided that such

Borrower shall give the Administrative Agent irrevocable notice (which notice must be received by the Administrative Agent (a) prior to 4:00 P.M., New York City time, three Business Days prior to the requested Borrowing Date, if all or any part of the requested Revolving Credit Loans are to be initially Eurodollar Loans, or (b) prior to 10:00 A.M., New York City time, on the requested Borrowing Date, otherwise), specifying (i) the amount to be borrowed, (ii) the requested Borrowing Date, (iii) whether the borrowing is to be of Eurodollar Loans, ABR Loans or a combination thereof and (iv) if the borrowing is to be entirely or partly of Eurodollar Loans, the respective amounts of each such Type of Loan and the respective lengths of the initial Interest Periods therefor. Each borrowing under the Commitments shall be in an amount equal to \$5,000,000 or a whole multiple of \$1,000,000 in excess thereof (or, in the case of ABR Loans, if the amount of the Available Commitments minus the Aggregate Outstanding Bilateral Option Loans is less than \$5,000,000, such lesser amount). Upon receipt of such notice from such Borrower, the Administrative Agent shall promptly notify each Bank thereof. Each Bank will make the amount of its pro rata share of each such borrowing available to the Borrower at the office of the Administrative Agent specified in subsection 13.2 prior to 12:00 noon, New York City time, on the Borrowing Date requested by such Borrower in funds immediately available to the Administrative Agent. Such borrowing will then be made available to such Borrower on the books of such office with the aggregate of the amounts made available to the Administrative Agent by the Banks and in like funds as received by the Administrative Agent.

2.4 Extension of Termination Date. (a) The Company may request, in a notice given as herein provided to the Administrative Agent and each of the Banks not less than 45 days and not more than 60 days prior to the Termination Date then in effect ("Existing Termination Date"), that the Termination Date be extended, which notice shall specify a date (which shall be the Existing Termination Date) as of which the requested extension is to be effective (the "Effective Date"), and the new Termination Date to be in effect following such extension (the "Requested Termination Date"), which date shall be no more than 364 days after the effectiveness of such extension (with the Effective Date being counted as the first day). Each Bank shall, not later than a date 30 days prior to the Effective Date, notify the Company and the Administrative Agent of its election to extend or not to extend the Termination Date with respect to its Commitment. Notwithstanding any provision of this Agreement to the contrary, any notice by any Bank of its willingness to extend the Termination Date with respect to its Commitment shall be revocable by such Bank in its sole and absolute discretion at any time more than 30 days prior to the Effective Date. Any Bank which shall not timely notify the Company and the Administrative Agent of its election to extend the Termination Date shall be deemed to have elected not to extend the Termination Date with respect to its Commitment.

- (b) If any one or more Banks shall timely notify the Company and the Administrative Agent pursuant to paragraph (a) of this subsection 2.4 of their election not to extend their Commitments or shall be deemed to have elected not to extend their Commitments, (such Banks being called "Terminating Banks"), then the Company may (i) designate from the Banks other than Terminating Banks (the "Continuing Banks") one or more such Continuing Banks to increase their Commitments, which Continuing Banks shall have given notice to the Company and the Administrative Agent of their willingness to so increase their Commitments, (ii) with notice to the Administrative Agent, designate one or more other banking institutions willing to extend Commitments until the Requested Termination Date (any such banking institution, an "Additional Bank"), or (iii) any combination thereof, the aggregate amount of the increases of such Continuing Banks' Commitments and the amount of such Additional Banks' Commitments not to exceed the aggregate of the Commitments of the Terminating Banks. Any such increase in the Commitment of a Continuing Bank shall be evidenced by a written instrument executed by such Continuing Bank, the Company and the Administrative Agent, and shall take effect on the Existing Termination Date. Any Additional Bank shall, on the Existing Termination Date, execute and deliver to the Company and the Administrative Agent a "Commitment Transfer Supplement", satisfactory to the Company and the Administrative Agent, setting forth the amount of such Additional Bank's Commitment and containing its agreement to become, and to perform all the obligations of, a Bank hereunder, and the Commitment of such Additional Bank shall become effective on the Existing Termination Date.
- (c) The Company and each other Borrower, if any, shall deliver to each Continuing Bank and each Additional Bank which shall have requested Revolving Credit Notes pursuant to subsection 2.2(c), on the Existing Termination Date in exchange for each Revolving Credit Note, if any, of each of the Borrowers held by such Bank, new Revolving Credit Notes, maturing on the Requested Termination Date, in the principal amount of such Bank's Commitment after giving effect to the adjustments made pursuant to this subsection 2.4.
- (d) If some of or all the Banks shall have elected to extend their Commitments as provided in this subsection 2.4, then (i) the Commitments of the Continuing Banks and any Additional Banks shall continue until the Requested Termination Date specified in the notice from the Company, and as to such Banks the term "Termination Date", as used herein shall on and after the Effective Date mean such Requested Termination Date; (ii) the Commitments of the Terminating Banks shall continue until the Termination Date in effect prior to such extension, and shall then terminate, and as to the Terminating Banks, the term "Termination Date", as used herein, shall continue to mean such Existing Termination Date; and (iii) from and after the Termination Date in effect prior to such extension, the term "Banks" shall be deemed to include the Additional Banks.

SECTION 3. BILATERAL OPTION LOANS

- 3.1 Requests for Offers. (a) From time to time during the period from the Closing Date until the Termination Date, any Borrower may request any or all of the Banks (each such Bank to which such a request is made, a "Requested Bank") to make offers to make Bilateral Option Loans, provided that immediately after making any such Bilateral Option Loan, the aggregate Loan Outstandings of all the Banks plus the Aggregate Outstanding Bilateral Option Loans will not exceed the aggregate Commitments. Any such request shall specify the principal amount and maturity date of the Bilateral Option Loans for which such Borrower is requesting offers, whether such Bilateral Option Loans are requested to be Dollar Bilateral Loans or Alternative Currency Bilateral Loans, the time by which offers to make such Bilateral Option Loans must be made by such Requested Bank and by which such offers shall be accepted or rejected by such Borrower, and if all or any part of the requested Bilateral Option Loans are requested to be made as Alternative Currency Bilateral Loans, the Alternative Currency to be applicable thereto. Each Requested Bank may, but shall have no obligation to, make such offers on such terms and conditions as are satisfactory to such Requested Bank, and such Borrower may, but shall have no obligation to, accept any such offers. No Bilateral Option Loan may mature after the Termination Date.
- (b) Each Borrower and Requested Bank shall separately agree as to the procedures, documentation, lending office and other matters relating to any Bilateral Option Loan.
- 3.2 Reports to Administrative Agent; Determination of Dollar Equivalents. (a) The Borrower shall deliver to the Administrative Agent a report in respect of each Bilateral Option Loan (a "Bilateral Option Loan Report") by 2:00 P.M. (New York City time) on the date on which the applicable Borrower accepts any Bilateral Option Loan, on the date on which any principal amount thereof is repaid prior to the scheduled maturity date, or on the scheduled maturity date if payment thereof is not made on such scheduled maturity date, specifying for such Bilateral Option Loan the date on which such Bilateral Option Loan was or will be made, such amount of principal is or will be repaid or such payment was not made as the case may be; in the case of Alternative Currency Bilateral Loans, the Alternative Currency thereof; and the principal amount of such Bilateral Option Loan or principal prepayment or repayment or the amount paid (in the case of any Alternative Currency Bilateral Loan, expressed in the Alternative Currency therefor).
- (b) Upon receipt of a Bilateral Option Loan Report with respect to the acceptance of a Bilateral Option Loan, the Administrative Agent shall determine the Dollar Equivalent thereof.
- (c) If on any Borrowing Date on which after giving effect to the Loans made on such date, the sum of the aggregate $\,$

Loan Outstandings of all the Banks plus the Aggregate Outstanding Bilateral Option Loans exceeds 85% of the aggregate Commitments, then the Administrative Agent shall redetermine as of such Borrowing Date, on the basis of the most recently delivered Bilateral Option Loan Report for each Bilateral Option Loan, the Dollar Equivalent of each Alternative Currency Bilateral Loan then outstanding. In addition, for so long as the condition specified in the preceding sentence remains in effect, the Administrative Agent shall determine, at the end of each fiscal quarter of the Company, on the basis of the most recently delivered Bilateral Option Loan Report for each Bilateral Option Loan, the Dollar Equivalent of each Alternative Currency Bilateral Loan then outstanding.

- (d) The Administrative Agent shall promptly notify the Company of each Dollar Equivalent under this subsection 3.2.
- 3.3 Judgment Currency. If for the purpose of obtaining judgment in any court, it is necessary to convert a sum due from any Borrower hereunder or under any of the Notes in the currency expressed to be payable herein or under the Notes (the "specified currency") into another currency, the parties hereto agree, to the fullest extent that they may effectively do so, that the rate of exchange used shall be that at which in accordance with normal banking procedures the Administrative Agent could purchase the specified currency with such other currency at the Administrative Agent's New York office on the Business Day preceding that on which final judgment is given. The obligations of each Borrower in respect of any sum due to any Bank or the Administrative Agent hereunder or under any Note shall, notwithstanding any judgment in a currency other than the specified currency, be discharged only to the extent that on the Business Day following receipt by such Bank or the Administrative Agent (as the case may be) of any sum adjudged to be so due in such other currency such Bank or the Administrative Agent (as the case may be) may in accordance with normal banking procedures purchase the specified currency with such other currency; if the amount of the specified currency so purchased is less than the sum originally due to such Bank or the Administrative Agent, as the case may be, in the specified currency, each Borrower agrees, to the fullest extent that it may effectively do so, as a separate obligation and notwithstanding any such judgment, to indemnify such Bank or the Administrative Agent, as the case may be, against such difference, and if the amount of the specified currency so purchased exceeds:
 - (a) the sum originally due to any Bank or the Administrative Agent, as the case may be, and
 - (b) any amounts shared with other Banks as a result of allocations of such excess as a disproportionate payment to such Bank under subsection 13.7,

such Bank or the Administrative Agent, as the case may be, agrees to remit such excess to the applicable Borrower.

3.4 Repayments. Each Borrower shall repay to each Bank which has made a Bilateral Option Loan on the maturity date of each Bilateral Option Loan (such maturity date being that specified in the documentation referred to in subsection 3.1(a)) the then unpaid principal amount of such Bilateral Option Loan.

SECTION 4. BID LOANS

- 4.1 The Bid Loans. Any Borrower may borrow Bid Loans from time to time on any Business Day during the period from the Closing Date until the Termination Date, in the manner set forth in this Section 4 and in amounts such that the aggregate Loan Outstandings of all the Banks at any time plus the Aggregate Outstanding Bilateral Option Loans at such time will not exceed the aggregate Commitments at such time, and provided, further, that no such Bid Loan shall be made if, after giving effect thereto, any Bid Loans would mature after the Termination Date.
- 4.2 Procedure for Bid Loans. (a) A Borrower shall request Bid Loans by delivering a Bid Loan Request to the Administrative Agent, in writing, by facsimile transmission, or by telephone, confirmed by facsimile transmission, not later than 1:00 P.M. (New York City time) one Business Day prior to the proposed Borrowing Date. Each Bid Loan Request may solicit bids for Bid Loans in an aggregate principal amount of \$5,000,000 or a whole multiple of \$1,000,000 in excess thereof and for not more than three alternative maturity dates for such Bid Loans. The Administrative Agent shall promptly notify each Bank by facsimile transmission of the contents of each Bid Loan Request received by it.
- (b) Upon receipt of notice from the Administrative Agent of the contents of a Bid Loan Request, any Bank that elects, in its sole discretion, to do so, shall irrevocably offer to make one or more Bid Loans at a rate of interest determined by such Bank in its sole discretion for each such Bid Loan. Any such irrevocable offer shall be made by delivering a Bid Loan Offer to the Administrative Agent, by telephone, immediately confirmed by facsimile transmission, before 9:30 A.M. (New York City time) on the proposed Borrowing Date, setting forth the maximum amount of Bid Loans for each maturity date, and the aggregate maximum amount for all maturity dates, which such Bank would be willing to make (which amounts may, subject to subsection 4.1, exceed such Bank's Commitments) and the rate of interest at which such Bank is willing to make each such Bid Loan; the Administrative Agent shall advise the Borrower before 10:00 A.M. (New York City time) on the proposed Borrowing Date of the contents of each such Bid Loan Offer received by it. If the Administrative Agent in its capacity as a Bank shall, in its sole discretion, elect to make any such offer, it shall advise the Borrower of the contents of its Bid Loan Offer before 9:15 A.M. (New York City time) on the proposed Borrowing Date.

- (c) The Borrower shall before 10:30 A.M. (New York City time) on the proposed Borrowing Date, in its absolute discretion, either:
 - (i) cancel such Bid Loan Request by giving the Administrative Agent telephone notice to that effect, and the Administrative Agent shall give prompt telephone notice thereof to the Banks and the Bid Loans requested thereby shall not be made: or
 - (ii) accept one or more of the offers made by any Bank or Banks by giving telephone notice to the Administrative Agent (confirmed as soon as practicable thereafter by delivery to the Administrative Agent of a Bid Loan Confirmation in writing or by facsimile transmission) of the amount of Bid Loans for each relevant maturity date to be made by each Bank (which amount for each such maturity date shall be equal to or less than the maximum amount for such maturity date specified in the Bid Loan Offer of such Bid Loan Bank, and for all maturity dates included in such Bid Loan Offer shall be equal to or less than the aggregate maximum amount specified in such Bid Loan Offer for all such maturity dates) and reject any remaining offers made by Banks; provided, however, that (x) the Borrower may not accept offers for Bid Loans for any maturity date in an aggregate principal amount in excess of the maximum principal amount requested in the related Bid Loan Request, (y) if the Borrower accepts any of such offers, it must accept offers strictly based upon pricing for such relevant maturity date and no other criteria whatsoever and (z) if two or more Banks submit offers for any maturity date at identical pricing and the Borrower accepts any of such offers but does not wish to (or by reason of the limitations set forth in subsection 4.1 or in clause (x) of this proviso, cannot) borrow the total amount offered by such Banks with such identical pricing, the Borrower shall accept offers from all of such Banks in amounts allocated among them pro rata according to the amounts offered by such Banks.
- (d) If the Borrower accepts pursuant to clause (c) (ii) above one or more of the offers made by any Bid Loan Bank or Bid Loan Banks, the Administrative Agent shall notify before 11:00 A.M. (New York City time) each Bid Loan Bank which has made such an offer, of the aggregate amount of such Bid Loans to be made on such Borrowing Date for each maturity date and of the acceptance or rejection of any offers to make such Bid Loans made by such Bid Loan Bank. Each Bid Loan Bank which is to make a Bid Loan shall, before 12:00 Noon (New York City time) on the

Borrowing Date specified in the Bid Loan Request applicable thereto, make available to the Administrative Agent at its office set forth in subsection 13.2 the amount of Bid Loans to be made by such Bid Loan Bank, in immediately available funds. The Administrative Agent will make such funds available to the Borrower at or before 2:00 P.M. (New York City time) on such date at the Administrative Agent's aforesaid address. As soon as practicable after each Borrowing Date, the Administrative Agent shall notify each Bank of the aggregate amount of Bid Loans advanced on such Borrowing Date and the respective maturity dates thereof.

- 4.3 Repayments. Each Borrower shall repay to the Administrative Agent for the account of each Bid Loan Bank which has made a Bid Loan on the maturity date of each Bid Loan (such maturity date being that specified by the Borrower for repayment of such Bid Loan in the related Bid Loan Request) the then unpaid principal amount of such Bid Loan. The Borrowers shall not have the right to prepay any principal amount of any Bid Loan without the prior written consent of the Bid Loan Bank which made such Bid Loan.
- 4.4 Interest on Bid Loans. Each Borrower which shall have borrowed a Bid Loan shall pay interest on the unpaid principal amount of such Bid Loan from the Borrowing Date to the stated maturity date thereof, at the rate of interest determined pursuant to subsection 4.2 above (calculated on the basis of a 360 day year for actual days elapsed), payable on the interest payment date or dates specified by such Borrower for such Bid Loan in the related Bid Loan Request. If all or a portion of the principal amount of any Bid Loan shall not be paid when due (whether at the stated maturity, by acceleration or otherwise), such overdue principal amount shall, without limiting any rights of any Bank under this Agreement, bear interest from the date on which such payment was due at a rate per annum which is 2% above the rate which would otherwise be applicable to such Bid Loan until the scheduled maturity date with respect thereto, and for each day thereafter at a rate per annum which is 2% above the Alternate Base Rate until paid in full (as well after as before judgment).
- 4.5 Obligations of Borrowers; Bid Loan Notes. (a) Each Borrower agrees that each Bid Loan made by each Bid Loan Bank to such Borrower pursuant hereto shall constitute the promise and obligation of such Borrower to pay to the Administrative Agent, on behalf of such Bid Loan Bank, at the office of the Administrative Agent specified in subsection 13.2, in lawful money of the United States of America and in immediately available funds the aggregate unpaid principal amount of each Bid Loan made by such Bid Loan Bank to such Borrower pursuant to subsection 4.2, which amounts shall be due and payable (whether at maturity or by acceleration) as set forth in the Bid Loan Request related to such Bid Loan and in this Agreement.

(b) Each Borrower agrees that each Bid Loan Bank and the Administrative Agent are authorized to record (i) the date and amount of each Bid Loan made by such Bid Loan Bank to such Borrower pursuant to subsection 4.2, and (ii) the date and amount of each payment or prepayment of principal of each such Bid Loan, in the books and records of such Bid Loan Bank or the Administrative Agent, as the case may be, and in such manner as is reasonable and customary for such Bank or the Administrative Agent, as the case may be, and a certificate of an officer of such Bid Loan Bank or the Administrative Agent, as the case may be, setting forth in reasonable detail the information so recorded, shall constitute prima facie evidence of the accuracy of the information so recorded; provided that the failure to make any such recording shall not in any way affect the obligations of such Borrower hereunder.

(c) Each Borrower agrees that, upon the request to the Administrative Agent by any Bid Loan Bank at any time, the Bid Loans made by such Bid Loan Bank to any Borrower shall be evidenced by a promissory note of such Borrower, substantially in the form of Exhibit B with appropriate insertions (a "Bid Loan Note"), payable to the order of such Bid Loan Bank and representing the obligation of such Borrower to pay the unpaid principal amount of all Bid Loans made by such Bid Loan Bank, with interest on the unpaid principal amount from time to time outstanding of each Bid Loan evidenced thereby as prescribed in subsection 4.4. Upon the request to the Administrative Agent by any such Bid Loan Bank at any time, such Borrower shall execute and deliver to such Bid Loan Bank a Bid Loan Note conforming to the requirements hereof and executed by a duly authorized officer of such Borrower. Each Bid Loan Bank is hereby authorized to record the date and amount of each Bid Loan made by such Bank, the maturity date thereof, the date and amount of each payment of principal thereof and the interest rate with respect thereto on the schedule annexed to and constituting part of its Bid Loan Note, and any such recordation shall constitute prima facie evidence of the accuracy of the information so recorded; provided, however, that the failure to make any such recordation shall not affect the obligations of such Borrower hereunder or under any Bid Loan Note. Each Bid Loan Note shall be dated the Closing Date and each Bid Loan evidenced thereby shall bear interest for the period from and including the Borrowing Date thereof on the unpaid principal amount thereof from time to time outstanding at the applicable rate per annum determined as provided in, and such interest shall be payable as specified in, subsection 4.4.

SECTION 5. LOAN FACILITY COMMON PROVISIONS

5.1 Interest Rates and Payment Dates. (a) Each Eurodollar Loan shall bear interest for each day during each Interest Period with respect thereto at a rate per annum equal to the Eurodollar Rate determined for such Interest Period plus the Applicable Margin.

- (c) Except as otherwise provided in subsection 4.4, if all or a portion of (i) the principal amount of any Loan or (ii) any interest payable thereon shall not be paid when due (whether at the stated maturity, by acceleration or otherwise), such overdue amount shall bear interest at a rate per annum which is (x) in the case of overdue principal, the rate that would otherwise be applicable thereto pursuant to the foregoing provisions of this subsection plus 2% or (y) in the case of overdue interest, the rate described in paragraph (b) of this subsection plus 2%, in each case from the date of such non-payment until such amount is paid in full (as well after as before judament).
- (d) Interest shall be payable in arrears on each Interest Payment Date, provided that interest accruing pursuant to paragraph (c) of this subsection shall be payable on demand.
- (e) Subject to the limitations set forth herein, each Borrower may use the Loans by borrowing, prepaying and reborrowing the Loans, all in accordance with the terms and conditions hereof.
- 5.2 Facility Fee. (a) The Company agrees to pay to the Administrative Agent for the account of each Bank a facility fee for the period from and including the date hereof to the Termination Date, computed at the rate per annum determined as set forth in paragraph (b) of this subsection on the average daily amount of the Commitment of such Bank during the period for which payment is made, payable quarterly in arrears on the fifteenth day of each March, June, September and December and on the Termination Date or such earlier date as the Commitments shall terminate as provided herein, commencing on the first of such dates to occur after the date hereof.
- (b) The rate per annum at which such facility fee under paragraph (a) above shall be computed (the "Applicable Facility Fee Rate"), for any day on which the long term senior unenhanced, unsecured debt of the Company is rated by both S&P and Moody's, shall be the rate per annum under the caption "Facility Fee Rate" (a "Facility Fee Rate") set forth below opposite the S&P and Moody's ratings applicable to such debt on such day (or, if such ratings are set opposite two different rates under said caption, then the Applicable Facility Fee Rate shall be the lower of said two Facility Fee Rates):

FACILITY FEE RATE	S&P	MOODY'S
.1500%	BB+ or lower	Ba1 or lower
.1000%	BBB-	ВааЗ
. 0850%	BBB	Baa2
.0650%	BBB+	Baa1
. 0550%	A- or higher	A3 or higher

provided that if on any day the long term senior unenhanced, unsecured debt of the Company is rated by only one of S&P or Moody's, such rate will be determined based on the rating by such rating agency, and provided, further, that if on any day the long term senior unenhanced, unsecured debt of the Company is rated by neither S&P nor Moody's, the Applicable Facility Fee Rate will be determined based on the rating of such debt by a Substitute Rating Agency and will be the Facility Fee Rate set forth above opposite the S&P and Moody's ratings comparable to the Substitute Rating Agency's rating of such debt on such date, and provided, further, that if on any day the long term senior unenhanced, unsecured debt of the Company is rated by none of S&P, Moody's or any Substitute Rating Agency, the Company, the Administrative Agent and the Banks will negotiate in good faith to determine an alternative basis for calculating such rate consistent with the table set forth above and, if agreement on such alternative basis is not reached with 30 days, such rate will be calculated on an alternative basis determined by the Administrative Agent and the Banks in their reasonable discretion consistent with the table above, and until such alternative basis is determined such rate will be the rate last determined as provided in the table above.

5.3 Termination or Reduction of Commitments; Change of Control Date. (a) The Company shall have the right, upon not less than five Business Days' notice to the Administrative Agent, to terminate the Commitments or, from time to time, to reduce the amount of the Commitments, provided that no such termination or reduction shall be permitted to the extent that, after giving effect thereto and to any prepayments of Loans made on the effective date thereof, the sum of the aggregate Loan Outstandings of all the Banks, plus the Aggregate Outstanding Bilateral Option Loans would exceed the Commitments then in effect. Any such partial reduction shall be in an amount equal to \$5,000,000 or a whole multiple of \$1,000,000 in excess thereof and shall reduce permanently the Commitments then in effect.

(b) (i) In the event that a Change of Control Date shall occur, (A) the Company shall, within 10 days after such Change of Control Date, give each Bank notice thereof in writing describing in reasonable detail the facts and circumstances giving rise thereto, and (B) such Bank, by written notice given

to the Company not later than 30 days after the Change of Control Date, may declare the Commitments of such Bank to be terminated in full or reduced as of the date of (or as of a later date specified in) such notice to the Company, and may require that the Borrowers prepay as provided in this subsection 5.3 any Loans payable to such Bank and outstanding on such date to the extent the principal amount thereof exceeds such Bank's Commitment, if any, remaining after such termination or reduction. To the extent such Bank so requires, the Borrowers shall prepay such Loans on the 75th day after the date of the Company's notice or, in the event such 75th day is not a Business Day, the Business Day next succeeding such 75th day ("Prepayment Date").

(ii) On the Prepayment Date, the Borrowers shall prepay the unpaid principal amount of the Loans payable to such Bank, without premium or penalty, together with accrued interest on the amount prepaid to the Prepayment Date.

(iii) Subsections 5.9(a), (b) and (c) shall not apply to prepayments under this subsection 5.3(b).

(iv) Paragraph (a) of this subsection 5.3 hereof shall not apply to any Commitment reductions pursuant to this paragraph (b).

(v) In the event that a Change of Control Date shall occur, the Company shall not thereafter, without the prior written consent of the Majority Banks, borrow any additional Loan (other than a Bilateral Option Loan) in order to make, directly or indirectly, any payment or prepayment on any Indebtedness subordinated as to the payment of principal and interest or on liquidation to the prior payment of any of the Obligations.

5.4 Prepayments. (a) Any Borrower may at any time and from time to time upon at least four Business Days' irrevocable notice to the Administrative Agent, in the case of Eurodollar Loans, or upon at least one Business Day's irrevocable notice to the Administrative Agent, in the case of ABR Loans, prepay the Loans (other than Bid Loans), in whole or in part, without premium or penalty (subject to subsection 5.13), specifying the date and amount of prepayment and whether the prepayment is of Eurodollar Loans, ABR Loans or a combination thereof, and, if of a combination thereof, the amount allocable to each. Upon receipt of any such notice the Administrative Agent shall promptly notify each Bank thereof. If any such notice is given, the amount specified in such notice shall be due and payable on the date specified therein. Partial prepayments shall be in an aggregate principal amount of \$5,000,000 or a whole multiple of \$1,000,000 in excess thereof.

(b) If at any time, the Administrative Agent shall determine (which determination shall be conclusive in the absence of manifest error) that the sum of the aggregate Loan Outstandings of all the Banks plus the Aggregate Outstanding Bilateral Option Loans exceeds the aggregate Commitments, the

Borrowers shall immediately prepay the Loans in an aggregate principal amount equal to such excess.

5.5 Conversion and Continuation Options. (a) Any Borrower may elect at any time and from time to time (subject to subsection 5.13) to convert its Eurodollar Loans to ABR Loans by giving the Administrative Agent at least two Business Days' prior irrevocable notice of such election. Any Borrower may elect at any time and from time to time to convert its ABR Loans to Eurodollar Loans by giving the Administrative Agent irrevocable notice of such election (which notice must be received by the Administrative Agent prior to 4:00 P.M., New York City time, three Business Days prior to the requested conversion date). Any such notice of conversion to Eurodollar Loans shall specify the length of the initial Interest Period or Interest Periods therefor. Upon receipt of any such notice the Administrative Agent shall promptly notify each Bank thereof. All or any part of outstanding Eurodollar Loans and ABR Loans may be converted as provided herein, provided that (i) no Loan may be converted into a Eurodollar Loan when any Event of Default has occurred and is continuing and the Administrative Agent or the Majority Banks have determined that such a conversion is not appropriate, and (ii) any such conversion may only be made if, after giving effect thereto, subsection 5.6 shall not have been contravened.

(b) Any Eurodollar Loans may be continued as such upon the expiration of the then current Interest Period with respect thereto by the Borrower thereof giving notice to the Administrative Agent, in accordance with the applicable provisions of the term "Interest Period" set forth in subsection 1.1, of the length of the next Interest Period to be applicable to such Loans, provided that no Eurodollar Loan may be continued as such (i) when any Event of Default has occurred and is continuing and the Administrative Agent or the Majority Banks have determined that such a continuation is not appropriate, or (ii) if, after giving effect thereto, subsection 5.6 would be contravened and provided, further, that if any Borrower shall fail to give any required notice as described above in this paragraph or if such continuation is not permitted pursuant to the preceding proviso such Loans shall be automatically converted to ABR Loans on the last day of such then expiring Interest Period.

5.6 Minimum Amounts of Eurodollar Tranches. All borrowings, conversions and continuations of Loans hereunder and all selections of Interest Periods hereunder shall be in such amounts and be made pursuant to such elections so that, after giving effect thereto, the aggregate principal amount of the Loans comprising each Eurodollar Tranche shall be equal to \$5,000,000 or a whole multiple of \$1,000,000 in excess thereof.

5.7 Computation of Interest and Fees. (a) Interest on ABR Loans and facility fees shall be calculated on the basis of a 365- (or 366-, as the case may be) day year for the actual days elapsed. Interest on Eurodollar Loans shall be calculated on the

basis of a 360-day year for the actual days elapsed. The Administrative Agent shall as soon as practicable notify the Borrowers and the Banks of each determination of a Eurodollar Rate. Any change in the interest rate on a Loan resulting from a change in the Prime Rate shall become effective as of the opening of business on the day on which such change in the Prime Rate is announced. The Administrative Agent shall as soon as practicable notify the Borrowers and the Banks of the effective date and the amount of each such change in interest rate.

(b) Each determination of an interest rate by the Administrative Agent pursuant to any provision of this Agreement shall be conclusive and binding on the Borrowers and the Banks in the absence of manifest error. The Administrative Agent shall, at the request of the Company, deliver to the Company a statement showing in reasonable detail the quotations and calculations used by the Administrative Agent in determining any interest rate pursuant to subsections 5.1 and 5.7(a).

- 5.8 Inability to Determine Interest Rate. In the event that prior to the first day of any Interest Period:
 - (a) the Administrative Agent shall have determined (which determination shall be conclusive and binding upon the Borrowers) that, by reason of circumstances affecting the relevant market, adequate and reasonable means do not exist for ascertaining the Eurodollar Rate for such Interest Period, or
 - (b) the Administrative Agent shall have received notice from the Majority Banks that the Eurodollar Rate determined or to be determined for such Interest Period will not adequately and fairly reflect the cost to such Banks (as conclusively certified by such Banks) of making or maintaining their affected Loans during such Interest Period,

the Administrative Agent shall give telecopy or telephonic notice thereof to the Borrowers and the Banks as soon as practicable thereafter. If such notice is given (x) any Eurodollar Loans requested to be made on the first day of such Interest Period shall be made as ABR Loans, (y) any Loans that were to have been converted on the first day of such Interest Period to Eurodollar Loans shall be continued as ABR Loans and (z) any outstanding Eurodollar Loans shall be converted, on the first day of such Interest Period, to ABR Loans. Until such notice has been withdrawn by the Administrative Agent, no further Eurodollar Loans shall be made or continued as such, nor shall any Borrower have the right to convert Loans to Eurodollar Loans.

5.9 Pro Rata Treatment and Payments. (a) Each borrowing by any Borrower of Revolving Credit Loans from the Banks hereunder, each payment by the Company on account of any facility fee hereunder, and any reduction of the Commitments of

the Banks shall be made pro rata according to the respective Commitment Percentages of the Banks.

- (b) Whenever any payment received by the Administrative Agent or any Bank under this Agreement or any Note is insufficient to pay in full all amounts then due and payable to the Administrative Agent and the Banks under this Agreement and the Notes, and the Administrative Agent has not received a Payment Sharing Notice (or if the Administrative Agent has received a Payment Sharing Notice but the Event of Default specified in such Payment Sharing Notice has been cured or waived), such payment shall be distributed and applied by the Administrative Agent and the Banks in the following order: first, to the payment of fees and expenses due and payable to the Administrative Agent in its capacity as Administrative Agent under and in connection with this Agreement; second, to the payment of all expenses due and payable under subsection 13.5, ratably among the Banks in accordance with the aggregate amount of such payments owed to each such Bank; third, to the payment of fees due and payable under subsections 5.2(a) and (b), ratably among the Banks in accordance with their Commitment Percentages; fourth, to the payment of interest then due and payable on the Loans, ratably among the Banks in accordance with the aggregate amount of interest owed to each such Bank; and fifth, to the payment of the principal amount of the Loans which is then due and payable, ratably among the Banks in accordance with the aggregate principal amount owed to each such Bank.
- (c) After the Administrative Agent has received a Payment Sharing Notice which remains in effect, all payments received by the Administrative Agent under this Agreement or any Note shall be distributed and applied by the Administrative Agent and the Banks in the following order: first, to the payment of all amounts described in clauses first through third of the foregoing paragraph (b), in the order set forth therein; and second, to the payment of the interest accrued on and the principal amount of all of the Loans, regardless of whether any such amount is then due and payable, ratably among the Banks in accordance with the aggregate accrued interest plus the aggregate principal amount owed to such Bank.
- (d) All payments (including prepayments) to be made by any Borrower hereunder and under the Notes, whether on account of principal, interest, fees or otherwise, shall be made without set-off or counterclaim and shall be made prior to 3:00 P.M., New York City time, on the due date thereof (i) in the case of fees and Loans other than Bilateral Option Loans, to the Administrative Agent, for the account of the Banks, at the Administrative Agent's office specified in subsection 13.2, and (ii) in the case of Bilateral Option Loans made by any Bank, to such Bank, at the Bank's office specified in Schedule I (or, with respect to Alternative Currency Bilateral Loans, if different, at such other office of the Bank that it shall designate), in each case in Dollars (or, with respect to Alternative Currency Bilateral Loans, in the relevant Alternative Currency) and in

immediately available funds. If any payment hereunder becomes due and payable on a day other than a Business Day, such payment shall be extended to the next succeeding Business Day (unless, with respect to any payment on a Eurodollar Loan, the result of such extension would be to extend such payment into another calendar month, in which event such payment shall be made on the immediately preceding Business Day), and, with respect to payments of principal, interest thereon shall be payable at the then applicable rate during such extension.

(e) Unless the Administrative Agent shall have been notified in writing by the Bank prior to a Borrowing Date that such Bank will not make the amount of any Loan it has committed to make on such date available to the Administrative Agent, the Administrative Agent may assume that such Bank has made such amount available to the Administrative Agent on such Borrowing Date, and the Administrative Agent may, in reliance upon such assumption, make available to the applicable Borrower a corresponding amount. If such amount is made available to the Administrative Agent on a date after such Borrowing Date, such Bank shall pay to the Administrative Agent on demand an amount equal to the product of (i) the daily average Federal Funds Effective Rate during such period, times (ii) the amount of the Loan such Bank was committed to make, times (iii) a fraction the numerator of which is the number of days that elapse from and including such Borrowing Date to the date on which such Bank's Loan shall have become immediately available to the Administrative Agent and the denominator of which is 360. A certificate of the Administrative Agent submitted to any Bank with respect to any amounts owing under this subsection shall be conclusive in the absence of manifest error. If such Bank's Commitment Percentage of such borrowing is not in fact made available to the Administrative Agent by such Bank within three Business Days of such Borrowing Date, the Administrative Agent shall be entitled to recover such amount with interest thereon at the rate per annum applicable to ABR Loans hereunder, on demand, from such Borrower.

5.10 Illegality. Notwithstanding any other provision herein, if any change after the date hereof in any Requirement of Law or in the interpretation or application thereof shall make it unlawful for any Bank to make or maintain Eurodollar Loans as contemplated by this Agreement, (a) the commitment of such Bank hereunder to make Eurodollar Loans, continue Eurodollar Loans as such and convert ABR Loans to Eurodollar Loans shall forthwith be cancelled and (b) such Bank's Loans then outstanding as Eurodollar Loans, if any, shall be converted automatically to ABR Loans on the respective last days of the then current Interest Periods with respect to such Loans or within such earlier period as required by law. If any such conversion of a Eurodollar Loan occurs on a day which is not the last day of the then current Interest Period with respect thereto, the Borrower of such Loan shall pay to such Bank such amounts, if any, as may be required pursuant to subsection 5.13.

5.11 Requirements of Law. (a) In the event that any change after the date hereof in any Requirement of Law or in the interpretation or application thereof or compliance by any Bank with any request or directive (whether or not having the force of law) from any central bank or other Governmental Authority made subsequent to the date hereof:

- (i) shall subject any Bank to any tax of any kind whatsoever with respect to this Agreement, any Note or any Eurodollar Loan made by it, or change the basis of taxation of payments to such Bank in respect thereof (except for taxes covered by subsection 5.12 and changes in taxes based upon or measured by income of such Bank);
- (ii) shall impose, modify or hold applicable any reserve, special deposit, compulsory loan or similar requirement against assets held by, deposits or other liabilities in or for the account of, advances, loans or other extensions of credit by, or any other acquisition of funds by, any office of such Bank which is not otherwise included in the determination of the Eurodollar Rate hereunder; or

(iii) shall impose on such Bank any other condition; and the result of any of the foregoing is to increase the cost to such Bank, by an amount which such Bank deems in its reasonable judgment to be material, of making, converting into, continuing or maintaining Eurodollar Loans or to reduce any amount receivable hereunder in respect thereof then, in any such case, the Company shall promptly pay such Bank, upon its demand, any additional amounts necessary to compensate such Bank for such increased cost or reduced amount receivable. If any Bank becomes entitled to claim any additional amounts pursuant to this subsection, it shall promptly notify the Company, through the Administrative Agent, of the event by reason of which it has become so entitled. A certificate as to any additional amounts payable pursuant to this subsection setting forth the calculation thereof in reasonable detail (as determined by such Bank in its reasonable discretion) submitted by such Bank, through the Administrative Agent, to the Company shall be conclusive in the absence of manifest error. This covenant shall survive the termination of this Agreement and the payment of the Notes and all other amounts payable hereunder.

(b) In the event that any Bank shall have determined that any change in any Requirement of Law regarding capital adequacy or in the interpretation or application thereof or compliance by such Bank or any corporation controlling such Bank with any request or directive regarding capital adequacy (whether or not having the force of law) from any Governmental Authority made subsequent to the date hereof does or shall have the effect of reducing the rate of return on such Bank's or such corporation's capital as a consequence of its obligations hereunder to a level below that which such Bank or such

corporation could have achieved but for such change or compliance (taking into consideration such Bank's or such corporation's policies with respect to capital adequacy) by an amount deemed by such Bank to be material, then from time to time, after submission by such Bank to the Company (with a copy to the Administrative Agent) of a written request therefor, the Company shall pay to such Bank such additional amount or amounts as will compensate such Bank for such reduction. A certificate as to any additional amount payable pursuant to this subsection setting forth the calculation thereof in reasonable detail (as determined by such Bank in its reasonable discretion) through the Administrative Agent, to the Company shall be conclusive in the absence of manifest error.

(c) Upon request by any Bank, through the Administrative Agent, from time to time, the Borrowers shall pay the cost of all Eurocurrency Reserve Requirements applicable to the Eurodollar Loans made by such Bank. If a Bank is or becomes entitled to receive payments in respect of Eurocurrency Reserve Requirements, pursuant to this subsection 5.11(c), it shall promptly notify the Borrowers thereof through the Administrative Agent. A certificate as to the amount of such Eurocurrency Reserve Requirements setting forth the calculation thereof in reasonable detail (as determined by such Bank in its reasonable discretion) submitted by such Bank, through the Administrative Agent, to the Borrowers shall be conclusive in the absence of manifest error. This covenant shall survive the termination of this Agreement and the payment of the Loans and all other amounts payable hereunder.

 $\,$ 5.12 Taxes. (a) All payments made by any Borrower under this Agreement and any Notes shall be made free and clear of, and without deduction or withholding for or on account of, any present or future income, stamp or other taxes, levies, imposts, duties, charges, fees, deductions or withholdings, now or hereafter imposed, levied, collected, withheld or assessed by any Governmental Authority, excluding, in the case of the Administrative Agent and each Bank, net income taxes and franchise taxes (imposed in lieu of net income taxes) that would not have been imposed on the Administrative Agent or such Bank, as the case may be, in the absence of a present or former connection between the jurisdiction of the government or taxing authority imposing such tax and the Administrative Agent or such Bank (other than a connection arising solely from the Administrative Agent or such Bank having executed, delivered or performed its obligations or received a payment under, or enforced, this Agreement or any Notes) or any political subdivision or taxing authority thereof or therein (all such non-excluded taxes, levies, imposts, duties, charges, fees, deductions and withholdings being hereinafter called "Taxes"). If any Taxes are required to be withheld from any amounts payable to the Administrative Agent or any Bank hereunder or under any Notes, the amounts so payable to the Administrative Agent or such Bank shall be increased to the extent necessary to yield to the Administrative Agent or such Bank (after payment of all Taxes)

interest or any such other amounts payable hereunder at the rates or in the amounts specified in this Agreement and the Notes. Whenever any Taxes are payable by any Borrower in respect of any payment made hereunder, as promptly as possible thereafter any Borrower shall send to the Administrative Agent for its own account or for the account of such Bank, as the case may be, a certified copy of an original official receipt received by such Borrower showing payment thereof. If such Borrower fails to pay any Taxes when due to the appropriate taxing authority or fails to remit to the Administrative Agent the required receipts or other required documentary evidence, such Borrower shall indemnify the Administrative Agent and the Banks for any incremental taxes, interest or penalties that may become payable by the Administrative Agent or any Bank as a result of any such failure. The agreements in this subsection shall survive the termination of this Agreement and the payment of the Notes and all other amounts payable hereunder.

(b) Each Bank that is not incorporated under the laws of the United States of America or a state thereof agrees that it will deliver to the Company and the Administrative Agent (i) two duly completed copies of United States Internal Revenue Service Form 1001 or 4224 or successor applicable form, as the case may be, and (ii) an Internal Revenue Service Form W-8 or W-9 or successor applicable form. Each such Bank also agrees to deliver to the Company and the Administrative Agent two further copies of the said Form 1001 or 4224 and Form W-8 or W-9, or successor applicable forms or other manner of certification, as the case may be, on or before the date that any such form expires or becomes obsolete or after the occurrence of any event requiring a change in the most recent form previously delivered by it to the Company, and such extensions or renewals thereof as may reasonably be requested by the Company or the Administrative Agent, unless in any such case an event (including, without limitation, any change in treaty, law or regulation) has occurred prior to the date on which any such delivery would otherwise be required which renders all such forms inapplicable or which would prevent such Bank from duly completing and delivering any such form with respect to it and such Bank so advises the Company and the Administrative Agent. Such Bank shall certify (i) in the case of a Form 1001 or 4224, that it is entitled to receive payments under this Agreement without deduction or withholding of any United States federal income taxes and (ii) in the case of a Form W-8 or W-9, that it is entitled to an exemption from United States backup withholding tax.

5.13 Indemnity. Each Borrower agrees to indemnify each Bank and to hold each Bank harmless from any loss or expense which such Bank may sustain or incur as a consequence of (a) default by any Borrower in payment when due of the principal amount of or interest on any Eurodollar Loan, (b) default by any Borrower in making a borrowing of, conversion into or continuation of Eurodollar Loans after such Borrower has given a notice requesting the same in accordance with the provisions of this Agreement, (c) default by any Borrower in making any

prepayment after such Borrower has given a notice thereof in accordance with the provisions of this Agreement or (d) the making of a payment or prepayment of Eurodollar Loans on a day which is not the last day of an Interest Period with respect thereto, including, without limitation, in each case, any such loss or expense arising from the reemployment of funds obtained by it or from fees payable to terminate the deposits from which such funds were obtained. This covenant shall survive the termination of this Agreement and the payment of the Loans or Notes, if any, and all other amounts payable hereunder.

SECTION 6. REPRESENTATIONS AND WARRANTIES

To induce the Banks to enter into this Agreement, each of the Company and the Parent represents and warrants to the Administrative Agent and each Bank that:

6.1 Corporate Existence; Compliance with Law. Each Loan Party (a) is a corporation duly organized, validly existing and in good standing under the laws of its jurisdiction of incorporation, (b) is duly qualified and in good standing in each jurisdiction wherein, in the opinion of the Company and the Parent, the conduct of its business or the ownership of its properties requires such qualification and (c) is in compliance with all Requirements of Law, except to the extent that the failure to comply with paragraph (a), (b) or (c) of this subsection would not, in the aggregate, reasonably be expected to have a Material Adverse Effect.

6.2 Corporate Power, Authorization; Enforceable Obligations. Each Loan Party has the corporate power and authority to make, deliver and perform its obligations under the Loan Documents to which it is or will be a party, and has taken all necessary corporate action to authorize (i) in the case of the Borrowers, the borrowings under this Agreement and any Notes to which it is or will be a party on the terms and conditions hereof and thereof and (ii) the execution, delivery and performance of this Agreement and the Loan Documents to which it is or will be a party. This Agreement has been, and any Note and the other Loan Documents to which it is or will be a party will be, duly executed and delivered on behalf of each relevant Loan Party. This Agreement constitutes, and each of the Notes, if any, and the other Loan Documents when executed and delivered will constitute, a legal, valid and binding obligation of the Loan Party thereto, enforceable against such Loan Party in accordance with its terms, such enforceability subject to limitations under any applicable bankruptcy, insolvency, moratorium or other laws affecting creditors' rights and by general equitable principles (whether applied in a proceeding in equity or at law). No consent of any other party (including stockholders of the Parent) and no consent, license, approval or authorization of, or registration or declaration with, any Governmental Authority is required to be obtained by any Loan

Party in connection with the execution, delivery, performance, validity or enforceability of this Agreement and any Notes.

- 6.3 No Legal Bar. The execution, delivery and performance of this Agreement, the Notes and the other Loan Documents, the borrowings hereunder and the use of the proceeds thereof, will not violate or contravene any material provision of any Requirement of Law or material Contractual Obligation of the Parent, the Company or any of its Subsidiaries and will not result in, or require, the creation or imposition of any material Lien (other than Liens permitted under subsection 9.2) on any of its or their respective properties or revenues pursuant to any such Requirement of Law or Contractual Obligation.
- 6.4 No Material Litigation. There is no legal action, administrative proceeding or arbitration (whether or not purportedly on behalf of the Parent or the Company or any of its Subsidiaries) presently pending, or to the knowledge of the Parent or the Company threatened, against or affecting the Parent or the Company or any of its Subsidiaries which would reasonably be expected to have a Material Adverse Effect, except that the foregoing is subject to the fact that the Parent and the Company cannot predict at this time the results and impact, if any, of (i) those investigations and litigations referred to in Item 3 of the Parent's Annual Report on Form 10-K for the fiscal year ended December 31, 1996 referred to in subsection 6.6 to the extent that inability to make such a prediction or determine liability is stated in Item 3 and (ii) the governmental investigation and related matters described in Note 6 to the Consolidated Financial Statements referred to in Item 3.
- 6.5 Ownership of Properties. Each of the Parent, the Company and each of its Subsidiaries is the tenant under valid leases or has good title to substantially all its properties and assets, real and personal (except defects in title and other matters that would not reasonably be expected to have a Material Adverse Effect), subject to no Lien except as permitted to exist under subsection 9.2.
- 6.6 Financial Condition. The consolidated balance sheet of the Parent and its Subsidiaries as at December 31, 1996 and the related consolidated statements of operations, shareholders' equity and of cash flows (together with the related notes), included or incorporated in the Parent's Annual Report on Form 10-K filed with the SEC for the fiscal year ended December 31, 1996, present fairly in all material respects the financial position of the Parent and its Subsidiaries as at such dates and the results of their operations and their cash flows for the fiscal years then ended. The unaudited consolidated balance sheet of the Parent and its Subsidiaries as at March 31, 1997 and the related unaudited consolidated statement of operations for the three-month interim period, and the related unaudited consolidated statement of cash flows for the three-month interim period, ended on such date, included in the Parent's Quarterly Report on Form 10-Q filed with the SEC for such period, present

fairly in all material respects the financial position of the Parent and its Subsidiaries as at such date and the results of their operations and their cash flows for the three-month period then ended. All of such financial statements, including the notes to such financial statements, have been prepared in conformity with GAAP (subject, in the case of interim statements, to normal year-end adjustments and to the fact that such financial statements may be abbreviated and may omit footnotes or contain incomplete footnotes) consistently applied throughout the periods involved except as stated therein.

- 6.7 Disclosure of Contingent Liabilities. To the best of the knowledge and belief of the Parent, neither the Parent nor any of its Subsidiaries has any contingent obligation, liability for taxes, long-term leases, unusual forward or other liabilities, which are material in amount in relation to the consolidated financial condition of the Parent and its Subsidiaries taken as a whole and which are not disclosed in the financial statements (including the related notes) described in subsection 6.6 above.
- 6.8 ERISA. Each Plan that is intended to qualify under Section 401(a) of the Code satisfies in all material respects the applicable requirements for qualification under that Code Section. No Reportable Event has occurred and is continuing with respect to any such Plan, and neither the Parent nor any of its Subsidiaries has incurred any liability to the PBGC under Section 4062 of ERISA with respect to any such Plan that would reasonably be expected to have a Material Adverse Effect.
- 6.9 Certain Federal Regulations. Neither the Company nor any of its Subsidiaries is engaged in or will engage in the business of extending credit for the purposes of "purchasing" or "carrying" any "margin stock" within the respective meanings of each of the quoted terms under Regulation U of the Board, and no part of the proceeds of any Loan will be used for any purpose which violates, or which would be inconsistent with, the provisions of Regulation U or X of the Board.
- 6.10 No Default. Neither the Parent nor any of its Subsidiaries is in default under or with respect to any of its Contractual Obligations in any respect which would reasonably be expected to have a Material Adverse Effect. No Default or Event of Default has occurred and is continuing.
- 6.11 Taxes. (a) Each of the Parent and its Subsidiaries has filed or caused to be filed all tax returns which, to the knowledge of the Parent, are required to be filed and has paid all taxes shown to be due and payable on said returns or on any assessments made against it or any of its property and all other taxes, fees or other charges imposed on it or any of its property by any Governmental Authority (other than any the amount or validity of which are currently being contested in good faith by appropriate proceedings and with respect to which adequate reserves to the extent required in conformity with

GAAP, have been provided on the books of the Parent or its Subsidiaries, as the case may be) except insofar as the failure to make such filings or payments would not reasonably be expected to have a Material Adverse Effect; and (b) no tax Lien (other than a Lien permitted under subsection 9.2(a)) has been filed, and, to the knowledge of the Parent, no claim is being asserted, with respect to any such tax, fee or other charge which would reasonably be expected to have a Material Adverse Effect.

- 6.12 Investment Company Act; Other Regulations. None of the Parent, the Company or any of its Subsidiaries is an "investment company", or a company "controlled" by an "investment company", within the meaning of the Investment Company Act of 1940, as amended. None of the Parent, the Company or any other Borrower is subject to regulation under any Federal or State statute or regulation which limits its ability to incur Indebtedness.
- 6.13 Purpose of Loans. The proceeds of the Loans shall be used by the Borrowers for general corporate purposes (which may include purchases by the Parent of its capital stock).
- 6.14 Environmental Matters. To the best of the knowledge of the Parent, the operations of the Parent and its Subsidiaries and all parcels of real estate owned or operated by the Parent or its Subsidiaries are in compliance with all Environmental Laws, except where the failure to so comply would not reasonably be expected to have a Material Adverse Effect.
- $\,$ 6.15 Principal Subsidiaries. Set forth on Schedule II are all of the Principal Subsidiaries as of the date hereof.

SECTION 7. CONDITIONS PRECEDENT

- 7.1 Conditions to Effectiveness. The parties hereto acknowledge that the effectiveness of this Agreement is subject to the satisfaction of the following conditions precedent:
 - (a) Loan Documents. The Administrative Agent shall have received (i) this Agreement, executed and delivered by a duly authorized officer of each of the Loan Parties, with a counterpart for each Bank, (ii) for the account of each Bank so requesting, a Revolving Credit Note and a Bid Loan Note conforming to the requirements hereof and executed by a duly authorized officer of the Borrowers and (iii) an incumbency certificate of each of the Loan Parties which covers such officers.
 - (b) Corporate Proceedings. The Administrative Agent shall have received, with a counterpart for each Bank, a copy of the resolutions, in form and substance satisfactory to the Administrative Agent, of the Board of Directors of each of the Loan Parties authorizing (i) the execution, delivery and performance of the Loan Documents to which it

is or will be a party and (ii) the borrowings contemplated hereunder (in the case of each Borrower), certified by the Secretary or an Assistant Secretary of such Loan Party as of the Closing Date, which certificate shall state that the resolutions thereby certified have not been amended, modified, revoked or rescinded and shall be in form and substance satisfactory to the Administrative Agent.

- (c) Legal Opinions. The Administrative Agent shall have received, with a counterpart for each Bank, the following executed legal opinions:
 - (i) the executed legal opinion of counsel to the Company and the Parent who may be the General Counsel of the Company, substantially in the form of Exhibit F-1;
 - (ii) to the extent required pursuant to subsection 13.15(a)(ii), the executed legal opinion of counsel to any other Borrower, in form and substance reasonably satisfactory to the Administrative Agent; and
 - (iii) the executed legal opinion of Simpson Thacher & Bartlett, counsel to the Administrative Agent, substantially in the form of Exhibit F-2.

Each such legal opinion shall cover such other matters incident to the transactions contemplated by this Agreement as the Administrative Agent may reasonably require.

- (d) Officer's Certificate. The Administrative Agent shall have received, with a counterpart for each Bank, a certificate respecting accuracy of representations and warranties, the absence of events having a Material Adverse Effect and the absence of Defaults and Events of Default, substantially in the form of Exhibit G hereto, signed by a Responsible Officer on behalf of each of the Company and the Parent.
- (e) Additional Matters. All corporate and other proceedings, and all documents, instruments and other legal matters in connection with the transactions contemplated by this Agreement shall be satisfactory in form and substance to the Administrative Agent, and the Administrative Agent shall have received such other documents and legal opinions in respect of any aspect or consequence of the transactions contemplated hereby or thereby as it shall reasonably request.
- 7.2 Conditions to Each Loan. The agreement of each Bank to make any Loan requested to be made by it on any date is subject to the satisfaction of the following conditions precedent:

- (a) Representations and Warranties. Each of the representations and warranties made by each of the Loan Parties in or pursuant to subsections 6.1, 6.2, 6.3, 6.5, 6.9, 6.10, 6.11, 6.12 and 6.13 of this Agreement and in or pursuant to any other Loan Document to which it is or will be a party, shall be true and correct in all material respects on and as of such date as if made on and as of such date, and the representation and warranty made pursuant to subsection 6.6 shall be true and correct in all material respects with respect to the financial statements most recently delivered pursuant to subsection 8.1, mutatis mutandis, as if such financial statements delivered pursuant to subsection 8.1 were the financial statements referred to in subsection 6.6.
- (b) No Default. No Default or Event of Default shall have occurred and be continuing on such date or after giving effect to the Loans requested to be made on such date.

Each borrowing by the Borrowers hereunder shall constitute a representation and warranty by the Loan Parties as of the date of such Loan that the conditions contained in this subsection 7.2 have been satisfied.

SECTION 8. AFFIRMATIVE COVENANTS

Each of the Company and the Parent hereby agrees that, so long as the Commitments remain in effect, any Note remains outstanding and unpaid or any other amount is owing to any Bank or the Administrative Agent hereunder, each of the Company and the Parent shall and the Company (except in the case of delivery of financial information, reports and notices) shall cause each of its Principal Subsidiaries to:

8.1 Financial Statements. Furnish to each Bank:

- (a) as soon as available, but in any event within 120 days after the end of each fiscal year of the Parent, a copy of the consolidated balance sheet of the Parent and its Subsidiaries as at the end of such year and the related consolidated statements of operations, shareholders' equity and cash flows for such year (as included or incorporated by reference in the Parent's Annual Report on Form 10-K or successor form filed with the SEC for each such fiscal year), setting forth in each case in comparative form the figures for the previous year, reported on without a "going concern" or like qualification or exception, or qualification arising out of the scope of the audit, by Price Waterhouse or other independent certified public accountants of nationally recognized standing not unacceptable to the Majority Banks; and
- (b) as soon as available, but in any event not later than 75 days after the end of each of the first three $\,$

quarterly periods of each fiscal year of the Parent, the unaudited consolidated balance sheet of the Parent and its Subsidiaries as at the end of such quarter and the related unaudited consolidated statements of operations for such quarter and the related unaudited consolidated statements of operations and cash flows for the portion of the fiscal year through the end of such quarter (as included in the Parent's Quarterly Report on Form 10-Q or successor form filed with the SEC for each such period), setting forth in each case in comparative form the figures for the previous year, certified by a Responsible Officer as being fairly stated in all material respects when considered in relation to the consolidated financial statements of the Parent and its Subsidiaries.

All such financial statements shall be prepared in conformity with GAAP (subject, in the case of interim statements, to normal year-end adjustments and to the fact that such financial statements may be abbreviated and may omit footnotes or contain incomplete footnotes) applied consistently throughout the periods reflected therein and with prior periods (except as disclosed therein).

8.2 Certificates; Other Information. Furnish to each

Bank:

- (a) concurrently with the delivery of the financial statements referred to in subsection 8.1(a), a certificate of the independent certified public accountants reporting on such financial statements stating that in making the examination necessary therefor no knowledge was obtained of any Default or Event of Default, except as specified in such certificate;
- (b) concurrently with the delivery of the financial statements referred to in subsections 8.1(a) and 8.1(b), a certificate of a Responsible Officer of the Parent in his capacity as such officer stating that, to the best of such Officer's knowledge, each of the Borrowers and the Parent during such period has observed or performed all of its covenants and other agreements, and satisfied every condition, contained in this Agreement and in the Notes and the other Loan Documents to which it is a party to be observed, performed or satisfied by it, and that such Officer has obtained no knowledge of any Default or Event of Default except as specified in such certificate and showing in detail the calculation of compliance with subsections 9.1 and 9.2;
- (c) concurrently with the delivery of the financial statements referred to in subsection 8.1(a), a list of the Principal Subsidiaries as of the corresponding fiscal year end, certified by a Responsible Officer in his capacity as such officer;

- (d) within ten Business Days after the same are sent, copies of all financial statements and reports which the Parent sends to its shareholders generally relating to the business of the Parent and its Subsidiaries, and within ten Business Days after the same are filed, copies of all reports on Forms 10-K, 10-Q, 8-K, 8 and 10, and Schedules 13D, 13E-3, 13E-4, 13-G, 14D-1 and 14D-9, or successor forms or schedules, and the final prospectus in each effective registration statement (other than registration statements on Form S-8) and each post-effective amendment to such registration statement which the Parent may make to, or file with, the SEC; and
- (e) promptly, subject to reasonable confidentiality requirements agreed to by the Company and such Bank, such additional financial and other information as any Bank may from time to time reasonably request.
- 8.3 Payment of Obligations. Pay, discharge or otherwise satisfy at or before maturity or before they become delinquent, as the case may be, all its obligations of whatever nature, except where the amount or validity thereof is currently being contested in good faith by appropriate proceedings and reserves, to the extent required in conformity with GAAP with respect thereto, have been provided on the books of the Parent or its Subsidiaries, as the case may be, and except to the extent that the failure to so pay, discharge or otherwise satisfy such obligations would not result in a Default or Event of Default under Section 10(e)(i).
- 8.4 Conduct of Business and Maintenance of Existence. Preserve, renew and keep in full force and effect its corporate existence and take all reasonable action to maintain all corporate rights, privileges and franchises necessary or desirable in the normal conduct of its business, except as otherwise permitted pursuant to subsection 9.3; comply with all Contractual Obligations and Requirements of Law except to the extent that failure to comply therewith would not, in the aggregate, reasonably be expected to have a Material Adverse Effect.
- 8.5 Insurance. Maintain with financially sound and reputable insurance companies (which may include, without limitation, captive insurers), such insurance coverage as is reasonable for the business activities of the Parent and its Subsidiaries; and furnish to the Administrative Agent, upon written request, such information as the Administrative Agent may reasonably request as to its insurance program.
- 8.6 Inspection of Property, Books and Records; Discussions. Permit representatives of any Bank (subject to reasonable safety and confidentiality requirements) to visit and inspect any of its properties and examine and make abstracts from any of its books and records at any reasonable time and as often as may reasonably be desired and to discuss the business,

operations, properties and financial and other condition of the Parent and its Subsidiaries with officers and employees of the Parent and its Subsidiaries and, provided representatives of the Parent are given an opportunity to participate, with its independent certified public accountants.

- $\,$ 8.7 Notices. Promptly give notice to the Administrative Agent and each Bank of:
 - (a) the occurrence of any Default or Event of Default;
 - (b) any (i) default or event of default under any Contractual Obligation of the Parent or any of its Subsidiaries or (ii) litigation, investigation or proceeding which may exist at any time between the Parent or any of its Subsidiaries and any Governmental Authority, which in either case, would reasonably be expected to have a Material Adverse Effect;
 - (c) any litigation or proceeding affecting the Parent or any of its Subsidiaries in which the then reasonably anticipated exposure of the Parent and its Subsidiaries is \$10,000,000 or more and not covered by insurance, or in which injunctive or similar relief is sought which is then reasonably anticipated to have an adverse economic effect on the Parent and its Subsidiaries of \$10,000,000 or more;
 - (d) the following events, as soon as possible and in any event within 30 days after the Company or the Parent knows or has reason to know thereof: (i) the occurrence or expected occurrence of any Reportable Event with respect to any Plan, or any withdrawal from, or the termination, Reorganization or Insolvency of any Multiemployer Plan or (ii) the institution of proceedings or the taking of any other action by the PBGC or the Company or the Parent or any Commonly Controlled Entity or any Multiemployer Plan with respect to the withdrawal from, or the terminating, Reorganization or Insolvency of, any Plan, where in connection with any of the events described in (i) or (ii) above the liability to the Company or a Commonly Controlled Entity would reasonably be expected to be \$10,000,000 or more;
 - (e) any upgrading, downgrading or cessation in the rating of the long term senior unenhanced, unsecured debt of the Company by the rating agency or agencies whose rating on such debt is then being used to determine the Applicable Margin and the Facility Fee Rate; and
 - (f) a development or event which would reasonably be expected to have a Material Adverse Effect. $\begin{tabular}{ll} \hline \end{tabular}$

Each notice pursuant to this subsection shall be accompanied by a statement of a Responsible Officer setting forth details of the

occurrence referred to therein and stating what action each of the Company and the Parent proposes to take with respect thereto.

8.8 Environmental Laws.

- (a) Comply with all Environmental Laws and obtain and comply with and maintain any and all licenses, approvals, registrations or permits required by Environmental Laws, except to the extent that failure to do so would not be reasonably expected to have a Material Adverse Effect; and
- (b) Defend, indemnify and hold harmless the Administrative Agent and the Banks, and their respective employees, agents, officers and directors, from and against any claims, demands, penalties, fines, liabilities, settlements, damages, costs and expenses of whatever kind or nature known or unknown, contingent or otherwise, arising out of, or in any way relating to the violation of or noncompliance with any Environmental Laws applicable to the real property owned or operated by the Company, the Parent or any of the Company's Subsidiaries, or any orders, requirements or demands of Governmental Authorities related thereto, including, without limitation, attorney's and consultant's fees, investigation and laboratory fees, court costs and litigation expenses, except to the extent that any of the foregoing arise out of the gross negligence or willful misconduct of the party seeking indemnification therefor.

SECTION 9. NEGATIVE COVENANTS

The Parent hereby agrees that, so long as the Commitments remain in effect, any Note remains outstanding and unpaid or any other amount is owing to any Bank or the Administrative Agent hereunder, it shall not, and (except with respect to subsections 9.1 and 9.5(b)) shall not permit any of its Subsidiaries to, directly or indirectly:

9.1 Financial Condition Covenants.

- (a) Consolidated Debt to EBITDA. Permit on the last day of any fiscal quarter of the Company commencing with June 30, 1997, the ratio of Consolidated Debt as at the end of such fiscal quarter to EBITDA for the period of four consecutive fiscal quarters then ended to be greater than 3.5 to 1.0.
- (b) Interest Coverage. Permit for any period of four consecutive fiscal quarters ending on the last day of any fiscal quarter of the Company commencing with June 30, 1997, the ratio of EBIT for such four fiscal quarter period to Consolidated Interest Expense for such four fiscal quarter period to be less than 2.0 to 1.0.
- $\,$ 9.2 Limitation on Liens. Create, incur, assume or suffer to exist any Lien upon any of its property, assets or

revenues (which property, assets or revenues are or would be reflected from time to time on the consolidated financial statements of the Parent and its Subsidiaries in accordance with GAAP), whether now owned or hereafter acquired, except for:

- (a) Liens for taxes or other governmental charges not yet due or which are being contested in good faith by appropriate proceedings, provided that adequate reserves with respect thereto are maintained on the books of the Parent or its Subsidiaries, as the case may be, to the extent required in conformity with GAAP;
- (b) carriers', warehousemen's, mechanics', materialmen's, repairmen's, vendors', landlords', brokers', bankers' and other like Liens arising in the ordinary course of business relating to obligations which are not overdue for a period of more than 60 days or which are being contested in good faith and Liens arising out of judgments or awards that are either discharged within 60 days after entry or execution of which has been stayed pending the outcome of appeal or review proceedings;
- (c) pledges or deposits in connection with workers' compensation, unemployment insurance and other social security legislation and deposits securing liability to insurance carriers under insurance or self-insurance arrangements;
- (d) pledges, deposits and similar arrangements in connection with or to secure performance of bids, tenders, leases and other deposits to secure the performance of bids, trade contracts (other than for borrowed money), leases, statutory obligations, surety and appeal bonds, performance bonds and other obligations of a like nature incurred in the ordinary course of business and contractual rights of other Persons to make set-offs and to require security in connection with letters of credit, currency, commodity and interest rate contracts, surety bonds, leases, banking and brokerage agreements and other transactions in the ordinary course of business;
- (e) leases, easements, rights-of-way, restrictions and other similar encumbrances incurred in the ordinary course of business which would not reasonably be expected to have a Material Adverse Effect:
- (f) Liens on the property, assets or revenues of a Person which becomes a Subsidiary after the date hereof, to the extent that (i) such Liens existed at the time such Person became a Subsidiary and were not created in anticipation thereof, (ii) any such Lien is not extended to cover any property, assets or revenues of such Person after the time such Person becomes a Subsidiary, and (iii) the amount of Indebtedness secured thereby is not thereafter increased;

- (g) Liens arising in connection with (i) industrial development, pollution control or other tax exempt financing transactions, provided that such Liens do not at any time encumber any property other than the property financed by such transaction and other property, assets or revenues related to the property so financed on which Liens are customarily granted in connection with such transactions, or (ii) conveyances of any production payment or other obligation to make a production payment (A) which is to be made solely from production from oil, gas or other underground mineral properties dedicated thereto or (B) as to which production payment amount the obligee's sole recourse is to such properties;
- (h) Liens (including, without limitation, Liens incurred in connection with Capitalized Leases, operating leases and sale-leaseback transactions) securing Indebtedness of the Parent and its Subsidiaries incurred to finance the acquisition of fixed or capital assets, and refinancings thereof, provided that (i) such Liens do not at any time encumber any property other than the property financed by such Indebtedness and other property, assets or revenues related to the property so financed on which Liens are customarily granted in connection with such financings or refinancings, and (ii) the principal amount of Indebtedness secured by any such Lien shall at no time exceed 100% of the greater of the original purchase price of such property at the time it was acquired and the fair market value of such property as reasonably determined by a Responsible Officer of the Company in good faith thereafter, plus fees and other costs related to the financing or refinancing thereof which have been agreed upon in an arm's length manner;
- (i) Liens incurred in connection with accounts receivable sale transactions entered into by the Parent or its Subsidiaries;
- (j) Liens securing Contractual Obligations of any Subsidiary to the Parent, the Company or any Domestic Subsidiary;
- $\mbox{(k)}$ Liens on the property, assets or revenues of any Foreign Subsidiary or any Excluded Subsidiary; and
- (1) Liens (not otherwise permitted hereunder) which secure obligations in an aggregate amount at any time outstanding not exceeding an amount equal to 5% of the amount recorded opposite the caption "Properties and equipment, net" (or the equivalent caption) on the consolidated balance sheet of the Parent and its Subsidiaries most recently delivered to the Administrative Agent pursuant to subsection 8.1.

- 9.3 Limitation on Fundamental Changes. Convey, sell, lease, assign, transfer or otherwise dispose of (including by merger, consolidation, sale of stock, liquidation or dissolution) all or substantially all of the property, assets or business of the Parent and its Subsidiaries taken as a whole.
- 9.4 Limitation on Asset Transfers to Foreign Subsidiaries. With respect to the Parent or any Domestic Subsidiary, convey, sell, lease, assign, transfer or otherwise dispose of (collectively, a "transfer") any of its property, business or assets (including, without limitation leasehold interests), whether now owned or hereafter acquired, to any Foreign Subsidiary, except such transfers which, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect.
- 9.5 Limitation on Subordinated Debt. Permit any Subsidiary of the Parent (other than the Company) to create, incur, assume or suffer to exist any subordinated indebtedness other than (a) subordinated indebtedness of a Person which becomes a Subsidiary after the date hereof to the extent such indebtedness existed at the time such Person became a Subsidiary and was not incurred in anticipation thereof and any refinancings of such indebtedness after such time so long as the principal amount thereof is not increased or (b) subordinated indebtedness of such Subsidiary held by the Parent or any other Subsidiary of the Parent.

SECTION 10. EVENTS OF DEFAULT

If any of the following events shall occur and be

continuing:

- (a) Any Borrower shall fail to pay any principal of any Loan or Note when due in accordance with the terms thereof or hereof; or any Borrower shall fail to pay any interest on any Loan or Note, or any other amount payable hereunder, within five Business Days after any such interest or other amount becomes due in accordance with the terms thereof or hereof; or
- (b) Any representation or warranty made, or pursuant to subsection 7.2, deemed made, by any Loan Party herein or in any other Loan Document or which is contained in any certificate, document or financial or other statement furnished at any time under or in connection with this Agreement shall prove to have been incorrect in any material adverse respect on or as of the date made or deemed made; or
- (c) The Parent or any Subsidiary shall default in the observance or performance of any agreement contained in subsection $9.1,\ 9.3,\ 9.4$ or 9.5; or

- (d) Any Loan Party shall default in the observance or performance of any other agreement contained in this Agreement (other than as provided in paragraphs (a) through (c) of this Section), and such default shall continue unremedied for a period of 30 days; or
- (e) The Parent or any of its Subsidiaries (other than the Excluded Subsidiaries) shall (i) default in any payment of principal of or interest on, or any other amount payable with respect to, any (A) Domestic Indebtedness (other than the Notes and Loans) in an aggregate principal amount for all such Domestic Indebtedness of \$10,000,000 or more, or (B) Foreign Subsidiary Indebtedness (other than the Notes and Loans) in an aggregate principal amount for all such Foreign Subsidiary Indebtedness of \$20,000,000 or more, beyond the period of grace (not to exceed 30 days), if any, provided in the instrument or agreement under which such Indebtedness was created; or (ii) default in the observance or performance of any other agreement relating to any such Indebtedness in the amounts specified in clause (i) above or contained in any instrument or agreement evidencing, securing or relating thereto, or any other event shall occur or condition exist in any case which continues uncured or unwaived (and, if waived, without any change in the material terms of such Indebtedness) after the expiration of all applicable grace periods, the effect of which default or other event or condition is to cause, or to permit the holder or holders of such Indebtedness (or a trustee or agent on behalf of such holder or holders) to cause, with the giving of notice if required, such Indebtedness to become due prior to its stated maturity; or
- (f) (i) The Parent or any Principal Subsidiary (other than the Excluded Subsidiaries) shall commence any case, proceeding or other action (A) under any existing or future law of any jurisdiction, domestic or foreign, relating to bankruptcy, insolvency, reorganization or relief of debtors, seeking to have an order for relief entered with respect to it, or seeking to adjudicate it a bankrupt or insolvent, or seeking reorganization, arrangement, adjustment, winding-up, liquidation, dissolution, composition or other relief with respect to it or its debts, or (B) seeking appointment of a receiver, trustee, custodian or other similar official for it or for all or any substantial part of its assets, or the Parent or any such Principal Subsidiary shall make a general assignment for the benefit of its creditors; or (ii) there shall be commenced against the Parent or any such Principal Subsidiary any case, proceeding or other action of a nature referred to in clause (i) above which (A) results in the entry of an order for relief or any such adjudication or appointment or (B) remains undismissed, undischarged or unbonded for a period of 60 days; or (iii) there shall be commenced against the Parent or any such Principal Subsidiary any case, proceeding or other action seeking issuance of a warrant of attachment, execution, distraint or

similar process against all or any substantial part of its assets which results in the entry of an order for any such relief which shall not have been vacated, discharged, or stayed or bonded pending appeal within 60 days from the entry thereof; or (iv) the Parent or any such Principal Subsidiary shall take any action in furtherance of, or indicating its consent to, approval of, or acquiescence in, any of the acts set forth in clause (i), (ii) or (iii) above; or (v) the Parent or any such Principal Subsidiary shall generally not, or shall be unable to, or shall admit in writing its inability to, pay its debts as they become due; or

- (g) (i) Any Person shall engage in any non-exempt "prohibited transaction" (as defined in Section 406 of ERISA or Section 4975 of the Code) involving any Plan, (ii) any "accumulated funding deficiency" (as defined in Section 302 of ERISA), whether or not waived, shall exist with respect to any Plan, (iii) a Reportable Event shall occur with respect to, or judicial proceedings shall commence to have a trustee appointed, or a trustee shall be appointed, to administer or to terminate, any Single Employer Plan, which Reportable Event or commencement of judicial proceedings or appointment of a trustee is, in the reasonable opinion of the Majority Banks, likely to result in the termination of such Plan for purposes of Title IV of ERISA, (iv) any Single Employer Plan shall terminate for purposes of Title IV of ERISA, (v) the Company or any Commonly Controlled Entity shall, or in the reasonable opinion of the Majority Banks is likely to, incur any liability in connection with a withdrawal from, or the Insolvency or Reorganization of, a Multiemployer Plan or (vi) any other event or condition shall occur or exist, with respect to a Plan; and in each case in clauses (i) through (vi) above, such event or condition, together with all other such events or conditions, if any, could reasonably be expected to subject the Company or any of its Subsidiaries to any tax, penalty or other liabilities which in the aggregate would have a Material Adverse Effect; or
- (h) One or more judgments or decrees shall be entered against the Parent or any of its Subsidiaries in aggregate amounts (not paid or fully covered by insurance) of \$10,000,000 or more and all such judgments or decrees shall not have been vacated, discharged, stayed or bonded pending appeal within 60 days from the entry thereof; or
- (i) The Parent shall cease to own directly or indirectly of record and beneficially free and clear of Liens at least 75% of the shares of the issued and outstanding capital stock of the Company;

then, and in any such event, (A) if such event is an Event of Default specified in clause (i), (ii) or (iii) of paragraph (f) above with respect to any of the Borrowers, automatically the

Commitments to such Borrower shall immediately terminate and the Loans made to such Borrower hereunder (with accrued interest thereon) and all other amounts owing under this Agreement and the Notes of such Borrower shall immediately become due and payable, and (B) if such event is any other Event of Default, either or both of the following actions may be taken: (i) with the consent of the Majority Banks, the Administrative Agent may, or upon the request of the Majority Banks, the Administrative Agent shall, by notice to the Company declare the Commitments of any or all of the Borrowers to be terminated forthwith, whereupon such Commitments shall immediately terminate; and (ii) with the consent of the Majority Banks, the Administrative Agent may, or upon the request of the Majority Banks, the Administrative Agent may, or upon the request of the Majority Banks, the Administrative Agent may, or upon the rough of the Borrowers (with accrued interest thereon) and all other amounts owing by such Borrower under this Agreement and the Notes of such Borrower to be due and payable forthwith, whereupon the same shall immediately become due and payable. Except as expressly provided above in this Section, presentment, demand, protest and all other notices of any kind are hereby expressly waived.

SECTION 11. THE ADMINISTRATIVE AGENT

11.1 Appointment. Each Bank hereby irrevocably designates and appoints Chase as the Administrative Agent of such Bank under this Agreement and the other Loan Documents, and each such Bank irrevocably authorizes Chase, as the Administrative Agent for such Bank, to take such action on its behalf under the provisions of this Agreement and the other Loan Documents and to exercise such powers and perform such duties as are expressly delegated to the Administrative Agent by the terms of this Agreement and the other Loan Documents, together with such other powers as are reasonably incidental thereto. Notwithstanding any provision to the contrary elsewhere in this Agreement, the Administrative Agent shall not have any duties or responsibilities, except those expressly set forth herein, or any fiduciary relationship with any Bank, and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or any other Loan Document or otherwise exist against the Administrative Agent.

11.2 Delegation of Duties. The Administrative Agent may execute any of its duties under this Agreement and the other Loan Documents by or through agents or attorneys-in-fact and shall be entitled to advice of counsel concerning all matters pertaining to such duties. The Administrative Agent shall not be responsible for the negligence or misconduct of any agents or attorneys-in-fact selected by it with reasonable care.

11.3 Exculpatory Provisions. Neither the Administrative Agent nor any of its officers, directors, employees, agents, attorneys-in-fact or Affiliates shall be (i) liable for any action lawfully taken or omitted to be taken by it

or such Person under or in connection with this Agreement or any other Loan Document (except for its or such Person's own gross negligence or willful misconduct) or (ii) responsible in any manner to any of the Banks for any recitals, statements, representations or warranties made by any Loan Party or any officer thereof contained in this Agreement or any other Loan Document or in any certificate, report, statement or other document referred to or provided for in, or received by the Administrative Agent under or in connection with, this Agreement or any other Loan Document or for the value, validity, effectiveness, genuineness, enforceability or sufficiency of this Agreement or the Notes or any other Loan Document or for any failure of any Loan Party to perform its obligations hereunder or thereunder. The Administrative Agent shall not be under any obligation to any Bank to ascertain or to inquire as to the observance or performance of any of the agreements contained in, or conditions of, this Agreement or any other Loan Document, or to inspect the properties, books or records of any Loan Party.

11.4 Reliance by Administrative Agent. The Administrative Agent shall be entitled to rely, and shall be fully protected in relying, upon any note, writing, resolution, notice, consent, certificate, affidavit, letter, cablegram, telegram, telecopy, statement, order or other document or conversation believed by it to be genuine and correct and to have been signed, sent or made by the proper Person or Persons and upon advice and statements of legal counsel (including, without limitation, counsel to the Company, the Parent or any other Borrower), independent accountants and other experts selected by the Administrative Agent. The Administrative Agent may deem and treat the payee of any Note as the owner thereof for all purposes unless a written notice of assignment, negotiation or transfer thereof shall have been filed with the Administrative Agent. The Administrative Agent shall be fully justified in failing or refusing to take any action under this Agreement or any other Loan Document unless it shall first receive such advice or concurrence of the Majority Banks as it deems appropriate or it shall first be indemnified to its satisfaction by the Banks against any and all liability and expense which may be incurred by it by reason of taking or continuing to take any such action. The Administrative Agent shall in all cases be fully protected in acting, or in refraining from acting, under this Agreement and the Notes and the other Loan Documents in accordance with a request of the Majority Banks, and such request and any action taken or failure to act pursuant thereto shall be binding upon all the Banks and all future holders of the Notes.

11.5 Notice of Default. The Administrative Agent shall not be deemed to have knowledge or notice of the occurrence of any Default or Event of Default hereunder unless the Administrative Agent has received notice from a Bank or any Loan Party referring to this Agreement, describing such Default or Event of Default and stating that such notice is a "notice of default". In the event that the Administrative Agent receives such a notice, the Administrative Agent shall give notice thereof

to the Banks. The Administrative Agent shall take such action with respect to such Default or Event of Default as shall be reasonably directed by the Majority Banks; provided that unless and until the Administrative Agent shall have received such directions, the Administrative Agent may (but shall not be obligated to) take such action, or refrain from taking such action, with respect to such Default or Event of Default as it shall deem advisable in the best interests of the Banks.

11.6 Non-Reliance on Administrative Agent and Other Banks. Each Bank expressly acknowledges that neither the Administrative Agent nor any of its officers, directors, employees, agents, attorneys-in-fact or Affiliates has made any representations or warranties to it and that no act by the Administrative Agent hereinafter taken, including any review of the affairs of the Loan Parties, shall be deemed to constitute any representation or warranty by the Administrative Agent to any Bank. Each Bank represents to the Administrative Agent that it has, independently and without reliance upon the Administrative Agent or any other Bank, and based on such documents and information as it has deemed appropriate, made its own appraisal of and investigation into the business, operations, property, financial and other condition and creditworthiness of the Loan Parties and made its own decision to make its Loans hereunder and enter into this Agreement. Each Bank also represents that it will, independently and without reliance upon the Administrative Agent or any other Bank, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit analysis, appraisals and decisions in taking or not taking action under this Agreement and the other Loan Documents, and to make such investigation as it deems necessary to inform itself as to the business, operations, property, financial and other condition and creditworthiness of the Loan Parties. Except for notices, reports and other documents expressly required to be furnished to the Banks by the Administrative Agent hereunder, the Administrative Agent shall not have any duty or responsibility to provide any Bank with any credit or other information concerning the business, operations, property, condition (financial or otherwise), prospects or creditworthiness of the Loan Parties which may come into the possession of the Administrative Agent or any of its officers, directors, employees, agents, attorneys-in-fact or Affiliates.

11.7 Indemnification. The Banks agree to indemnify the Administrative Agent in its capacity as such (to the extent not reimbursed by the Loan Parties and without limiting the obligation of the Parent, the Company and any other Borrower to do so), ratably according to the respective amounts of their Commitments as in effect on the date on which the claim for indemnity by the Administrative Agent is sought, from and against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind whatsoever which may at any time (including, without limitation, at any time following the payment of the Notes) be imposed on, incurred by or asserted against the Administrative

Agent in any way relating to or arising out of this Agreement, any of the other Loan Documents or any documents contemplated by or referred to herein or therein or the transactions contemplated hereby or thereby or any action taken or omitted by the Administrative Agent under or in connection with any of the foregoing; provided that no Bank shall be liable for the payment of any portion of such liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements resulting solely from the Administrative Agent's gross negligence or willful misconduct. The agreements in this subsection shall survive the payment of the Notes and all other amounts payable hereunder.

11.8 Administrative Agent in Its Individual Capacity. The Administrative Agent and its Affiliates may make loans to, accept deposits from and generally engage in any kind of business with the Parent, the Company or any other Borrower as though the Administrative Agent were not the Administrative Agent hereunder and under the other Loan Documents. With respect to its Loans made or renewed by it and any Note issued to it, the Administrative Agent shall have the same rights and powers under this Agreement and the other Loan Documents as any Bank and may exercise the same as though it were not the Administrative Agent, and the terms "Bank" and "Banks" shall include the Administrative Agent in its individual capacity.

11.9 Successor Administrative Agent. The Administrative Agent may resign as Administrative Agent upon 10 days' notice to the Banks. If the Administrative Agent shall resign as Administrative Agent under this Agreement and the other Loan Documents, then the Majority Banks shall appoint from among the Banks a successor agent for the Banks, which successor agent shall be approved by the Company, whereupon such successor agent shall succeed to the rights, powers and duties of the Administrative Agent, and the term "Administrative Agent" shall mean such successor agent effective upon its appointment, and the former Administrative Agent's rights, powers and duties as Administrative Agent shall be terminated, without any other or further act or deed on the part of such former Administrative Agent or any of the parties to this Agreement or any holders of the Notes. After any retiring Administrative Agent's resignation as Administrative Agent, the provisions of this subsection shall inure to its benefit as to any actions taken or omitted to be taken by it while it was Administrative Agent under this Agreement and the other Loan Documents.

SECTION 12. GUARANTEES

12.1 Parent Guarantee. The Parent hereby unconditionally and irrevocably guarantees to the Administrative Agent and the Banks the prompt and complete payment and performance by each of the Borrowers when due (whether at the stated maturity, by acceleration or otherwise) of the Obligations owing to the Administrative Agent and the Banks by such

Borrowers. This guarantee (the "Parent Guarantee") shall remain in full force and effect until the Obligations of each of the Borrowers are indefeasibly paid in full, notwithstanding that from time to time prior thereto any Borrower may be free from any Obligations. The Parent agrees that whenever, at any time, or from time to time, it shall make any payment to the Administrative Agent or any Bank on account of its liability under this Parent Guarantee, it will notify the Administrative Agent and such Bank in writing that such payment is made under this Parent Guarantee for such purpose. No payment or payments made by any Borrower or any other Person or received or collected by the Administrative Agent or any Bank from any Borrower or any other Person by virtue of any action or proceeding or any offset or appropriation or application, at any time or from time to time, in reduction of or in payment of the Obligations of such Borrower shall be deemed to modify, reduce, release or otherwise affect the liability of the Parent under this Parent Guarantee, which shall remain obligated under this Parent Guarantee, notwithstanding any such payment or payments until the Obligations are paid in full.

12.2 Company Guarantee. The Company hereby unconditionally and irrevocably guarantees to the Administrative Agent and the Banks, the prompt and complete payment and performance by each of the other Borrowers when due (whether at the stated maturity, by acceleration or otherwise) of the Obligations owing to the Administrative Agent and the Banks by such Borrowers. This guarantee (the "Company Guarantee") shall remain in full force and effect until the Obligations of each such Borrower are indefeasibly paid in full, notwithstanding that from time to time prior thereto any such Borrower may be free from any Obligations. The Company agrees that whenever, at any time, or from time to time, it shall make any payment to the Administrative Agent or any Bank on account of its liability under this Company Guarantee, it will notify the Administrative Agent and such Bank in writing that such payment is made under this Company Guarantee for such purpose. No payment or payments made by any such Borrower or any other Person or received or collected by the Administrative Agent or any Bank from any such Borrower or any other Person by virtue of any action or proceeding or any offset or appropriation or application, at any time or from time to time, in reduction of or in payment of the Obligations of such Borrowers shall be deemed to modify, reduce, release or otherwise affect the liability of the Company under this Company Guarantee, which shall remain obligated under this Company Guarantee, notwithstanding any such payment or payments until the Obligations of such Borrowers are paid in full.

12.3 No Subrogation, Contribution, Reimbursement or Indemnity. Notwithstanding anything to the contrary in the Parent Guarantee and the Company Guarantee (together, the "Guarantees", each a "Guarantee"), the Parent and the Company (together, the "Guaranteeing Parties," each a "Guaranteeing Party") hereby irrevocably waives all rights which may have arisen in connection with its Guarantee to be subrogated to any

of the rights (whether contractual, under the Bankruptcy Code, including Section 509 thereof, under common law or otherwise) of the Administrative Agent or any Bank against the Company or any other Borrowers (together, the "Guaranteed Parties", each a "Guaranteed Party") for the payment of the Obligations. Each Guaranteeing Party hereby further irrevocably waives all contractual, common law, statutory or other rights of reimbursement, contribution, exoneration or indemnity (or any similar right) from or against any Guaranteed Party or Parties or any other Person which may have arisen in connection with its Guarantee. So long as the Obligations remain outstanding, if any amount shall be paid by or on behalf of any Guaranteed Party to the Guaranteeing Party on account of any of the rights waived in this subsection, such amount shall be held by such Guaranteeing Party in trust, segregated from other funds of such Guaranteeing Party, and shall, forthwith upon receipt by such Guaranteeing Party, be turned over to the Administrative Agent in the exact form received by such Guaranteeing Party (duly endorsed by such Guaranteeing Party to the Administrative Agent, if required), to be applied against the Obligations of such Guaranteed Party or Parties, whether matured or unmatured, in such order as the Administrative Agent may determine. The provisions of this subsection as they apply to each of the Guaranteeing Parties shall survive the payment in full of the Obligations of its Guaranteed Party or Parties.

12.4 Amendments, etc., with respect to the Obligations. Each Guaranteeing Party shall remain obligated under its Guarantee notwithstanding that, without any reservation of rights against such Guaranteeing Party, and without notice to or further assent by such Guaranteeing Party, any demand for payment of any of the Obligations made by the Administrative Agent or any Bank may be rescinded by the Administrative Agent or such Bank, and any of the Obligations continued, and the Obligations, or the liability of any other party upon or for any part thereof, or any collateral security or guarantee therefor or right of offset with respect thereto, may, from time to time, in whole or in part, be renewed, extended, amended, modified, accelerated, compromised, waived, surrendered or released by the Administrative Agent or any Bank, and this Agreement, the Notes and the other Loan Documents may be amended, modified, supplemented or terminated, in whole or in part, as the Administrative Agent or the Banks (or the Majority Banks, as the case may be) may deem advisable from time to time in accordance with the provisions of subsection 13.1(a), and any collateral security, guarantee or right of set-off at any time held by the Administrative Agent or any Bank for the payment of the Obligations may be sold, exchanged, waived, surrendered or released. Neither the Administrative Agent nor any Bank shall have any obligation to protect, secure, perfect or insure any Lien at any time held by it as security for the Obligations or for the obligations of any Guaranteeing Party under its Guarantee or any property subject thereto.

12.5 Guarantee Absolute and Unconditional. Each Guaranteeing Party waives any and all notice of the creation, renewal, extension or accrual of any of the Obligations and notice of or proof of reliance by the Administrative Agent or any Bank upon its Guarantee or acceptance of its Guarantee; the Obligations, and any of them, shall conclusively be deemed to have been created, contracted or incurred in reliance upon the Guarantees; and all dealings between the Borrowers and the Parent, on the one hand, and the Administrative Agent and the Banks, on the other, shall likewise be conclusively presumed to have been had or consummated in reliance upon the Guarantees. Each Guaranteeing Party waives diligence, presentment, protest, notice of intent to accelerate, notice of acceleration, demand for payment and notice of default or nonpayment to or upon any Guaranteed Party or such Guaranteeing Party with respect to the Obligations. The Guarantees shall be construed as a continuing, absolute and unconditional guarantee of payment without regard to (a) the validity or enforceability of this Agreement, any Note, any other Loan Document, any of the Obligations or any collateral security therefor or guarantee or right of set-off with respect thereto at any time or from time to time held by the Administrative Agent or any Bank, (b) any defense, offset or counterclaim (other than a defense of payment or performance) which may at any time be available to or be asserted by any of the Guaranteed Parties against the Administrative Agent or any Bank or (c) any other circumstance whatsoever (with or without notice to or knowledge of any of the Guaranteed Parties or such Guaranteeing Party) which constitutes, or might be construed to constitute, an equitable or legal discharge of any of the Guaranteed Parties for the Obligations of such Guaranteed Party, or of such Guaranteeing Party under its Guarantee, in bankruptcy or in any other instance. When the Administrative Agent is pursuing its rights and remedies hereunder against any Guaranteeing Party, the Administrative Agent or any Bank may, but shall be under no obligation to, pursue such rights and remedies as it may have against its Guaranteed Party or any other Person or against any collateral security or guarantee for the Obligations or any right of offset with respect thereto, and any failure by the Administrative Agent or any Bank to pursue such other rights or remedies or to collect any payments from such Guaranteed Party or such other Person or to realize upon any such collateral security or guarantee or to exercise such right of offset or any release of such Guaranteed Party or such other Person or of any such collateral security, guarantee or right of offset, shall not relieve such Guaranteeing Party of any liability under its Guarantee, and shall not impair or affect the rights and remedies, whether express, implied or available as a matter of law, of the Administrative Agent and the Banks against such Guaranteeing Party.

12.6 Reinstatement. Each Guarantee shall continue to be effective, or be reinstated, as the case may be, if at any time payment, or any part thereof, of any of the Obligations of any Guaranteed Party thereunder is rescinded or must otherwise be restored or returned by the Administrative Agent or any Bank upon

the insolvency, bankruptcy, dissolution, liquidation or reorganization of such Guaranteed Party or upon or as a result of the appointment of a receiver, intervenor or conservator of, or trustee or similar officer for, such Guaranteed Party or any substantial part of any of its property, or otherwise, all as though such payments had not been made.

12.7 Payments. Each Guaranteeing Party hereby agrees that the Obligations will be paid to the Administrative Agent for the benefit of the Administrative Agent and the Banks, as the case may be, without set-off or counterclaim in Dollars or Alternative Currency, as appropriate, in immediately available funds at the office of the Administrative Agent, c/o Loan and Agency Services Group (Clearing Account No. 144810547) located at One Chase Manhattan Plaza, 8th Floor, New York, New York 10081.

SECTION 13. MISCELLANEOUS

13.1 Amendments and Waivers; Replacement of Banks. (a) Neither this Agreement, any Note, any other Loan Document, nor any terms hereof or thereof may be amended, supplemented or modified except in accordance with the provisions of this subsection. With the written consent of the Majority Banks, the Administrative Agent, the Parent and the Company may, from time to time, enter into written amendments, supplements or modifications hereto and to the Notes, if any, and the other Loan Documents for the purpose of adding any provisions to this Agreement or the Notes, if any, or the other Loan Documents or changing in any manner the rights of the Banks, the Parent or of the Borrowers hereunder or thereunder or waiving, on such terms and conditions as the Administrative Agent may specify in such instrument, any of the requirements of this Agreement or the Notes, if any, or the other Loan Documents or any Default or Event of Default and its consequences; provided, however, that no such waiver and no such amendment, supplement or modification shall (i) reduce the amount or extend the maturity of any Loan or Note or any installment thereof, or reduce the rate or extend the time of payment of interest thereon, or reduce any fee payable to any Bank hereunder, or change the amount of any Bank's Commitment, in each case without the consent of the Bank affected thereby, or (ii) amend, modify or waive any provision of this subsection or reduce the percentage specified in the definition of Majority Banks, or consent to the assignment or transfer by the Parent or any Borrower of any of its rights and obligations under this Agreement and the other Loan Documents, or amend, modify or waive any provision of Section 12, in each case without the written consent of all the Banks, or (iii) amend, modify or waive any provision of Section 11 without the written consent of the then Administrative Agent. Any such waiver and any such amendment, supplement or modification shall apply equally to each of the Banks and shall be binding upon the Parent, the Borrowers, the Banks, the Administrative Agent, all future holders of the Notes, if any, and all future obligees under the Loans. In the case of any waiver, the Parent, the Borrowers, the Banks and the

Administrative Agent shall be restored to their former position and rights hereunder and under the outstanding Loans or Notes, if any, and any other Loan Documents, and any Default or Event of Default waived shall be deemed to be cured and not continuing; but no such waiver shall extend to any subsequent or other Default or Event of Default, or impair any right consequent thereon.

(b) Notwithstanding anything to the contrary contained in subsection 13.1(a), so long as no Default or Event of Default has occurred and is continuing the Borrowers and the Parent shall be permitted in their discretion (but, if any Revolving Credit Loans are then outstanding, with the consent of the Majority Banks (which consent shall not be unreasonably withheld)) to amend this Agreement to replace one or more Banks without the consent of any Bank to be so replaced pursuant to this subsection 13.1(b) (a "Replaced Bank") and to provide for (w) the termination of the Commitments of such Replaced Bank, (x) the addition to this Agreement of one or more other banking institutions, or an increase in the Commitments of one or more of the other Banks (with the consent of such other Banks), so that the total Commitments after giving effect to such amendment shall be in the same amount as the total Commitments immediately before giving effect to such amendment, (y) if any Loans are outstanding at the time of such amendment, the making of such additional Loans by such new financial institutions or other Bank or Banks, as the case may be, as may be necessary to repay in full the outstanding Loans of such Replaced Bank together with interest thereon and all accrued fees and indemnities with respect thereto immediately before giving effect to such amendment and (z) such other modifications to this Agreement as may be necessary to effect the replacement of such Replaced Bank.

(c) Notwithstanding anything to the contrary contained in paragraph (a) or (b) of this subsection 13.1, if as a result of a change in any Requirement of Law after the date hereof any Borrower or the Parent has become obligated to, or reasonably believes that it will become obligated to pay to any Bank any increased amount pursuant to subsection 5.11, 5.12 or 5.13, and such Bank shall not have waived payment of such increased amounts, then the Borrowers and the Parent may, if no Default or Event of Default has occurred and is continuing and payment of any such increased amounts as have become due has been made or appropriately provided for, upon five Business Days' notice to the Administrative Agent and such Bank, amend this Agreement, without the consent of any Bank or the Administrative Agent, to replace any one or more of the Banks to which such increased amounts have become payable or would become payable and to provide for the matters referred to in clauses (w), (x), (y) and (z) of subsection 13.1(b), and such replaced Bank or Banks shall be deemed to be Replaced Banks for purposes of such clauses.

13.2 Notices. All notices, requests and demands to or upon the respective parties hereto to be effective shall be in writing (including by telecopy), and, unless otherwise expressly

provided herein, shall be deemed to have been duly given or made on receipt, addressed as follows in the case of the Company, the Parent and the Administrative Agent, as set forth in paragraph 5 of the Notice of Additional Borrower relating to any Borrower other than the Company, in the case of such other Borrower, and as set forth in Schedule I in the case of the other parties hereto, or to such other address as may be hereafter notified by the respective parties hereto and any future holders of the Notes, if any, or any future obligees under the Loans:

> The Company: W. R. Grace & Co.-Conn

One Town Center Road

Boca Raton, Florida 33486-1010 Attention: Treasurer Telecopy: (561) 362-1944 Telephone: (561) 362-1949

The Parent: W. R. Grace & Co.

One Town Center Road

Boca Raton, Florida 33486-1010

Attention: Treasurer Telecopy: (561) 362-1944 Telephone: (561) 362-1949

The Administrative

Agent: The Chase Manhattan Bank

270 Park Avenue

New York, New York 10017 Attention: Robert Sacks Telecopy: (212) 270-4118 Telephone: (212) 270-1355

With a copy to: Loan and Agency Services Group

One Chase Manhattan Plaza, 8th Floor New York, New York 10081

Attention: Margaret Swales Telecopy: (212) 552-5662 Telephone: (212) 552-7472

13.3 No Waiver; Cumulative Remedies. No failure to exercise and no delay in exercising, on the part of the Administrative Agent or any Bank, any right, remedy, power or privilege hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. The rights, remedies, powers and privileges herein provided are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law.

13.4 Survival of Representations and Warranties. All representations and warranties made hereunder and in any document, certificate or statement delivered pursuant hereto or in connection herewith shall survive the execution and delivery of this Agreement and the Notes, if any.

13.5 Payment of Expenses and Taxes. The Company agrees (a) to pay or reimburse the Administrative Agent for all its out-of-pocket costs and expenses incurred in connection with the development, preparation and execution of, and any amendment, supplement or modification to, this Agreement and any Notes and the other Loan Documents and any other documents prepared in connection herewith or therewith, and the consummation and administration of the transactions contemplated hereby and thereby, including, without limitation, the fees and disbursements of counsel to the Administrative Agent, (b) to pay or reimburse each Bank and the Administrative Agent for all its costs and expenses incurred in connection with the enforcement or preservation of any rights under this Agreement, any Notes, the other Loan Documents and any such other documents, including, without limitation, fees and disbursements of counsel to the Administrative Agent and to the several Banks, and (c) to pay, indemnify, and hold each Bank and the Administrative Agent harmless from, any and all recording and filing fees and any and all liabilities with respect to, or resulting from any delay in paying, stamp, excise and other transactional taxes, if any, which may be payable or determined to be payable in connection with the execution and delivery of, or consummation or administration of any of the transactions contemplated by, or any amendment, supplement or modification of, or any waiver or consent under or in respect of, this Agreement, any Notes, the other Loan Documents and any such other documents, and (d) to pay, indemnify, and hold each Bank and the Administrative Agent harmless from and against any and all other liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind or nature whatsoever with respect to the execution, delivery and performance by the Loan Parties, and administration and enforcement by the Administrative Agent and the Banks of this Agreement, any Notes and the other Loan Documents and any such other documents (all the foregoing, collectively, the "indemnified liabilities"), provided, that the Company shall have no obligation hereunder to the Administrative Agent or any Bank with respect to indemnified liabilities arising from (i) the gross negligence or willful misconduct of the Administrative Agent or any such Bank, (ii) legal proceedings commenced against the Administrative Agent or any such Bank by any security holder or creditor thereof arising out of and based upon rights afforded any such security holder or creditor solely in its capacity as such, or (iii) legal proceedings commenced against the Administrative Agent or any such Bank by any other Bank or by any Transferee (as defined in subsection 13.6). The agreements in this subsection shall survive repayment of the Loans or Notes, if any, and all other amounts payable hereunder.

13.6 Successors and Assigns; Participations;

Purchasing Banks.

- (a) This Agreement shall be binding upon and inure to the benefit of the Parent, the Borrowers, the Banks, the Administrative Agent, all future holders of the Notes, if any, all future obligees under the Loans and their respective successors and assigns, except that neither the Parent nor any Borrower may assign or transfer any of its rights or obligations under this Agreement without the prior written consent of each Bank.
- (b) Any Bank may, in the ordinary course of its commercial banking business and in accordance with applicable law, at any time sell to one or more banks or other entities ("Participants") participating interests in any Loan owing to such Bank, any Note held by such Bank, any Commitments of such Bank or any other interest of such Bank hereunder and under the other Loan Documents. In the event of any such sale by a Bank of participating interests to a Participant, such Bank's obligations under this Agreement to the other parties to this Agreement shall remain unchanged, such Bank shall remain solely responsible for the performance thereof, such Bank shall remain the holder of any such Note, if any, and the obligee under any such Loan for all purposes under this Agreement and the other Loan Documents, and the Parent, the Borrowers and the Administrative Agent shall continue to deal solely and directly with such Bank in connection with such Bank's rights and obligations under this Agreement and the other Loan Documents. The Parent and each of the Borrowers agrees that if amounts outstanding under this Agreement and the Loans or the Notes, if any, are due or unpaid, or shall have been declared or shall have become due and payable upon the occurrence of an Event of Default, each Participant shall be deemed to have the right of set-off in respect of its participating interest in amounts owing under this Agreement and any Loan or Note to the same extent as if the amount of its participating interest were owing directly to it as a Bank under this Agreement or any Loan or Note, provided that such Participant shall only be entitled to such right of set-off if it shall have agreed in the agreement pursuant to which it shall have acquired its participating interest to share with the Banks the proceeds thereof as provided in subsection 13.7. The Parent and each of the Borrowers also agrees that each Participant shall be entitled to the benefits of subsections 5.11, 5.12, 5.13 and 13.5 with respect to its participation in the Commitments and the Loans outstanding from time to time; provided, that no Participant shall be entitled to receive any greater amount pursuant to such subsections than the transferor Bank would have been entitled to receive in respect of the amount of the participation transferred by such transferor Bank to such Participant had no such transfer occurred.

(c) Any Bank may, in the ordinary course of its commercial banking business and in accordance with applicable law, at any time sell to any Bank or any affiliate thereof and, with the consent of the Company and upon notice to the

Administrative Agent, to one or more additional banks or financial institutions ("Purchasing Banks") all or any part of its rights and obligations under this Agreement and the Loans or the Notes, if any, pursuant to a Commitment Transfer Supplement, substantially in the form of Exhibit H, executed by such Purchasing Bank, such transferor Bank (and, in the case of a Purchasing Bank that is not then a Bank or an affiliate thereof, by the Company and the Administrative Agent) and delivered to the Administrative Agent for its acceptance and recording in the Register. Upon such execution, delivery, acceptance and recording, from and after the Transfer Effective Date determined pursuant to such Commitment Transfer Supplement, (x) the Purchasing Bank thereunder shall be a party hereto and, to the extent provided in such Commitment Transfer Supplement, have the rights and obligations of a Bank hereunder with a Commitment as set forth therein, and (y) the transferor Bank thereunder shall, to the extent provided in such Commitment Transfer Supplement, be released from its obligations under this Agreement (and, in the case of a Commitment Transfer Supplement covering all or the remaining portion of a transferor Bank's rights and obligations under this Agreement, such transferor Bank shall cease to be a party hereto). Such Commitment Transfer Supplement shall be deemed to amend this Agreement to the extent, and only to the extent, necessary to reflect the addition of such Purchasing Bank and the resulting adjustment of Commitment Percentages arising from the purchase by such Purchasing Bank of all or a portion of the rights and obligations of such transferor Bank under this Agreement and the Loan or the Notes, if any. On or prior to the Transfer Effective Date determined pursuant to such Commitment Transfer Supplement, the relevant Borrower, at its own expense, if the Purchasing Bank so requests, shall execute and deliver to the Administrative Agent in exchange for any surrendered Revolving Credit Note and Bid Loan Note a new Revolving Credit Note and Bid Loan Note to the order of such Purchasing Bank in an amount equal to the Commitment assumed by it pursuant to such Commitment Transfer Supplement and, if the transferor Bank has retained a Commitment hereunder, new Notes to the order of the transferor Bank in an amount equal to the Commitment retained by it hereunder. Such new Notes shall be dated the Closing Date and shall otherwise be in the form of the Notes replaced thereby. Any Notes surrendered by the transferor Bank shall be returned by the Administrative Agent to the Company marked "cancelled".

(d) The Administrative Agent shall maintain at its address referred to in subsection 13.2 a copy of each Commitment Transfer Supplement delivered to it and a register (the "Register") for the recordation of the names and addresses of the Banks and the Commitment of, and principal amount of the Loans owing to, each Bank from time to time. The entries in the Register shall be conclusive, in the absence of manifest error, and the Parent, the Borrowers, the Administrative Agent and the Banks may treat each Person whose name is recorded in the Register as the owner of the Loan recorded therein for all purposes of this Agreement. The Register shall be available for inspection by the Parent, the Borrowers or any Bank at any

reasonable time and from time to time upon reasonable prior notice.

- (e) Upon its receipt of a Commitment Transfer Supplement executed by a transferor Bank and Purchasing Bank (and, in the case of a Purchasing Bank that is not then a Bank or an affiliate thereof, by the Company and the Administrative Agent) together with payment to the Administrative Agent of a registration and processing fee of \$3,500, the Administrative Agent shall (i) promptly accept such Commitment Transfer Supplement (ii) on the Transfer Effective Date determined pursuant thereto record the information contained therein in the Register and give notice of such acceptance and recordation to the Banks and the Company.
- (f) The Parent and the Borrowers authorizes each Bank to disclose to any Participant or Purchasing Bank (each, a "Transferee") and any prospective Transferee any and all financial information in such Bank's possession concerning such Borrower and its affiliates which has been delivered to such Bank by or on behalf of the Parent, the Company or such Borrower pursuant to this Agreement or which has been delivered to such Bank by or on behalf of the Parent, the Company or such Borrower in connection with such Bank's credit evaluation of such Borrower and its affiliates prior to becoming a party to this Agreement.
- (g) If, pursuant to this subsection, any interest in this Agreement or any Note is transferred to any Transferee which is organized $\,$ under the laws of any jurisdiction other than the United States or any state thereof, the transferor Bank shall cause such Transferee, concurrently with the effectiveness of such transfer, (i) to represent to the transferor Bank (for the benefit of the transferor Bank, the Administrative Agent, the Parent and the Borrowers) that under applicable law and treaties no taxes will be required to be withheld by the Administrative Agent, the Parent, the Borrowers or the transferor Bank with respect to any payments to be made to such Transferee in respect of the Loans, (ii) to furnish to the transferor Bank (and, in the case of any Purchasing Bank registered in the Register, the Administrative Agent, the Parent and the Company) either U.S. Internal Revenue Service Form 4224 or U.S. Internal Revenue Service Form 1001 (wherein such Transferee claims entitlement to complete exemption from U.S. federal withholding tax on all interest payments hereunder) and (iii) to agree (for the benefit of the transferor Bank, the Administrative Agent, the Parent and the Company) to provide the transferor Bank (and, in the case of any Purchasing Bank registered in the Register, the Administrative Agent, the Parent and the Company) a new Form 4224 or Form 1001 upon the expiration or obsolescence of any previously delivered form and comparable statements in accordance with applicable U.S. laws and regulations and amendments duly executed and completed by such Transferee, and to comply from time to time with all applicable U.S. laws and regulations with regard to such withholding tax exemption.

(h) Nothing herein shall prohibit any Bank from pledging or assigning any Note to any Federal Reserve Bank in accordance with applicable law.

13.7 Adjustments; Set-off.

- (a) If any Bank (a "benefitted Bank") shall at any time receive any payment of all or part of its Revolving Credit Loans, or interest thereon, or receive any collateral in respect thereof (whether voluntarily or involuntarily, by set-off, pursuant to events or proceedings of the nature referred to in Section 10(f), or otherwise), in a greater proportion than any such payment to or collateral received by any other Bank, if any, in respect of such other Bank's Revolving Credit Loans, or interest thereon, such benefitted Bank shall purchase for cash from the other Banks such portion of each such other Bank's Loan, or shall provide such other Banks with the benefits of any such collateral, or the proceeds thereof, as shall be necessary to cause such benefitted Bank to share the excess payment or benefits of such collateral or proceeds ratably with each of the Banks; provided, however, that if all or any portion of such excess payment or benefits is thereafter recovered from such benefitted Bank, such purchase shall be rescinded, and the purchase price and benefits returned, to the extent of such recovery, but without interest. Each Borrower agrees that each Bank so purchasing a portion of another Bank's Loan may exercise all rights of payment (including, without limitation, rights of set-off) with respect to such portion as fully as if such Bank were the direct holder of such portion.
- (b) In addition to any rights and remedies of the Banks provided by law, each Bank shall have the right, without prior notice to the Parent and the Borrowers, any such notice being expressly waived by the Parent and the Borrowers, to the extent permitted by applicable law, upon any amount not being paid when due and payable by any Borrower hereunder or under the Notes (whether at the stated maturity, by acceleration or otherwise) to set-off and appropriate and apply against such amount any and all deposits (general or special, time or demand, provisional or final), in any currency, and any other credits, indebtedness or claims, in any currency, in each case whether direct or indirect, absolute or contingent, matured or unmatured, at any time held or owing by such Bank or any branch or agency thereof to or for the credit or the account of the Parent or such Borrower. Each Bank agrees promptly to notify the Parent, the Borrowers and the Administrative Agent after any such set-off and application made by such Bank, provided that the failure to give such notice shall not affect the validity of such set-off and application.
- 13.8 Counterparts. This Agreement may be executed by one or more of the parties to this Agreement on any number of separate counterparts, and all of said counterparts taken together shall be deemed to constitute one and the same instrument. A set of the copies of this Agreement signed by all

the parties shall be lodged with the Parent, the Company and the $\operatorname{Administrative}$ Agent.

- 13.9 Severability. Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.
- 13.10 Integration. This Agreement represents the agreement of the Parent, each Borrower, the Administrative Agent and the Banks with respect to the subject matter hereof, and there are no promises, undertakings, representations or warranties by the Administrative Agent or any Bank relative to subject matter hereof not expressly set forth or referred to herein, in the other Loan Documents or in any documentation entered into pursuant to subsection 3.1(b).
- 13.11 GOVERNING LAW. THIS AGREEMENT (INCLUDING SECTION 12) AND THE NOTES AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES UNDER THIS AGREEMENT AND THE NOTES SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK.
- $\,$ 13.12 Submission to Jurisdiction; Waivers. (a) The Parent, each Borrower, the Administrative Agent and the Banks hereby irrevocably and unconditionally:
 - (i) submits for itself and its property in any legal action or proceeding relating to this Agreement and the other Loan Documents to which it is a party, or for recognition and enforcement of any judgment in respect thereof, to the non-exclusive general jurisdiction of the Courts of the State of New York sitting in New York County, the courts of the United States of America for the Southern District of New York, and the appellate courts from any thereof;
 - (ii) consents that any such action or proceeding may be brought in such courts and waives any objection that it may now or hereafter have to the venue of any such action or proceeding in any such court or that such action or proceeding was brought in an inconvenient court and agrees not to plead or claim the same;
 - (iii) agrees that service of process in any such action or proceeding may be effected by mailing a copy thereof by registered or certified mail (or any substantially similar form of mail), postage prepaid, to the Parent or such Borrower at its address set forth in subsection 13.2 or, with respect to Borrowers other than the Company, the Notice of Additional Borrower relating to such Borrower or at such

other address of which the Administrative Agent shall have been notified pursuant thereto;

- (iv) agrees that nothing herein shall affect the right to effect service of process in any other manner permitted by law or shall limit the right to sue in any other jurisdiction; and
- (v) waives, to the maximum extent not prohibited by law, any right it may have to claim or recover in any legal action or proceeding referred to in this subsection any special, exemplary, punitive or consequential damages.
- (b) Each Borrower other than the Company hereby appoints and empowers the Parent and the Company, 1114 Avenue of the Americas, New York, New York 10036-7794, Attention: Treasurer, as its authorized agent (the "Process Administrative Agent") to receive on behalf of such Borrower service of any and all process and documents in any such legal action or proceeding brought in a New York state or federal court sitting in New York City. It is understood that a copy of such process served on the Process Administrative Agent will be promptly hand delivered or mailed (by registered or certified airmail if available), postage prepaid, to such Borrower at its address set forth in paragraph 5 of such Borrower's Notice of Additional Borrower, but the failure of such Borrower to receive such copy shall not affect in any way the service of such process on the Process Administrative Agent. If the Process Administrative Agent shall refuse or be prevented from acting as agent, notice thereof shall immediately be given by such Borrowers to the Administrative Agent by registered or certified airmail (if available), postage prepaid, and such Borrowers agree promptly to designate another agent in New York City, satisfactory to the Administrative Agent, to serve in place of the Process Administrative Agent and deliver to the Administrative Agent written evidence of such substitute agent's acceptance of such designation.
- 13.13 Acknowledgments. The Parent, each Borrower, the Administrative Agent and the Banks hereby acknowledge that:
 - (a) it has been advised by counsel in the negotiation, execution and delivery of this Agreement and the Notes and the other Loan Documents;
 - (b) neither the Administrative Agent nor any Bank has any fiduciary relationship with or duty to the Parent or such Borrower, as the case may be, arising out of or in connection with this Agreement or any of the other Loan Documents, and the relationship between Administrative Agent and Banks, on one hand, and the Parent and the Borrowers, on the other hand, in connection herewith or therewith is solely that of debtor and creditor; and

- (c) as to any matter relating to any Loan Documents, no joint venture exists among the Banks or among the Parent, the Borrowers and the Banks.
- 13.14 WAIVERS OF JURY TRIAL. THE PARENT, THE BORROWERS, THE ADMINISTRATIVE AGENT AND THE BANKS HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVE TRIAL BY JURY IN ANY LEGAL ACTION OR PROCEEDING RELATING TO THIS AGREEMENT OR THE NOTES OR ANY OTHER LOAN DOCUMENT AND FOR ANY COUNTERCLAIM THEREIN.
- 13.15 Additional Borrowers. (a) Any Subsidiary of the Company shall have the right to become a "Borrower" hereunder, and to borrow hereunder subject to the terms and conditions hereof applicable to a Borrower and to the following additional conditions:
 - (i) the Company shall deliver a notice in substantially the form of Exhibit I hereto (a "Notice of Additional Borrower") signed by such Subsidiary and countersigned by the Parent and the Company to the Administrative Agent and the Banks stating that such Subsidiary desires to become a "Borrower" under this Agreement and agrees to be bound by the terms hereof. From the time of receipt of such Notice of Additional Borrower by the Administrative Agent and the Banks and subject to the satisfaction of each condition precedent contained in such Notice of Additional Borrower, such Subsidiary shall be a "Borrower" hereunder with all of the rights and obligations of a Borrower hereunder; provided, however, that the Company may revoke a Notice of Additional Borrower with respect to any Subsidiary (other than the Company) upon five Business Days' written notice to the Administrative Agent, so long as such Borrower has no Obligations outstanding. No Notice of Additional Borrower relating to a Subsidiary may be revoked as to amounts owed by such Subsidiary to the Banks under this Agreement or any Notes or when an irrevocable notice pursuant to subsection 2.3, or a notice of acceptance pursuant to subsection 3.1 or 4.2, has been given by such Subsidiary as a Borrower and is effective;
 - (ii) if such Subsidiary is a Foreign Subsidiary, if reasonably requested by the Majority Banks, such Notice of Additional Borrower shall be accompanied by an opinion of counsel for such Subsidiary as specified in paragraph 4(a)(ii) of such Notice of Additional Borrower;
 - (iii) and the other conditions set forth in such Notice of Additional Borrower shall have been satisfied (including the representations and warranties contained therein being true and correct as of the date thereof).
- (b) Promptly, upon receipt of any Notice of Additional Borrower by the Administrative Agent, the Administrative Agent shall notify each Bank thereof, and shall deliver to each Bank

copies of each document delivered to the Administrative Agent pursuant to such Notice of Additional Borrower.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed and delivered in New York, New York by their proper and duly authorized officers as of the day and year first above written.

W. R. GRACE & CO.-CONN.

By: /s/ Leonard R. Byrne

·-----

Name: Leonard R. Byrne

Title: President-

Authorized Representative

W. R. GRACE & CO.

By: /s/ Bruce L. Hagins

Name: Bruce L. Hagins

Title: Authorized Representative

THE CHASE MANHATTAN BANK, as Administrative Agent and as a Bank

By: /s/ Robert T. Sacks

Name: Robert T. Sacks Title: Managing Director

NATIONSBANK, N.A. (SOUTH), as Documentation Agent and as a Bank

By: /s/ Andrew M. Airheart

Name: Andrew M. Airheart Title: Senior Vice President ABN AMRO BANK N.V., NEW YORK BRANCH

By: /s/ David W. Stack

Name: David W. Stack

Title: Vice President

By: /s/ George M. Dugan

Name: George M. Dugan Title: Vice President

BANK OF AMERICA NATIONAL TRUST AND SAVINGS ASSOCIATION

By: /s/ Laurens F. Schaad, Jr.

Name: Laurens F. Schaad, Jr.

Title: Vice President

THE BANK OF NEW YORK

By: /s/ David C. Siegel

Name: David C. Ciagal

Name: David C. Siegel Title: Assistant Vice President

THE BANK OF NOVA SCOTIA

By: /s/ William Brown

Name: William Brown Title: Vice President

BARCLAYS BANK PLC

By: /s/ George E. McKean

Name: Geroge E. McKean

Title: Director

CITIBANK, N.A.

By: /s/ Mary Corkran

Name: Mary Corkran Title: Vice President

COMMERZBANK AG, ATLANTA AGENCY

By: /s/ W.D. Suttles Name: W.D. Suttles Title: Vice President By: /s/ Mary B. Smith Name: Mary B. Smith Title: Asst. Vice President CREDIT LYONNAIS ATLANTA AGENCY By: /s/ David M. Cawrse Name: David M. Cawrse Title: First Vice President CREDIT SUISSE FIRST BOSTON By: /s/ James P. Moran -----Name: James P. Moran Title: Director By: /s/ J. Scott Karro Name: J. Scott Karro Title: Associate DRESDNER BANK AG, NEW YORK AND GRAND CAYMAN BRANCHES By: /s/ Andrew P. Nesi Name: Andrew P. Nesi Title: Vice President By: /s/ John D. Padilla

Name: John D. Padilla Title: Vice President

FIRST UNION NATIONAL BANK OF FLORIDA

By: /s/ Mary A. Morgan

Name: Mary A. Morgan Title: Vice President & Senior Portfolio Manager

MARINE MIDLAND BANK

By: /s/ William M. Holland

Name: William M. Holland Title: Vice President

MORGAN GUARANTY TRUST COMPANY OF NEW YORK

By: /s/ Jeffrey Hwang

ama. Jaffray Ilyana

Name: Jeffrey Hwang Title: Vice President

UNION BANK OF SWITZERLAND - NEW YORK

By: /s/ Hamilton W. Bullard

Name: Hamilton W. Bullard Title: Assistant Treasurer

By: /s/ Samuel Azizo

Name: Samuel Azizo Title: Vice President

71

Commitments

The Chase Manhattan Bank \$83,000,000 ABN AMRO Bank N.V., New York Branch \$42,000,000 Bank of America National Trust & Savings Association \$42,000,000 The Bank of New York \$42,000,000 The Bank of Nova Scotia \$42,000,000 Barclays Bank PLC \$42,000,000 Citibank, N.A. \$21,200,000 Commerzbank AG, Atlanta Agency \$42,000,000 Credit Lyonnais Atlanta Agency \$21,200,000 Credit Suisse First Boston \$21,200,000 Dresdner Bank AG, New York and Grand Cayman Branches \$42,000,000 First Union National Bank of Florida \$42,000,000 Morgan Guaranty Trust Company of New York \$42,000,000 Marine Midland Bank \$21,200,000 NationsBank, N.A. (South) \$83,000,000 Union Bank of Switzerland \$21,200,000 TOTAL \$650,000,000	BANK	COMMITMENT
Bank of America National Trust & Savings Association The Bank of New York \$42,000,000 The Bank of Nova Scotia \$42,000,000 Barclays Bank PLC \$42,000,000 Citibank, N.A. \$21,200,000 Commerzbank AG, Atlanta Agency \$42,000,000 Credit Lyonnais Atlanta Agency \$21,200,000 Credit Suisse First Boston \$21,200,000 Dresdner Bank AG, New York and Grand Cayman Branches First Union National Bank of Florida Morgan Guaranty Trust Company of New York Marine Midland Bank \$21,200,000 NationsBank, N.A. (South) Union Bank of Switzerland \$21,200,000 \$650,000,000	The Chase Manhattan Bank	\$83,000,000
& Savings Association \$42,000,000 The Bank of New York \$42,000,000 The Bank of Nova Scotia \$42,000,000 Barclays Bank PLC \$42,000,000 Citibank, N.A. \$21,200,000 Commerzbank AG, Atlanta Agency \$42,000,000 Credit Lyonnais Atlanta Agency \$21,200,000 Credit Suisse First Boston \$21,200,000 Dresdner Bank AG, New York and Grand Cayman Branches \$42,000,000 First Union National Bank of Florida \$42,000,000 Morgan Guaranty Trust Company of New York \$42,000,000 Marine Midland Bank \$21,200,000 NationsBank, N.A. (South) \$83,000,000 Union Bank of Switzerland \$21,200,000 TOTAL \$650,000,000	ABN AMRO Bank N.V., New York Branch	\$42,000,000
The Bank of Nova Scotia \$42,000,000 Barclays Bank PLC \$42,000,000 Citibank, N.A. \$21,200,000 Commerzbank AG, Atlanta Agency \$42,000,000 Credit Lyonnais Atlanta Agency \$21,200,000 Credit Suisse First Boston \$21,200,000 Dresdner Bank AG, New York and Grand Cayman Branches \$42,000,000 First Union National Bank of Florida \$42,000,000 Morgan Guaranty Trust Company of New York \$42,000,000 Marine Midland Bank \$21,200,000 NationsBank, N.A. (South) \$83,000,000 Union Bank of Switzerland \$21,200,000		\$42,000,000
Barclays Bank PLC \$42,000,000 Citibank, N.A. \$21,200,000 Commerzbank AG, Atlanta Agency \$42,000,000 Credit Lyonnais Atlanta Agency \$21,200,000 Credit Suisse First Boston \$21,200,000 Dresdner Bank AG, New York and Grand Cayman Branches \$42,000,000 First Union National Bank of Florida \$42,000,000 Morgan Guaranty Trust Company of New York \$42,000,000 Marine Midland Bank \$21,200,000 NationsBank, N.A. (South) \$83,000,000 Union Bank of Switzerland \$21,200,000 TOTAL \$650,000,000	The Bank of New York	\$42,000,000
Citibank, N.A. \$21,200,000 Commerzbank AG, Atlanta Agency \$42,000,000 Credit Lyonnais Atlanta Agency \$21,200,000 Credit Suisse First Boston \$21,200,000 Dresdner Bank AG, New York and Grand Cayman Branches \$42,000,000 First Union National Bank of Florida \$42,000,000 Morgan Guaranty Trust Company of New York \$42,000,000 Marine Midland Bank \$21,200,000 NationsBank, N.A. (South) \$83,000,000 Union Bank of Switzerland \$21,200,000 TOTAL \$650,000,000	The Bank of Nova Scotia	\$42,000,000
Commerzbank AG, Atlanta Agency \$42,000,000 Credit Lyonnais Atlanta Agency \$21,200,000 Credit Suisse First Boston \$21,200,000 Dresdner Bank AG, New York and Grand Cayman Branches \$42,000,000 First Union National Bank of Florida \$42,000,000 Morgan Guaranty Trust Company of New York \$42,000,000 Marine Midland Bank \$21,200,000 NationsBank, N.A. (South) \$83,000,000 Union Bank of Switzerland \$21,200,000	Barclays Bank PLC	\$42,000,000
Credit Lyonnais Atlanta Agency \$21,200,000 Credit Suisse First Boston \$21,200,000 Dresdner Bank AG, New York and Grand Cayman Branches \$42,000,000 First Union National Bank of Florida \$42,000,000 Morgan Guaranty Trust Company of New York \$42,000,000 Marine Midland Bank \$21,200,000 NationsBank, N.A. (South) \$83,000,000 Union Bank of Switzerland \$21,200,000	Citibank, N.A.	\$21,200,000
Credit Suisse First Boston \$21,200,000 Dresdner Bank AG, New York and Grand Cayman Branches \$42,000,000 First Union National Bank of Florida \$42,000,000 Morgan Guaranty Trust Company of New York \$42,000,000 Marine Midland Bank \$21,200,000 NationsBank, N.A. (South) \$83,000,000 Union Bank of Switzerland \$21,200,000 TOTAL \$650,000,000	Commerzbank AG, Atlanta Agency	\$42,000,000
Dresdner Bank AG, New York and Grand Cayman Branches \$42,000,000 First Union National Bank of Florida \$42,000,000 Morgan Guaranty Trust Company of New York \$42,000,000 Marine Midland Bank \$21,200,000 NationsBank, N.A. (South) \$83,000,000 Union Bank of Switzerland \$21,200,000 TOTAL \$650,000,000	Credit Lyonnais Atlanta Agency	\$21,200,000
Grand Cayman Branches \$42,000,000 First Union National Bank of Florida \$42,000,000 Morgan Guaranty Trust Company of New York \$42,000,000 Marine Midland Bank \$21,200,000 NationsBank, N.A. (South) \$83,000,000 Union Bank of Switzerland \$21,200,000 TOTAL \$650,000,000	Credit Suisse First Boston	\$21,200,000
Morgan Guaranty Trust Company of New York \$42,000,000 Marine Midland Bank \$21,200,000 NationsBank, N.A. (South) \$83,000,000 Union Bank of Switzerland \$21,200,000 TOTAL \$650,000,000		\$42,000,000
of New York \$42,000,000 Marine Midland Bank \$21,200,000 NationsBank, N.A. (South) \$83,000,000 Union Bank of Switzerland \$21,200,000 TOTAL \$650,000,000	First Union National Bank of Florida	\$42,000,000
NationsBank, N.A. (South) \$83,000,000 Union Bank of Switzerland \$21,200,000 TOTAL \$650,000,000		\$42,000,000
Union Bank of Switzerland \$21,200,000 TOTAL \$650,000,000	Marine Midland Bank	\$21,200,000
TOTAL \$650,000,000	NationsBank, N.A. (South)	\$83,000,000
	Union Bank of Switzerland	\$21,200,000
	TOTAL	

CREDIT AGREEMENT

DATED AS OF MAY 16, 1997

AMONG

W. R. GRACE & CO.-CONN., W. R. GRACE & CO.,

The Banks Party Hereto,

and

THE CHASE MANHATTAN BANK, AS ADMINISTRATIVE AGENT

TABLE OF CONTENTS

		Page
SECTION 1. 1. 1.	1 Defined	ITIONS
SECTION 2. 2. 2.	1 Commitr 2 Obligat	AND TERMS OF COMMITMENTS
2.		ure for Revolving Credit Borrowing
SECTION 3. 3. 3.	1 Request 2 Reports	RAL OPTION LOANS
3. 3.	3 Judgmer	nt Currency
SECTION 4. 4. 4. 4. 4.	1 The Bio 2 Procedu 3 Repaymo 4 Interes	ANS
SECTION 5. 5. 5.	1 Interes 2 Facilit 3 Termina	ACILITY COMMON PROVISIONS
5. 5. 5. 5. 5. 5. 5.	4 Prepayi 5 Convers 6 Minimur 7 Computa 8 Inabil: 9 Pro Rai 10 Illega: 11 Require 12 Taxes.	Control Date 24 ments 25 sion and Continuation Options 26 n Amounts of Eurodollar Tranches 26 ation of Interest and Fees 26 ity to Determine Interest Rate 27 ta Treatment and Payments 27 Lity 29 ements of Law 30 31 ity 32
SECTION 6. 6. 6.	1 Corpora	ENTATIONS AND WARRANTIES
6. 6. 6. 6.	No Lega No Mate Ownersl Financ: Disclos	igations 33 al Bar 34 erial Litigation 34 nip of Properties 34 ial Condition 34 sure of Contingent Liabilities 35 35

i

		Pa	age
	6.9 6.10 6.11 6.12 6.13 6.14	Certain Federal Regulations. No Default. Taxes. Investment Company Act; Other Regulations. Purpose of Loans. Environmental Matters. Principal Subsidiaries.	35 35 36 36 36
SECTION	7. 7.1 7.2	CONDITIONS PRECEDENT	36
SECTION	8. 8.1 8.2 8.3 8.4	AFFIRMATIVE COVENANTS	39 39 40
	8.5 8.6	Insurance	41
	8.7 8.8	Discussions	41
SECTION	9. 9.1 9.2 9.3 9.4	NEGATIVE COVENANTS Financial Condition Covenants Limitation on Liens Limitation on Fundamental Changes Limitation on Asset Transfers to Foreign Subsidiaries	43 43 45
	9.5	Limitation on Subordinated Debt	
SECTION	10.	EVENTS OF DEFAULT	46
SECTION	11. 11.1 11.2 11.3 11.4 11.5	THE ADMINISTRATIVE AGENT	49 49 49 50
	11.7 11.8 11.9	Banks	51 52
SECTION	12. 12.1 12.2 12.3	GUARANTEES	52
	12.4	Indemnity	53
		Obligations	54

12.5 12.6 12.7	Guarantee Absolute and Unconditional 54 Reinstatement 55 Payments 56
SECTION 13.	MISCELLANEOUS
13.1	Amendments and Waivers; Replacement of Banks 56
13.2	Notices 57
13.3	No Waiver; Cumulative Remedies 58
13.4	Survival of Representations and Warranties 58
13.5	Payment of Expenses and Taxes 59
13.6	Successors and Assigns; Participations;
	Purchasing Banks 60
13.7	Adjustments; Set-off
13.8	Counterparts 64
13.9	Severability 64
13.10	Integration 64
13.11	GOVERNING LAW 64
13.12	Submission to Jurisdiction; Waivers 64
13.13	Acknowledgments65
13.14	WAIVERS OF JURY TRIAL 66
13.15	Additional Borrowers

Page

SCHEDULES

Schedule I	Commitments;	Lending	Offices	and	Addresses	for
	Notices					

Schedule II Principal Subsidiaries

EXHIBITS

Exhibit A	Form of Revolving Credit Note				
Exhibit B	Form of Bid Loan Note				
Exhibit C	Form of Bid Loan Confirmation				
Exhibit D	Form of Bid Loan Offer				
Exhibit E	Form of Bid Loan Request				
Exhibit F-1	Form of Opinion of Counsel to the Company and the				
Parent					
Exhibit F-2	Form of Opinion of Simpson Thacher & Bartlett				
Exhibit G	Form of Officer's Certificate				
Exhibit H	Form of Commitment Transfer Supplement				
Exhibit I	Form of Notice of Additional Borrower				

CREDIT AGREEMENT, dated as of May 16, 1997, among W. R. GRACE & CO.-CONN., a Connecticut corporation (the "Company"), W. R. GRACE & CO., a Delaware corporation and sole shareholder of the Company (the "Parent"), the several banks from time to time parties to this Agreement (the "Banks"), and THE CHASE MANHATTAN BANK, a New York banking corporation, as administrative agent for the Banks hereunder (in such capacity, the "Administrative Agent").

The parties hereto hereby agree as follows:

SECTION 1. DEFINITIONS

 ${\bf 1.1}\,$ Defined Terms. As used in this Agreement, the following terms shall have the following meanings:

"ABR Loans": Loans the rate of interest applicable to which is based upon the Alternate Base Rate.

"Affiliate": as to any Person, (a) any other Person (other than a Subsidiary) which, directly or indirectly, is in control of, is controlled by, or is under common control with, such Person or (b) any Person who is a director, officer, shareholder or partner (i) of such Person, (ii) of any Subsidiary of such Person or (iii) of any Person described in the preceding clause (a). For purposes of this definition, "control" of a Person means the power, directly or indirectly, either to (i) vote 10% or more of the securities having ordinary voting power for the election of directors of such Person or (ii) direct or cause the direction of the management and policies of such Person whether by contract or otherwise.

"Aggregate Outstanding Bilateral Option Loans": at any time, (i) the aggregate outstanding principal amount of all Dollar Bilateral Loans and (ii) the aggregate Dollar Equivalents at such time with respect to all outstanding Alternative Currency Bilateral Loans.

"Agreement": this Credit Agreement, as the same may be amended, supplemented or otherwise modified from time to time.

"Alternate Base Rate": for any day, a rate per annum (rounded upwards, if necessary, to the next 1/16 of 1%) equal to the greater of (a) the Prime Rate in effect on such day and (b) the Federal Funds Effective Rate in effect on such day plus 1/2 of 1%. "Prime Rate" shall mean the rate of interest per annum publicly announced from time to time by the Administrative Agent as its prime rate in effect at its principal office in New York City. "Federal Funds Effective Rate" shall mean, for any day, the weighted average of the rates on overnight federal funds transactions

with members of the Federal Reserve System arranged by federal funds brokers, as published on the next succeeding Business Day by the Federal Reserve Bank of New York, or, if such rate is not so published for any day which is a Business Day, the average of the quotations for the day of such transactions received by the Administrative Agent from three federal funds brokers of recognized standing selected by it. If for any reason the Administrative Agent shall have determined (which determination shall be conclusive absent manifest error) that it is unable to ascertain the Federal Funds Effective Rate, for any reason, including, the inability or failure of the Administrative Agent to obtain sufficient quotations in accordance with the terms hereof, the Alternate Base Rate shall be determined without regard to clause (b) of the first sentence of this definition until the circumstances giving rise to such inability no longer exist. Any change in the Alternate Base Rate due to a change in the Prime Rate or the Federal Funds Effective Rate shall be effective on the effective date of such change in the Prime Rate or the Federal Funds Effective Rate, respectively.

"Alternative Currency Bilateral Loan": a Loan made by a Bank to any Borrower in an Alternative Currency pursuant to Section 3.

"Applicable Margin": for any day on which the long term senior unenhanced, unsecured debt of the Company is rated by both S&P and Moody's, the rate per annum under the caption "Margin" (a "Margin Rate") set forth below opposite the S&P and Moody's ratings applicable to such debt on such day (or, if such ratings are set opposite two different Margin Rates, then the Applicable Margin shall be the lower of said two Margin Rates):

Margin	S&P	Moody's
. 400%	BB+ or lower	Ba1 or lower
. 250%	BBB-	ВааЗ
. 215%	BBB	Baa2
.185%	BBB+	Baa1
.175%	A- or higher	A3 or higher

provided that if on any day the long term senior unenhanced, unsecured debt of the Company is rated by only one of either S&P or Moody's, the Applicable Margin will be determined based on the rating by such rating agency, and provided,

further, that if on any day the long term senior unenhanced, unsecured debt of the Company is rated by neither S&P nor Moody's, the Applicable Margin will be determined based on the rating of such debt by Duff & Phelps, Fitch or another nationally recognized statistical rating organization agreed to by and among the Company, the Administrative Agent and the Majority Banks (each, a "Substitute Rating Agency") and will be the Margin Rate set forth above opposite the S&P and Moody's ratings comparable to such Substitute Rating Agency's rating of such debt on such date, and provided, further that if on any day the long term senior unenhanced, unsecured debt of the Company is rated by none of S&P, Moody's or any Substitute Rating Agency, the Company, the Administrative Agent and the Banks will negotiate in good faith to determine an alternative basis for calculating the Applicable Margin consistent with the table set forth above and, if agreement on such alternative basis is not reached within 30 days, the Applicable Margin will be calculated on an alternative basis determined by the Administrative Agent and the Banks in their reasonable discretion consistent with the table above, and until such alternative basis is determined the Applicable Margin will be the Applicable Margin last determined as provided in the

"Available Commitment": as to any Bank at any time, an amount equal to the excess, if any, of (a) the amount of such Bank's Commitment over (b) the Loan Outstandings of such Bank at such time.

"Bid Loan": each Bid Loan made pursuant to Section 4.

"Bid Loan Banks": Banks which have outstanding Bid Loans or which are making Bid Loans.

"Bid Loan Confirmation": each confirmation by the Borrower of its acceptance of Bid Loan Offers, which Bid Loan Confirmation shall be substantially in the form of Exhibit C

"Bid Loan Note": as defined in subsection 4.5(c); collectively, the "Bid Loan Notes".

"Bid Loan Offer": each offer by a Bank to make Bid Loans pursuant to a Bid Loan Request, which Bid Loan Offer shall contain the information specified in Exhibit D.

"Bid Loan Request": each request by a Borrower for Banks to submit bids to make Bid Loans at a fixed rate, which shall contain the information in respect of such requested Bid Loans specified in Exhibit E and shall be delivered to the Administrative Agent.

"Bilateral Option Loan": a Loan made by a Bank to a Borrower pursuant to Section 3. Bilateral Option Loans may be either Dollar Bilateral Loans or Alternative Currency Bilateral Loans.

"Bilateral Option Loan Report": as defined in subsection 3.2. $\,$

"Board": The Board of Governors of the Federal Reserve System of the United States of America or any successor thereto.

"Borrower": the Company and any Subsidiary of the Company with respect to which a Notice of Additional Borrower has been given and all conditions precedent to the effectiveness thereof have been satisfied.

"Borrowing Date": any Business Day specified in a notice pursuant to subsection 2.3 and 4.2, as a date on which a Borrower requests the Banks to make Loans hereunder, or any date that a Bilateral Option Loan is made in accordance with subsection 3.1.

"Business Day": a day other than a Saturday, Sunday or other day on which commercial banks in New York City are authorized or required by law to close.

"Capitalized Lease": any lease of property, real or personal, the obligations of the lessee in respect of which are required to be capitalized in accordance with GAAP.

"Change of Control Date": (i) the first day on which the Company determines that any Person or group of related Persons has direct or indirect beneficial ownership of 30% or more of the outstanding capital stock of the Parent having ordinary voting power (other than stock having such power only by reason of the happening of a contingency) for the election of a majority of the board of directors of the Parent or (ii) the first day on which any Person or group of related Persons shall acquire all or substantially all of the assets of the Parent.

"Chase": The Chase Manhattan Bank.

"Closing Date": the first date on which the conditions set forth in subsection 7.1 have been satisfied or waived.

"Code": the Internal Revenue Code of 1986, as amended from time to time. $\,$

"Commitment": as to any Bank, the obligation of such Bank to make Revolving Credit Loans hereunder to the Borrowers in an aggregate principal amount at any one time outstanding not to exceed the amount set forth opposite such Bank's name on Schedule I under the heading "Commitment".

"Commitment Percentage": as to any Bank at any time, the percentage of the aggregate Commitments then constituted by such Bank's Commitment.

"Commitment Period": the period from and including the date hereof to but not including the Termination Date or such earlier date on which the Commitments shall terminate as provided herein.

"Commonly Controlled Entity": an entity, whether or not incorporated, which is under common control with the Company within the meaning of Section 4001(a)(14) of ERISA or is part of a group which includes the Company and which is treated as a single employer under subsection (b) or (c) of Section 414 of the Code.

"Consolidated Debt": at a particular date, with respect to the Parent and its Subsidiaries, and without duplication, the sum of the amounts set forth on a consolidated balance sheet of the Parent and its Subsidiaries prepared as of such date in accordance with GAAP opposite the captions (1) "Long-term debt" (or the equivalent caption) and (2) "Short-term debt" (or the equivalent caption) but always to include all indebtedness for borrowed money of the Parent and its Subsidiaries in accordance with GAAP.

"Consolidated Interest Expense": for any period, with respect to the Parent and its Subsidiaries, the amount which, in conformity with GAAP, would be set forth opposite the caption "Interest expense and related financing costs" (or the equivalent caption) on a consolidated statement of operations of the Parent and its Subsidiaries for such period.

"Contractual Obligation": as to any Person, any provision of any security issued by such Person or of any agreement, instrument or other undertaking to which such Person is a party or by which it or any of its property is bound.

"Default": any of the events specified in Section 10, whether or not any requirement for the giving of notice, the lapse of time, or both, has been satisfied.

"Dollar Bilateral Loan": a Bilateral Option Loan denominated in Dollars.

"Dollar Equivalent": on any date of determination by the Administrative Agent pursuant to subsection 3.2(b) or 3.2(c), as applicable, in respect of any Alternative Currency Bilateral Loan the amount of Dollars obtained by converting the outstanding amount of currency of such Alterative Currency Bilateral Loan, as specified in the then most recent Bilateral Option Loan Report, into Dollars at

the spot rate for the purchase of Dollars with such currency as quoted by the Administrative Agent at its principal foreign exchange trading operations office in New York City on such date.

"Dollars" and "\$": dollars in lawful currency of the United States of America. $\,$

"Domestic Indebtedness": any Indebtedness of the Parent and any Domestic Subsidiary.

"Domestic Subsidiary": any Subsidiary of the Parent other than a Foreign Subsidiary. $\,$

"Duff & Phelps": Duff & Phelps, Inc.

"EBIT": for any period, with respect to the Parent and its Subsidiaries, (a) all amounts which would be set forth opposite the caption "Income from continuing operations before income taxes" (or the equivalent caption) on a consolidated statement of operations of the Parent and its Subsidiaries prepared in accordance with GAAP for such period plus (b) non-cash pre-tax charges arising from: (1) asset disposals (excluding the retirement of property, plant and equipment in the ordinary course of business) by the Parent and its Subsidiaries, (2) the implementation or modified application of financial accounting standards applicable to the Parent and its Subsidiaries, and (3) other special non-recurring transactions (including charges relating to Restructuring Activities, discontinued operations and asbestos related litigation and claims) (to the extent that such amounts have been deducted in determining the amount set forth opposite the caption "Income from continuing operations" (or the equivalent caption) for such period), plus (c) Consolidated Interest Expense for such period, plus (d) any payments received in such period in respect of non-cash pre-tax gains referred to in clause (e) of this definition, minus (e) non-cash pre-tax gains arising from: (1) asset disposals (excluding the retirement of property, plant and equipment in the ordinary course of business) by the Parent and its Subsidiaries, (2) the implementation or modified application of financial accounting standards applicable to the Parent and its Subsidiaries, and (3) other special non-recurring transactions (including charges relating to Restructuring Activities, discontinued operations and asbestos related litigation and claims) (to the extent that such amounts have been added in determining the amount set forth opposite the caption "Income from continuing operations" (or the equivalent caption) for such period), minus (f) any payments made in such period in respect of non-cash pre-tax charges referred to in clause (b) of this definition.

"EBITDA": for any period, EBIT plus all amounts which would be set forth opposite the caption "Depreciation and $\,$

amortization" (or the equivalent caption) on a consolidated statement of operations of the Parent and its Subsidiaries prepared in accordance with GAAP for such period.

"Environmental Laws": any and all federal, state, local or municipal laws, rules, orders, regulations, statutes, ordinances, codes, decrees or requirements of any Governmental Authority regulating, relating to or imposing liability or standards of conduct concerning environmental protection matters, including without limitation, Hazardous Materials, as now or may at any time hereafter be in effect.

"ERISA": the Employee Retirement Income Security Act of 1974, as amended from time to time.

"Eurocurrency Reserve Requirements": for any day as applied to a Eurodollar Loan, the aggregate (without duplication) of the rates (expressed as a decimal fraction) of any reserve requirements in effect on such day (including, without limitation, basic, supplemental, marginal and emergency reserves under any regulations of the Board of Governors of the Federal Reserve System or other Governmental Authority having jurisdiction with respect thereto) dealing with reserve requirements prescribed for eurocurrency funding (currently referred to as "Eurocurrency Liabilities" in Regulation D of such Board) maintained by a member bank of such System.

"Eurodollar Loans": Loans the rate of interest applicable to which is based upon the Eurodollar Rate.

"Eurodollar Rate": with respect to each day during each Interest Period pertaining to a Eurodollar Loan, the rate per annum equal to the rate at which Chase is offered Dollar deposits at or about 10:00 A.M., New York City time, two Business Days prior to the beginning of such Interest Period in the interbank eurodollar market where the eurodollar and foreign currency and exchange operations in respect of its Eurodollar Loans are then being conducted for delivery on the first day of such Interest Period for the number of days comprised therein and in an amount comparable to the amount of its Eurodollar Loan to be outstanding during such Interest Period.

"Eurodollar Tranche": the collective reference to Eurodollar Loans, the Interest Periods with respect to all of which begin on the same date and end on the same later date (whether or not such Loans shall originally have been made on the same day).

"Event of Default": any of the events specified in Section 10, provided that any requirement for the giving of notice, the lapse of time, or both, has been satisfied.

"Excluded Subsidiaries": GN Holdings, Inc., a Delaware corporation; CCHP, Inc., a Delaware corporation; Assignment America, Inc., a Delaware corporation; Cross Country Staffing, a Delaware general partnership, and each of their respective Subsidiaries.

"Existing Credit Agreement": the Amended and Restated Credit Agreement, dated as of May 17, 1996, among W. R. Grace & Co.-Conn., W. R. Grace & Co., Grace Holding, Inc., the several banks from time to time parties thereto and Chemical Bank, as administrative agent.

"FASB 5": Statement of Financial Accounting Standards No. 5, Accounting for Contingencies, of the Financial Accounting Standards Board, as the same may be from time to time supplemented, amended or interpreted by such Board.

"Fitch": Fitch Investors Service Inc.

"Foreign Subsidiary": any Subsidiary of the Parent (i) that is organized under the laws of any jurisdiction other than any state (including the District of Columbia), territory or possession of the United States of America (a "foreign jurisdiction"), or (ii) more than 50 percent of the book value of the assets of which (as of the end of the most recent fiscal period for which financial statements are required to have been provided pursuant to subsection 8.1(a) or (b)) are located in one or more foreign jurisdictions, or (iii) more than 50 percent of the Net Sales and Revenues of which (for the most recent fiscal year for which financial statements are required to have been provided pursuant to subsection 8.1(a)) were from sales made and/or services provided in one or more foreign jurisdictions, or (iv) more than 50 percent of the book value of the assets of which (as of the end of the most recent fiscal period for which financial statements are required to have been provided pursuant to subsection 8.1(a) or (b)) consists of equity interests in and/or Indebtedness of one or more Subsidiaries that are "Foreign Subsidiaries" within clauses (i), (ii), (iii) or (iv) of this definition.

"Foreign Subsidiary Indebtedness": any Indebtedness of any Foreign Subsidiary.

"GAAP": generally accepted accounting principles in the United States of America in effect from time to time.

"Governmental Authority": any nation or government, any state or other political subdivision thereof and any entity exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government.

"Hazardous Materials": any hazardous materials, hazardous wastes, hazardous constituents, hazardous or toxic substances, petroleum products (including crude oil or any fraction thereof), defined or regulated as such in or under any Environmental

"Indebtedness": of any Person at any date, (a) all indebtedness of such Person for borrowed money or for the deferred purchase price of property or services (other than current trade liabilities incurred in the ordinary course of business and payable in accordance with customary practices) or which is evidenced by a note, bond, debenture or similar instrument, (b) all obligations of such Person under Capitalized Leases, and (c) without duplication, all "loss contingencies" of such Person of the types described in paragraph 12 of FASB 5, whether or not disclosed or required to be disclosed on the financial statements or footnotes thereto of such Person pursuant to GAAP.

"Insolvency": with respect to any Multiemployer Plan, the condition that such Plan is insolvent within the meaning of Section 4245 of ERISA.

"Insolvent": pertaining to a condition of Insolvency.

"Interest Payment Date": (a) as to any ABR Loan, the fifteenth day of each March, June, September and December to occur while such Loan is outstanding and, if different, the Termination Date, and (b) as to any Eurodollar Loan having an Interest Period of three months or less, the last day of such Interest Period, and (c) as to any Eurodollar Loan having an Interest Period longer than three months, if any, as agreed by the Borrower of such Loan and the Banks.

"Interest Period": with respect to any Eurodollar Loan:

(i) initially, the period commencing on the borrowing or conversion date, as the case may be, with respect to such Eurodollar Loan and ending one or two weeks, or one, two, three or six months thereafter, or such other period as may be requested by the Borrower and agreed to by the Banks making such Loan, as selected by the Borrower of such Loan in its notice of borrowing or notice of conversion, as the case may be, given with respect thereto; and

(ii) thereafter, each period commencing on the last day of the next preceding Interest Period applicable to such Eurodollar Loan and ending one or two weeks, or one, two, three or six months thereafter, or such other period as may be requested by the Borrower and agreed to by the Banks making such Loan, as selected by such Borrower by irrevocable notice to the Administrative Agent and the Banks which made such Eurodollar Loan not less than two Business Days prior

to the last day of the then current Interest Period with respect thereto;

provided that, all of the foregoing provisions relating to Interest Periods are subject to the following:

- (1) if any Interest Period pertaining to a Eurodollar Loan would otherwise end on a day that is not a Business Day, such Interest Period shall be extended to the next succeeding Business Day unless the result of such extension would be to carry such Interest Period into another calendar month in which event such Interest Period shall end on the immediately preceding Business Day;
- (3) any Interest Period pertaining to a Eurodollar Loan that begins on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall end on the last Business Day of a calendar month.

"Lien": any mortgage, pledge, hypothecation, assignment as security, security deposit arrangement, encumbrance, lien (statutory or other), conditional sale or other title retention agreement or other similar arrangement.

"Loan": any loan made by any Bank pursuant to this $\ensuremath{\mathsf{Agreement}}.$

"Loan Documents": this Agreement, the Notes and the Notices of Additional Borrower. $\,$

"Loan Outstandings": as to any Bank at any time, the sum of (a) the aggregate principal amount of all Revolving Credit Loans made by such Bank then outstanding, and (b) such Bank's Commitment Percentage multiplied by the aggregate principal amount of all Bid Loans then outstanding.

"Loan Parties": the collective reference to the Company, the other Borrowers and the Parent. $\,$

"Majority Banks": at any time, Banks the Commitment Percentages of which aggregate (or, if at such time all of the Commitments shall have been terminated, Banks the Commitment Percentages of which immediately prior to such termination aggregated) at least 51%.

"Material Adverse Effect": a material adverse effect on (a) the business, operations, properties, or condition (financial or otherwise) of the Parent and its Subsidiaries taken as a whole, (b) the ability of the Company, or any Borrower or any other Loan Party to perform their respective obligations hereunder and under the other Loan Documents to which such Person is a party, or (c) the validity or enforceability of the Loan Documents or the rights or remedies of the Administrative Agent or the Banks hereunder or thereunder.

"Moody's": Moody's Investors Services, Inc.

"Multiemployer Plan": a Plan which is a multiemployer plan as defined in Section 4001(a)(3) of ERISA.

"Net Sales and Revenues": with respect to any Person for any period, all sales and operating revenues of such Person during such period computed in accordance with GAAP after deducting therefrom sales returns, discounts and allowances.

"Notes": the collective reference to the Revolving Credit Notes and the Bid Loan Notes, if any.

"Notice of Additional Borrower": as defined in subsection 13.15(a). $\label{eq:continuous} % \begin{subarray}{ll} \end{subarray} % \begin{subarr$

"Obligations": the unpaid principal of and interest on (including, without limitation, interest accruing after the maturity of the Loans and interest accruing after the filing of any petition in bankruptcy, or the commencement of any insolvency, reorganization or like proceeding, relating to any of the Loan Parties or any of the Borrowers, whether or not a claim for post-filing or post-petition interest is allowed in such proceeding) the Loans and the Notes, if any, and all other obligations and liabilities of any of the Loan Parties or the Borrowers to the Administrative Agent or to the Banks, whether direct or indirect, absolute or contingent, due or to become due, or now existing or hereafter incurred, which may arise under, out of, or in connection with, this Agreement, the Notes, any other Loan Document and any other document made, delivered or given in connection herewith or therewith, whether on account of principal, interest, reimbursement obligations, fees, indemnities, costs, expenses (including, without limitation, all fees and disbursements of counsel to the Administrative Agent or to the Banks that are required to be paid by the Loan Parties and/or the Borrowers pursuant to the terms of this Agreement) or any other obligation hereunder or thereunder.

"Parent Guarantee": as defined in subsection 12.1.

"Participant": as defined in subsection 13.6(b).

"Payment Sharing Notice": a written notice from the Company or any Bank informing the Administrative Agent that an Event of Default has occurred and is continuing and directing the Administrative Agent to allocate payments thereafter received from the Borrower in accordance with subsection 5.9(c).

"PBGC": the Pension Benefit Guaranty Corporation established pursuant to Subtitle A of Title IV of ERISA.

"Person": an individual, partnership, corporation, business trust, joint stock company, trust, unincorporated association, joint venture, Governmental Authority or other entity of whatever nature.

"Plan": at a particular time, any employee benefit plan which is covered by ERISA and in respect of which the Company or a Commonly Controlled Entity is (or, if such plan were terminated at such time, would under Section 4069 of ERISA be deemed to be) an "employer" as defined in Section 3(5) of ERISA.

"Prepayment Date": as defined in subsection 5.3(b).

"Principal Subsidiary": (a) any Borrower and (b) any other Subsidiary if it shall have Total Assets at the end of the most recent fiscal year for which financial statements are required to have been furnished pursuant to subsection 8.1(a) in excess of \$75,000,000 or have had during such year Net Sales and Revenues in excess of \$75,000,000.

"Purchasing Banks": as defined in subsection 13.6(c).

"Register": as defined in subsection 13.6(d).

"Regulation U": Regulation U of the Board.

"Regulation X": Regulation X of the Board.

"Reorganization": with respect to any Multiemployer Plan, the condition that such plan is in reorganization within the meaning of Section 4241 of ERISA.

"Reportable Event": any of the events set forth in Section 4043(c) of ERISA, other than those events as to which the thirty day notice requirement is waived under PBGC Reg. Section 4043.

"Requested Bank": as defined in subsection 3.1(a).

"Requirement of Law": as to any Person, the Certificate of Incorporation and By-Laws or other organizational or governing documents of such Person, and any law, treaty, rule or regulation or determination of an arbitrator or a court or other Governmental Authority, in

each case applicable to or binding upon such Person or any of its property or to which such Person or any of its property is subject.

"Responsible Officer": the chief executive officer, the president, the chief financial officer or the treasurer, assistant treasurer or controller of, respectively, the Parent and the Company.

"Restructuring Activities": all reductions in carrying value of assets or investments and provisions for the termination and/or relocation of operations and employees.

"Revolving Credit Loans": as defined in subsection 2.1(a).

"Revolving Credit Notes": as defined in subsection 2.2.

"SEC": the Securities and Exchange Commission, and any successor or analogous federal Governmental Authority.

"Single Employer Plan": any Plan which is covered by Title IV of ERISA, but which is not a Multiemployer Plan.

"S&P": Standard & Poor's Ratings Group.

"Subsidiary": as to any Person, a corporation, partnership or other entity which is required to be consolidated with such Person in accordance with GAAP; provided, that any such corporation, partnership or other entity which is controlled by a receiver or trustee under any bankruptcy, insolvency or similar law shall continue to be a "Subsidiary" of such Person for purposes of this Agreement. Unless otherwise qualified, all references to a "Subsidiary" or to "Subsidiaries" in this Agreement shall refer to a Subsidiary or Subsidiaries of the Parent.

"Substitute Rating Agency": as defined in the definition of "Applicable Margin".

"Termination Date": May 16, 2002.

"Total Assets": with respect to any Person at any time, the total of all assets appearing on the asset side of the balance sheet of such Person prepared in accordance with GAAP as of such time.

"Transferee": as defined in subsection 13.6(f).

"Type": as to any Loan, its nature as an ABR Loan or a Eurodollar Loan. $\,$

1.2 Other Definitional Provisions. (a) Unless otherwise specified therein, all terms defined in this Agreement $\,$

shall have the defined meanings when used in the Notes, if any, or any certificate or other document made or delivered pursuant hereto.

- (b) As used herein and in the Notes, if any, and any certificate or other document made or delivered pursuant hereto, accounting terms relating to the Parent and its Subsidiaries not defined in subsection 1.1, to the extent not defined, shall have the respective meanings given to them under GAAP.
- (c) The words "hereof", "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement, and Section, subsection, Schedule and Exhibit references are to this Agreement, unless otherwise specified.
- (d) The meanings given to terms defined herein shall be equally applicable to both the singular and plural forms of such terms.

SECTION 2. AMOUNT AND TERMS OF COMMITMENTS

- 2.1 Commitments. (a) Subject to the terms and conditions hereof, each Bank severally agrees to make revolving credit loans ("Revolving Credit Loans") to any Borrower from time to time during the Commitment Period in an aggregate principal amount at any one time outstanding not to exceed the amount of such Bank's Commitment, provided that no Bank shall make any Revolving Credit Loan if, after giving effect to such Loan, the aggregate Loan Outstandings of all of the Banks plus the Aggregate Outstanding Bilateral Option Loans would exceed the aggregate Commitments.
- (b) The Revolving Credit Loans may from time to time be (i) Eurodollar Loans, (ii) ABR Loans or (iii) a combination thereof, as determined by the Borrower thereof and notified to the Administrative Agent in accordance with subsections 2.3 and 5.5, provided that no Revolving Credit Loan shall be made as a Eurodollar Loan maturing after the Termination Date.
- 2.2 Obligations of Borrowers; Revolving Credit Notes. (a) Each Borrower agrees that each Revolving Credit Loan made by each Bank to such Borrower pursuant hereto shall constitute the promise and obligation of such Borrower to pay to the Administrative Agent, on behalf of such Bank, at the office of the Administrative Agent specified in subsection 13.2, in lawful money of the United States of America and in immediately available funds the aggregate unpaid principal amount of all Revolving Credit Loans made by such Bank to such Borrower pursuant to subsection 2.1, which amounts shall be due and payable (whether at maturity or by acceleration) as set forth in this Agreement and, in any event, on the Termination Date.

- (b) Each Borrower agrees that each Bank and the Administrative Agent are authorized to record (i) the date, amount and Type of each Revolving Credit Loan made by such Bank to such Borrower pursuant to subsection 2.1, (ii) the date of each continuation thereof pursuant to subsection 5.5(b), (iii) the date of each conversion of all or a portion thereof to another Type pursuant to subsection 5.5(a), (iv) the date and amount of each payment or prepayment of principal of each such Revolving Credit Loan and (v) in the case of each such Revolving Credit Loan which is a Eurodollar Loan, the length of each Interest Period and the Eurodollar Rate with respect thereto, in the books and records of such Bank or the Administrative Agent, as the case may be, and in such manner as is reasonable and customary for such Bank or the Administrative Agent, as the case may be, and a certificate of an officer of such Bank or the Administrative Agent, as the case may be, setting forth in reasonable detail the information so recorded, shall constitute prima facie evidence of the accuracy of the information so recorded; provided that the failure to make any such recording shall not in any way affect the obligations of such Borrower hereunder.
- (c) Each Borrower agrees that, upon the request to the Administrative Agent by any Bank at any time, the Revolving Credit Loans made by such Bank to such Borrower shall be evidenced by a promissory note of such Borrower, substantially in the form of Exhibit A with appropriate insertions as to Borrower, payee, date and principal amount (a "Revolving Credit Note"), payable to the order of such Bank and in a principal amount equal to the lesser of (a) the amount of the initial Commitment of such Bank and (b) the aggregate unpaid principal amount of all Revolving Credit Loans made by such Bank to such Borrower. Upon the request to the Administrative Agent by any such Bank at any time, such Borrower shall execute and deliver to such Bank a Revolving Credit Note conforming to the requirements hereof and executed by a duly authorized officer of such Borrower. Each Bank is hereby authorized to record the date, Type and amount of each Revolving Credit Loan made by such Bank to such Borrower, each continuation thereof, each conversion of all or a portion thereof to another Type, the date and amount of each payment or prepayment of principal thereof and, in the case of Eurodollar Loans, the length of each Interest Period and the Eurodollar Rate with respect thereto, on the schedule annexed to and constituting a part of its Revolving Credit Note and any such recordation shall constitute prima facie evidence of the accuracy of the information so recorded; provided that the failure to make any such recording shall not in any way affect the obligations of such Borrower hereunder or thereunder. Each Revolving Credit Note shall (x) be dated the Closing Date, (y) be stated to mature on the Termination Date and (z) provide for the payment of interest in accordance with subsection 5.1.
- 2.3 Procedure for Revolving Credit Borrowing. Any Borrower may borrow under the Commitments from all Banks during the Commitment Period on any Business Day, provided that such

Borrower shall give the Administrative Agent irrevocable notice (which notice must be received by the Administrative Agent (a) prior to 4:00 P.M., New York City time, three Business Days prior to the requested Borrowing Date, if all or any part of the requested Revolving Credit Loans are to be initially Eurodollar Loans, or (b) prior to 10:00 A.M., New York City time, on the requested Borrowing Date, otherwise), specifying (i) the amount to be borrowed, (ii) the requested Borrowing Date, (iii) whether the borrowing is to be of Eurodollar Loans, ABR Loans or a combination thereof and (iv) if the borrowing is to be entirely or partly of Eurodollar Loans, the respective amounts of each such Type of Loan and the respective lengths of the initial Interest Periods therefor. Each borrowing under the Commitments shall be in an amount equal to \$5,000,000 or a whole multiple of \$1,000,000 in excess thereof (or, in the case of ABR Loans, if the amount of the Available Commitments minus the Aggregate Outstanding Bilateral Option Loans is less than \$5,000,000, such lesser amount). Upon receipt of such notice from such Borrower, the Administrative Agent shall promptly notify each Bank thereof. Each Bank will make the amount of its pro rata share of each such borrowing available to the Borrower at the office of the Administrative Agent specified in subsection 13.2 prior to 12:00 noon, New York City time, on the Borrowing Date requested by such Borrower in funds immediately available to the Administrative Agent. Such borrowing will then be made available to such Borrower on the books of such office with the aggregate of the amounts made available to the Administrative Agent by the Banks and in like funds as received by the Administrative Agent.

SECTION 3. BILATERAL OPTION LOANS

3.1 Requests for Offers. (a) From time to time during the period from the Closing Date until the Termination Date, any Borrower may request any or all of the Banks (each such Bank to which such a request is made, a "Requested Bank") to make offers to make Bilateral Option Loans, provided that immediately after making any such Bilateral Option Loan, the aggregate Loan Outstandings of all the Banks plus the Aggregate Outstanding Bilateral Option Loans will not exceed the aggregate Commitments. Any such request shall specify the principal amount and maturity date of the Bilateral Option Loans for which such Borrower is requesting offers, whether such Bilateral Option Loans are requested to be Dollar Bilateral Loans or Alternative Currency Bilateral Loans, the time by which offers to make such Bilateral Option Loans must be made by such Requested Bank and by which such offers shall be accepted or rejected by such Borrower, and if all or any part of the requested Bilateral Option Loans are requested to be made as Alternative Currency Bilateral Loans, the Alternative Currency to be applicable thereto. Each Requested Bank may, but shall have no obligation to, make such offers on such terms and conditions as are satisfactory to such Requested Bank, and such Borrower may, but shall have no obligation to, accept any such offers. No Bilateral Option Loan may mature after the Termination

16

- (b) Each Borrower and Requested Bank shall separately agree as to the procedures, documentation, lending office and other matters relating to any Bilateral Option Loan.
- 3.2 Reports to Administrative Agent; Determination of Dollar Equivalents. (a) The Borrower shall deliver to the Administrative Agent a report in respect of each Bilateral Option Loan (a "Bilateral Option Loan Report") by 2:00 P.M. (New York City time) on the date on which the applicable Borrower accepts any Bilateral Option Loan, on the date on which any principal amount thereof is repaid prior to the scheduled maturity date, or on the scheduled maturity date if payment thereof is not made on such scheduled maturity date, specifying for such Bilateral Option Loan the date on which such Bilateral Option Loan was or will be made, such amount of principal is or will be repaid or such payment was not made as the case may be; in the case of Alternative Currency Bilateral Loans, the Alternative Currency thereof; and the principal amount of such Bilateral Option Loan or principal prepayment or repayment or the amount paid (in the case of any Alternative Currency Bilateral Loan, expressed in the Alternative Currency therefor).
- (b) Upon receipt of a Bilateral Option Loan Report with respect to the acceptance of a Bilateral Option Loan, the Administrative Agent shall determine the Dollar Equivalent thereof.
- (c) If on any Borrowing Date on which after giving effect to the Loans made on such date, the sum of the aggregate Loan Outstandings of all the Banks plus the Aggregate Outstanding Bilateral Option Loans exceeds 85% of the aggregate Commitments, then the Administrative Agent shall redetermine as of such Borrowing Date, on the basis of the most recently delivered Bilateral Option Loan Report for each Bilateral Option Loan, the Dollar Equivalent of each Alternative Currency Bilateral Loan then outstanding. In addition, for so long as the condition specified in the preceding sentence remains in effect, the Administrative Agent shall determine, at the end of each fiscal quarter of the Company, on the basis of the most recently delivered Bilateral Option Loan Report for each Bilateral Option Loan, the Dollar Equivalent of each Alternative Currency Bilateral Loan then outstanding.
- (d) The Administrative Agent shall promptly notify the Company of each Dollar Equivalent under this subsection 3.2.
- 3.3 Judgment Currency. If for the purpose of obtaining judgment in any court, it is necessary to convert a sum due from any Borrower hereunder or under any of the Notes in the currency expressed to be payable herein or under the Notes (the "specified currency") into another currency, the parties hereto agree, to the fullest extent that they may effectively do so, that the rate of exchange used shall be that at which in accordance with normal banking procedures the Administrative Agent could purchase the specified currency with such other

currency at the Administrative Agent's New York office on the Business Day preceding that on which final judgment is given. The obligations of each Borrower in respect of any sum due to any Bank or the Administrative Agent hereunder or under any Note shall, notwithstanding any judgment in a currency other than the specified currency, be discharged only to the extent that on the Business Day following receipt by such Bank or the Administrative Agent (as the case may be) of any sum adjudged to be so due in such other currency such Bank or the Administrative Agent (as the case may be) may in accordance with normal banking procedures purchase the specified currency with such other currency; if the amount of the specified currency so purchased is less than the sum originally due to such Bank or the Administrative Agent, as the case may be, in the specified currency, each Borrower agrees, to the fullest extent that it may effectively do so, as a separate obligation and notwithstanding any such judgment, to indemnify such Bank or the Administrative Agent, as the case may be, against such difference, and if the amount of the specified currency so purchased exceeds:

- (a) the sum originally due to any Bank or the Administrative Agent, as the case may be, and $% \left(1\right) =\left\{ 1\right\} =\left\{ 1\right\}$
- (b) any amounts shared with other Banks as a result of allocations of such excess as a disproportionate payment to such Bank under subsection 13.7,

such Bank or the Administrative Agent, as the case may be, agrees to remit such excess to the applicable Borrower.

3.4 Repayments. Each Borrower shall repay to each Bank which has made a Bilateral Option Loan on the maturity date of each Bilateral Option Loan (such maturity date being that specified in the documentation referred to in subsection 3.1(a)) the then unpaid principal amount of such Bilateral Option Loan.

SECTION 4. BID LOANS

- 4.1 The Bid Loans. Any Borrower may borrow Bid Loans from time to time on any Business Day during the period from the Closing Date until the Termination Date, in the manner set forth in this Section 4 and in amounts such that the aggregate Loan Outstandings of all the Banks at any time plus the Aggregate Outstanding Bilateral Option Loans at such time will not exceed the aggregate Commitments at such time, and provided, further, that no such Bid Loan shall be made if, after giving effect thereto, any Bid Loans would mature after the Termination Date.
- 4.2 Procedure for Bid Loans. (a) A Borrower shall request Bid Loans by delivering a Bid Loan Request to the Administrative Agent, in writing, by facsimile transmission, or by telephone, confirmed by facsimile transmission, not later than 1:00 P.M. (New York City time) one Business Day prior to the proposed Borrowing Date. Each Bid Loan Request may solicit bids

for Bid Loans in an aggregate principal amount of \$5,000,000 or a whole multiple of \$1,000,000 in excess thereof and for not more than three alternative maturity dates for such Bid Loans. The Administrative Agent shall promptly notify each Bank by facsimile transmission of the contents of each Bid Loan Request received by it.

- (b) Upon receipt of notice from the Administrative Agent of the contents of a Bid Loan Request, any Bank that elects, in its sole discretion, to do so, shall irrevocably offer to make one or more Bid Loans at a rate of interest determined by such Bank in its sole discretion for each such Bid Loan. Any such irrevocable offer shall be made by delivering a Bid Loan Offer to the Administrative Agent, by telephone, immediately confirmed by facsimile transmission, before 9:30 A.M. (New York City time) on the proposed Borrowing Date, setting forth the maximum amount of Bid Loans for each maturity date, and the aggregate maximum amount for all maturity dates, which such Bank would be willing to make (which amounts may, subject to subsection 4.1, exceed such Bank's Commitments) and the rate of interest at which such Bank is willing to make each such Bid Loan; the Administrative Agent shall advise the Borrower before 10:00 A.M. (New York City time) on the proposed Borrowing Date of the contents of each such Bid Loan Offer received by it. If the Administrative Agent in its capacity as a Bank shall, in its sole discretion, elect to make any such offer, it shall advise the Borrower of the contents of its Bid Loan Offer before 9:15 A.M. (New York City time) on the proposed Borrowing Date.
- (c) The Borrower shall before 10:30 A.M. (New York City time) on the proposed Borrowing Date, in its absolute discretion, either:
 - (i) cancel such Bid Loan Request by giving the Administrative Agent telephone notice to that effect, and the Administrative Agent shall give prompt telephone notice thereof to the Banks and the Bid Loans requested thereby shall not be made; or
 - (ii) accept one or more of the offers made by any Bank or Banks by giving telephone notice to the Administrative Agent (confirmed as soon as practicable thereafter by delivery to the Administrative Agent of a Bid Loan Confirmation in writing or by facsimile transmission) of the amount of Bid Loans for each relevant maturity date to be made by each Bank (which amount for each such maturity date shall be equal to or less than the maximum amount for such maturity date specified in the Bid Loan Offer of such Bid Loan Bank, and for all maturity dates included in such Bid Loan Offer shall be equal to or less than the aggregate maximum amount specified in such Bid Loan Offer for all such maturity dates) and reject

any remaining offers made by Banks; provided, however, that (x) the Borrower may not accept offers for Bid Loans for any maturity date in an aggregate principal amount in excess of the maximum principal amount requested in the related Bid Loan Request, (y) if the Borrower accepts any of such offers, it must accept offers strictly based upon pricing for such relevant maturity date and no other criteria whatsoever and (z) if two or more Banks submit offers for any maturity date at identical pricing and the Borrower accepts any of such offers but does not wish to (or by reason of the limitations set forth in subsection 4.1 or in clause (x) of this proviso, cannot) borrow the total amount offered by such Banks with such identical pricing, the Borrower shall accept offers from all of such Banks in amounts allocated among them pro rata according to the amounts offered by such Banks.

(d) If the Borrower accepts pursuant to clause (c) (ii) above one or more of the offers made by any Bid Loan Bank or Bid Loan Banks, the Administrative Agent shall notify before 11:00 A.M. (New York City time) each Bid Loan Bank which has made such an offer, of the aggregate amount of such Bid Loans to be made on such Borrowing Date for each maturity date and of the acceptance or rejection of any offers to make such Bid Loans made by such Bid Loan Bank. Each Bid Loan Bank which is to make a Bid Loan shall, before 12:00 Noon (New York City time) on the Borrowing Date specified in the Bid Loan Request applicable thereto, make available to the Administrative Agent at its office set forth in subsection 13.2 the amount of Bid Loans to be made by such Bid Loan Bank, in immediately available funds. The Administrative Agent will make such funds available to the Borrower at or before 2:00 P.M. (New York City time) on such date at the Administrative Agent's aforesaid address. As soon as practicable after each Borrowing Date, the Administrative Agent shall notify each Bank of the aggregate amount of Bid Loans advanced on such Borrowing Date and the respective maturity dates thereof.

4.3 Repayments. Each Borrower shall repay to the Administrative Agent for the account of each Bid Loan Bank which has made a Bid Loan on the maturity date of each Bid Loan (such maturity date being that specified by the Borrower for repayment of such Bid Loan in the related Bid Loan Request) the then unpaid principal amount of such Bid Loan. The Borrowers shall not have the right to prepay any principal amount of any Bid Loan without the prior written consent of the Bid Loan Bank which made such Bid Loan

4.4 Interest on Bid Loans. Each Borrower which shall have borrowed a Bid Loan shall pay interest on the unpaid principal amount of such Bid Loan from the Borrowing Date to the stated maturity date thereof, at the rate of interest determined

20

pursuant to subsection 4.2 above (calculated on the basis of a 360 day year for actual days elapsed), payable on the interest payment date or dates specified by such Borrower for such Bid Loan in the related Bid Loan Request. If all or a portion of the principal amount of any Bid Loan shall not be paid when due (whether at the stated maturity, by acceleration or otherwise), such overdue principal amount shall, without limiting any rights of any Bank under this Agreement, bear interest from the date on which such payment was due at a rate per annum which is 2% above the rate which would otherwise be applicable to such Bid Loan until the scheduled maturity date with respect thereto, and for each day thereafter at a rate per annum which is 2% above the Alternate Base Rate until paid in full (as well after as before judgment).

4.5 Obligations of Borrowers; Bid Loan Notes. (a) Each Borrower agrees that each Bid Loan made by each Bid Loan Bank to such Borrower pursuant hereto shall constitute the promise and obligation of such Borrower to pay to the Administrative Agent, on behalf of such Bid Loan Bank, at the office of the Administrative Agent specified in subsection 13.2, in lawful money of the United States of America and in immediately available funds the aggregate unpaid principal amount of each Bid Loan made by such Bid Loan Bank to such Borrower pursuant to subsection 4.2, which amounts shall be due and payable (whether at maturity or by acceleration) as set forth in the Bid Loan Request related to such Bid Loan and in this Agreement.

(b) Each Borrower agrees that each Bid Loan Bank and the Administrative Agent are authorized to record (i) the date and amount of each Bid Loan made by such Bid Loan Bank to such Borrower pursuant to subsection 4.2, and (ii) the date and amount of each payment or prepayment of principal of each such Bid Loan, in the books and records of such Bid Loan Bank or the Administrative Agent, as the case may be, and in such manner as is reasonable and customary for such Bank or the Administrative Agent, as the case may be, and a certificate of an officer of such Bid Loan Bank or the Administrative Agent, as the case may be, setting forth in reasonable detail the information so recorded, shall constitute prima facie evidence of the accuracy of the information so recorded; provided that the failure to make any such recording shall not in any way affect the obligations of such Borrower hereunder.

(c) Each Borrower agrees that, upon the request to the Administrative Agent by any Bid Loan Bank at any time, the Bid Loans made by such Bid Loan Bank to any Borrower shall be evidenced by a promissory note of such Borrower, substantially in the form of Exhibit B with appropriate insertions (a "Bid Loan Note"), payable to the order of such Bid Loan Bank and representing the obligation of such Borrower to pay the unpaid principal amount of all Bid Loans made by such Bid Loan Bank, with interest on the unpaid principal amount from time to time outstanding of each Bid Loan evidenced thereby as prescribed in

subsection 4.4. Upon the request to the Administrative Agent by any such Bid Loan Bank at any time, such Borrower shall execute and deliver to such Bid Loan Bank a Bid Loan Note conforming to the requirements hereof and executed by a duly authorized officer of such Borrower. Each Bid Loan Bank is hereby authorized to record the date and amount of each Bid Loan made by such Bank, the maturity date thereof, the date and amount of each payment of principal thereof and the interest rate with respect thereto on the schedule annexed to and constituting part of its Bid Loan Note, and any such recordation shall constitute prima facie evidence of the accuracy of the information so recorded; provided, however, that the failure to make any such recordation shall not affect the obligations of such Borrower hereunder or under any Bid Loan Note. Each Bid Loan Note shall be dated the Closing Date and each Bid Loan evidenced thereby shall bear interest for the period from and including the Borrowing Date thereof on the unpaid principal amount thereof from time to time outstanding at the applicable rate per annum determined as provided in, and such interest shall be payable as specified in, subsection 4.4.

SECTION 5. LOAN FACILITY COMMON PROVISIONS

- 5.1 Interest Rates and Payment Dates. (a) Each Eurodollar Loan shall bear interest for each day during each Interest Period with respect thereto at a rate per annum equal to the Eurodollar Rate determined for such Interest Period plus the Applicable Margin.
- (b) Each ABR Loan shall bear interest at a fluctuating rate per annum equal to the Alternate Base Rate.
- (c) Except as otherwise provided in subsection 4.4, if all or a portion of (i) the principal amount of any Loan or (ii) any interest payable thereon shall not be paid when due (whether at the stated maturity, by acceleration or otherwise), such overdue amount shall bear interest at a rate per annum which is (x) in the case of overdue principal, the rate that would otherwise be applicable thereto pursuant to the foregoing provisions of this subsection plus 2% or (y) in the case of overdue interest, the rate described in paragraph (b) of this subsection plus 2%, in each case from the date of such non-payment until such amount is paid in full (as well after as before judgment).
- (d) Interest shall be payable in arrears on each Interest Payment Date, provided that interest accruing pursuant to paragraph (c) of this subsection shall be payable on demand.
- (e) Subject to the limitations set forth herein, each Borrower may use the Loans by borrowing, prepaying and reborrowing the Loans, all in accordance with the terms and conditions hereof.

5.2 Facility Fee. (a) The Company agrees to pay to the Administrative Agent for the account of each Bank a facility fee for the period from and including the date hereof to the Termination Date, computed at the rate per annum determined as set forth in paragraph (b) of this subsection on the average daily amount of the Commitment of such Bank during the period for which payment is made, payable quarterly in arrears on the fifteenth day of each March, June, September and December and on the Termination Date or such earlier date as the Commitments shall terminate as provided herein, commencing on the first of such dates to occur after the date hereof.

(b) The rate per annum at which such facility fee under paragraph (a) above shall be computed (the "Applicable Facility Fee Rate"), for any day on which the long term senior unenhanced, unsecured debt of the Company is rated by both S&P and Moody's, shall be the rate per annum under the caption "Facility Fee Rate" (a "Facility Fee Rate") set forth below opposite the S&P and Moody's ratings applicable to such debt on such day (or, if such ratings are set opposite two different rates under said caption, then the Applicable Facility Fee Rate shall be the lower of said two Facility Fee Rates):

FACILITY FEE RATE	S&P	MOODY'S
. 2000%	BB+ or lower	Ba1 or lower
.1500%	BBB-	Baa3
.1100%	BBB	Baa2
.0900%	BBB+	Baa1
.0750%	A- or higher	A3 or higher

provided that if on any day the long term senior unenhanced, unsecured debt of the Company is rated by only one of S&P or Moody's, such rate will be determined based on the rating by such rating agency, and provided, further, that if on any day the long term senior unenhanced, unsecured debt of the Company is rated by neither S&P nor Moody's, the Applicable Facility Fee Rate will be determined based on the rating of such debt by a Substitute Rating Agency and will be the Facility Fee Rate set forth above opposite the S&P and Moody's ratings comparable to the Substitute Rating Agency's rating of such debt on such date, and provided, further, that if on any day the long term senior unenhanced, unsecured debt of the Company is rated by none of S&P, Moody's or any Substitute Rating Agency, the Company, the Administrative Agent and the Banks will negotiate in good faith to determine an alternative basis for calculating such rate consistent with the table set forth above and, if agreement on such alternative basis is not reached with 30 days, such rate will be calculated on an alternative basis determined by the Administrative Agent and the Banks in their reasonable discretion consistent with the table

above, and until such alternative basis is determined such rate will be the rate last determined as provided in the table above.

- 5.3 Termination or Reduction of Commitments; Change of Control Date. (a) The Company shall have the right, upon not less than five Business Days' notice to the Administrative Agent, to terminate the Commitments or, from time to time, to reduce the amount of the Commitments, provided that no such termination or reduction shall be permitted to the extent that, after giving effect thereto and to any prepayments of Loans made on the effective date thereof, the sum of the aggregate Loan Outstandings of all the Banks, plus the Aggregate Outstanding Bilateral Option Loans would exceed the Commitments then in effect. Any such partial reduction shall be in an amount equal to \$5,000,000 or a whole multiple of \$1,000,000 in excess thereof and shall reduce permanently the Commitments then in effect.
- (b) (i) In the event that a Change of Control Date shall occur, (A) the Company shall, within 10 days after such Change of Control Date, give each Bank notice thereof in writing describing in reasonable detail the facts and circumstances giving rise thereto, and (B) such Bank, by written notice given to the Company not later than 30 days after the Change of Control Date, may declare the Commitments of such Bank to be terminated in full or reduced as of the date of (or as of a later date specified in) such notice to the Company, and may require that the Borrowers prepay as provided in this subsection 5.3 any Loans payable to such Bank and outstanding on such date to the extent the principal amount thereof exceeds such Bank's Commitment, if any, remaining after such termination or reduction. To the extent such Bank so requires, the Borrowers shall prepay such Loans on the 75th day after the date of the Company's notice or, in the event such 75th day is not a Business Day, the Business Day next succeeding such 75th day ("Prepayment Date").
- (ii) On the Prepayment Date, the Borrowers shall prepay the unpaid principal amount of the Loans payable to such Bank, without premium or penalty, together with accrued interest on the amount prepaid to the Prepayment Date.
- (iii) Subsections 5.9(a), (b) and (c) shall not apply to prepayments under this subsection 5.3(b).
- (iv) Paragraph (a) of this subsection 5.3 hereof shall not apply to any Commitment reductions pursuant to this paragraph (b).
- (v) In the event that a Change of Control Date shall occur, the Company shall not thereafter, without the prior written consent of the Majority Banks, borrow any additional Loan (other than a Bilateral Option Loan) in order to make, directly or indirectly, any payment or prepayment on any Indebtedness subordinated as to the payment of principal and interest or on liquidation to the prior payment of any of the Obligations.

5.4 Prepayments. (a) Any Borrower may at any time and from time to time upon at least four Business Days' irrevocable notice to the Administrative Agent, in the case of Eurodollar Loans, or upon at least one Business Day's irrevocable notice to the Administrative Agent, in the case of ABR Loans, prepay the Loans (other than Bid Loans), in whole or in part, without premium or penalty (subject to subsection 5.13), specifying the date and amount of prepayment and whether the prepayment is of Eurodollar Loans, ABR Loans or a combination thereof, and, if of a combination thereof, the amount allocable to each. Upon receipt of any such notice the Administrative Agent shall promptly notify each Bank thereof. If any such notice is given, the amount specified in such notice shall be due and payable on the date specified therein. Partial prepayments shall be in an aggregate principal amount of \$5,000,000 or a whole multiple of \$1,000,000 in excess thereof.

(b) If at any time, the Administrative Agent shall determine (which determination shall be conclusive in the absence of manifest error) that the sum of the aggregate Loan Outstandings of all the Banks plus the Aggregate Outstanding Bilateral Option Loans exceeds the aggregate Commitments, the Borrowers shall immediately prepay the Loans in an aggregate principal amount equal to such excess.

5.5 Conversion and Continuation Options. (a) Any Borrower may elect at any time and from time to time (subject to subsection 5.13) to convert its Eurodollar Loans to ABR Loans by giving the Administrative Agent at least two Business Days' prior irrevocable notice of such election. Any Borrower may elect at any time and from time to time to convert its ABR Loans to Eurodollar Loans by giving the Administrative Agent irrevocable notice of such election (which notice must be received by the Administrative Agent prior to 4:00 P.M., New York City time, three Business Days prior to the requested conversion date). Any such notice of conversion to Eurodollar Loans shall specify the length of the initial Interest Period or Interest Periods therefor. Upon receipt of any such notice the Administrative Agent shall promptly notify each Bank thereof. All or any part of outstanding Eurodollar Loans and ABR Loans may be converted as provided herein, provided that (i) no Loan may be converted into a Eurodollar Loan when any Event of Default has occurred and is continuing and the Administrative Agent or the Majority Banks have determined that such a conversion is not appropriate, and (ii) any such conversion may only be made if, after giving effect thereto, subsection 5.6 shall not have been contravened.

(b) Any Eurodollar Loans may be continued as such upon the expiration of the then current Interest Period with respect thereto by the Borrower thereof giving notice to the Administrative Agent, in accordance with the applicable provisions of the term "Interest Period" set forth in subsection 1.1, of the length of the next Interest Period to be applicable to such Loans, provided that no Eurodollar Loan may be continued as such (i) when any Event of Default has occurred and is

continuing and the Administrative Agent or the Majority Banks have determined that such a continuation is not appropriate, or (ii) if, after giving effect thereto, subsection 5.6 would be contravened and provided, further, that if any Borrower shall fail to give any required notice as described above in this paragraph or if such continuation is not permitted pursuant to the preceding proviso such Loans shall be automatically converted to ABR Loans on the last day of such then expiring Interest Period.

- 5.6 Minimum Amounts of Eurodollar Tranches. All borrowings, conversions and continuations of Loans hereunder and all selections of Interest Periods hereunder shall be in such amounts and be made pursuant to such elections so that, after giving effect thereto, the aggregate principal amount of the Loans comprising each Eurodollar Tranche shall be equal to \$5,000,000 or a whole multiple of \$1,000,000 in excess thereof.
- 5.7 Computation of Interest and Fees. (a) Interest on ABR Loans and facility fees shall be calculated on the basis of a 365- (or 366-, as the case may be) day year for the actual days elapsed. Interest on Eurodollar Loans shall be calculated on the basis of a 360-day year for the actual days elapsed. The Administrative Agent shall as soon as practicable notify the Borrowers and the Banks of each determination of a Eurodollar Rate. Any change in the interest rate on a Loan resulting from a change in the Prime Rate shall become effective as of the opening of business on the day on which such change in the Prime Rate is announced. The Administrative Agent shall as soon as practicable notify the Borrowers and the Banks of the effective date and the amount of each such change in interest rate.
- (b) Each determination of an interest rate by the Administrative Agent pursuant to any provision of this Agreement shall be conclusive and binding on the Borrowers and the Banks in the absence of manifest error. The Administrative Agent shall, at the request of the Company, deliver to the Company a statement showing in reasonable detail the quotations and calculations used by the Administrative Agent in determining any interest rate pursuant to subsections 5.1 and 5.7(a).
- $\,$ 5.8 Inability to Determine Interest Rate. In the event that prior to the first day of any Interest Period:
 - (a) the Administrative Agent shall have determined (which determination shall be conclusive and binding upon the Borrowers) that, by reason of circumstances affecting the relevant market, adequate and reasonable means do not exist for ascertaining the Eurodollar Rate for such Interest Period, or
 - (b) the Administrative Agent shall have received notice from the Majority Banks that the Eurodollar Rate determined or to be determined for such Interest Period will not adequately and fairly reflect the cost to such Banks (as

conclusively certified by such Banks) of making or maintaining their affected Loans during such Interest Period.

the Administrative Agent shall give telecopy or telephonic notice thereof to the Borrowers and the Banks as soon as practicable thereafter. If such notice is given (x) any Eurodollar Loans requested to be made on the first day of such Interest Period shall be made as ABR Loans, (y) any Loans that were to have been converted on the first day of such Interest Period to Eurodollar Loans shall be continued as ABR Loans and (z) any outstanding Eurodollar Loans shall be converted, on the first day of such Interest Period, to ABR Loans. Until such notice has been withdrawn by the Administrative Agent, no further Eurodollar Loans shall be made or continued as such, nor shall any Borrower have the right to convert Loans to Eurodollar Loans.

5.9 Pro Rata Treatment and Payments. (a) Each borrowing by any Borrower of Revolving Credit Loans from the Banks hereunder, each payment by the Company on account of any facility fee hereunder, and any reduction of the Commitments of the Banks shall be made pro rata according to the respective Commitment Percentages of the Banks.

(b) Whenever any payment received by the Administrative Agent or any Bank under this Agreement or any Note is insufficient to pay in full all amounts then due and payable to the Administrative Agent and the Banks under this Agreement and the Notes, and the Administrative Agent has not received a Payment Sharing Notice (or if the Administrative Agent has received a Payment Sharing Notice but the Event of Default specified in such Payment Sharing Notice has been cured or waived), such payment shall be distributed and applied by the Administrative Agent and the Banks in the following order: first, to the payment of fees and expenses due and payable to the Administrative Agent in its capacity as Administrative Agent under and in connection with this Agreement; second, to the payment of all expenses due and payable under subsection 13.5, ratably among the Banks in accordance with the aggregate amount of such payments owed to each such Bank; third, to the payment of fees due and payable under subsections 5.2(a) and (b), ratably among the Banks in accordance with their Commitment Percentages; fourth, to the payment of interest then due and payable on the Loans, ratably among the Banks in accordance with the aggregate amount of interest owed to each such Bank; and fifth, to the payment of the principal amount of the Loans which is then due and payable, ratably among the Banks in accordance with the aggregate principal amount owed to each such Bank.

(c) After the Administrative Agent has received a Payment Sharing Notice which remains in effect, all payments received by the Administrative Agent under this Agreement or any Note shall be distributed and applied by the Administrative Agent and the Banks in the following order: first, to the payment of all amounts described in clauses first through third of the

foregoing paragraph (b), in the order set forth therein; and second, to the payment of the interest accrued on and the principal amount of all of the Loans, regardless of whether any such amount is then due and payable, ratably among the Banks in accordance with the aggregate accrued interest plus the aggregate principal amount owed to such Bank.

(d) All payments (including prepayments) to be made by any Borrower hereunder and under the Notes, whether on account of principal, interest, fees or otherwise, shall be made without set-off or counterclaim and shall be made prior to 3:00 P.M., New York City time, on the due date thereof (i) in the case of fees and Loans other than Bilateral Option Loans, to the Administrative Agent, for the account of the Banks, at the Administrative Agent's office specified in subsection 13.2, and (ii) in the case of Bilateral Option Loans made by any Bank, to such Bank, at the Bank's office specified in Schedule I (or, with respect to Alternative Currency Bilateral Loans, if different, at such other office of the Bank that it shall designate), in each case in Dollars (or, with respect to Alternative Currency Bilateral Loans, in the relevant Alternative Currency) and in immediately available funds. If any payment hereunder becomes due and payable on a day other than a Business Day, such payment shall be extended to the next succeeding Business Day (unless, with respect to any payment on a Eurodollar Loan, the result of such extension would be to extend such payment into another calendar month, in which event such payment shall be made on the immediately preceding Business Day), and, with respect to payments of principal, interest thereon shall be payable at the then applicable rate during such extension.

(e) Unless the Administrative Agent shall have been notified in writing by the Bank prior to a Borrowing Date that such Bank will not make the amount of any Loan it has committed to make on such date available to the Administrative Agent, the Administrative Agent may assume that such Bank has made such amount available to the Administrative Agent on such Borrowing Date, and the Administrative Agent may, in reliance upon such assumption, make available to the applicable Borrower a corresponding amount. If such amount is made available to the Administrative Agent on a date after such Borrowing Date, such Bank shall pay to the Administrative Agent on demand an amount equal to the product of (i) the daily average Federal Funds Effective Rate during such period, times (ii) the amount of the Loan such Bank was committed to make, times (iii) a fraction the numerator of which is the number of days that elapse from and including such Borrowing Date to the date on which such Bank's Loan shall have become immediately available to the Administrative Agent and the denominator of which is 360. A certificate of the Administrative Agent submitted to any Bank with respect to any amounts owing under this subsection shall be conclusive in the absence of manifest error. If such Bank's Commitment Percentage of such borrowing is not in fact made available to the Administrative Agent by such Bank within three Business Days of such Borrowing Date, the Administrative Agent

shall be entitled to recover such amount with interest thereon at the rate per annum applicable to ABR Loans hereunder, on demand, from such Borrower.

- 5.10 Illegality. Notwithstanding any other provision herein, if any change after the date hereof in any Requirement of Law or in the interpretation or application thereof shall make it unlawful for any Bank to make or maintain Eurodollar Loans as contemplated by this Agreement, (a) the commitment of such Bank hereunder to make Eurodollar Loans, continue Eurodollar Loans as such and convert ABR Loans to Eurodollar Loans shall forthwith be cancelled and (b) such Bank's Loans then outstanding as Eurodollar Loans, if any, shall be converted automatically to ABR Loans on the respective last days of the then current Interest Periods with respect to such Loans or within such earlier period as required by law. If any such conversion of a Eurodollar Loan occurs on a day which is not the last day of the then current Interest Period with respect thereto, the Borrower of such Loan shall pay to such Bank such amounts, if any, as may be required pursuant to subsection 5.13.
- 5.11 Requirements of Law. (a) In the event that any change after the date hereof in any Requirement of Law or in the interpretation or application thereof or compliance by any Bank with any request or directive (whether or not having the force of law) from any central bank or other Governmental Authority made subsequent to the date hereof:
 - (i) shall subject any Bank to any tax of any kind whatsoever with respect to this Agreement, any Note or any Eurodollar Loan made by it, or change the basis of taxation of payments to such Bank in respect thereof (except for taxes covered by subsection 5.12 and changes in taxes based upon or measured by income of such Bank);
 - (ii) shall impose, modify or hold applicable any reserve, special deposit, compulsory loan or similar requirement against assets held by, deposits or other liabilities in or for the account of, advances, loans or other extensions of credit by, or any other acquisition of funds by, any office of such Bank which is not otherwise included in the determination of the Eurodollar Rate hereunder; or
 - (iii) shall impose on such Bank any other condition;

and the result of any of the foregoing is to increase the cost to such Bank, by an amount which such Bank deems in its reasonable judgment to be material, of making, converting into, continuing or maintaining Eurodollar Loans or to reduce any amount receivable hereunder in respect thereof then, in any such case, the Company shall promptly pay such Bank, upon its demand, any additional amounts necessary to compensate such Bank for such increased cost or reduced amount receivable. If any Bank becomes entitled to claim any additional amounts pursuant to this

subsection, it shall promptly notify the Company, through the Administrative Agent, of the event by reason of which it has become so entitled. A certificate as to any additional amounts payable pursuant to this subsection setting forth the calculation thereof in reasonable detail (as determined by such Bank in its reasonable discretion) submitted by such Bank, through the Administrative Agent, to the Company shall be conclusive in the absence of manifest error. This covenant shall survive the termination of this Agreement and the payment of the Notes and all other amounts payable hereunder.

- (b) In the event that any Bank shall have determined that any change in any Requirement of Law regarding capital adequacy or in the interpretation or application thereof or compliance by such Bank or any corporation controlling such Bank with any request or directive regarding capital adequacy (whether or not having the force of law) from any Governmental Authority made subsequent to the date hereof does or shall have the effect of reducing the rate of return on such Bank's or such corporation's capital as a consequence of its obligations hereunder to a level below that which such Bank or such corporation could have achieved but for such change or compliance (taking into consideration such Bank's or such corporation's policies with respect to capital adequacy) by an amount deemed by such Bank to be material, then from time to time, after submission by such Bank to the Company (with a copy to the Administrative Agent) of a written request therefor, the Company shall pay to such Bank such additional amount or amounts as will compensate such Bank for such reduction. A certificate as to any additional amount payable pursuant to this subsection setting forth the calculation thereof in reasonable detail (as determined by such Bank in its reasonable discretion) through the Administrative Agent, to the Company shall be conclusive in the absence of manifest error.
- (c) Upon request by any Bank, through the Administrative Agent, from time to time, the Borrowers shall pay the cost of all Eurocurrency Reserve Requirements applicable to the Eurodollar Loans made by such Bank. If a Bank is or becomes entitled to receive payments in respect of Eurocurrency Reserve Requirements, pursuant to this subsection 5.11(c), it shall promptly notify the Borrowers thereof through the Administrative Agent. A certificate as to the amount of such Eurocurrency Reserve Requirements setting forth the calculation thereof in reasonable detail (as determined by such Bank in its reasonable discretion) submitted by such Bank, through the Administrative Agent, to the Borrowers shall be conclusive in the absence of manifest error. This covenant shall survive the termination of this Agreement and the payment of the Loans and all other amounts payable hereunder.

5.12 Taxes. (a) All payments made by any Borrower under this Agreement and any Notes shall be made free and clear of, and without deduction or withholding for or on account of, any present or future income, stamp or other taxes, levies,

imposts, duties, charges, fees, deductions or withholdings, now or hereafter imposed, levied, collected, withheld or assessed by any Governmental Authority, excluding, in the case of the Administrative Agent and each Bank, net income taxes and franchise taxes (imposed in lieu of net income taxes) that would not have been imposed on the Administrative Agent or such Bank, as the case may be, in the absence of a present or former connection between the jurisdiction of the government or taxing authority imposing such tax and the Administrative Agent or such Bank (other than a connection arising solely from the Administrative Agent or such Bank having executed, delivered or performed its obligations or received a payment under, or enforced, this Agreement or any Notes) or any political subdivision or taxing authority thereof or therein (all such non-excluded taxes, levies, imposts, duties, charges, fees, deductions and withholdings being hereinafter called "Taxes"). If any Taxes are required to be withheld from any amounts payable to the Administrative Agent or any Bank hereunder or under any Notes, the amounts so payable to the Administrative Agent or such Bank shall be increased to the extent necessary to yield to the Administrative Agent or such Bank (after payment of all Taxes) interest or any such other amounts payable hereunder at the rates or in the amounts specified in this Agreement and the Notes. Whenever any Taxes are payable by any Borrower in respect of any payment made hereunder, as promptly as possible thereafter any Borrower shall send to the Administrative Agent for its own account or for the account of such Bank, as the case may be, a certified copy of an original official receipt received by such Borrower showing payment thereof. If such Borrower fails to pay any Taxes when due to the appropriate taxing authority or fails to remit to the Administrative Agent the required receipts or other required documentary evidence, such Borrower shall indemnify the Administrative Agent and the Banks for any incremental taxes, interest or penalties that may become payable by the Administrative Agent or any Bank as a result of any such failure. The agreements in this subsection shall survive the termination of this Agreement and the payment of the Notes and all other amounts payable hereunder.

(b) Each Bank that is not incorporated under the laws of the United States of America or a state thereof agrees that it will deliver to the Company and the Administrative Agent (i) two duly completed copies of United States Internal Revenue Service Form 1001 or 4224 or successor applicable form, as the case may be, and (ii) an Internal Revenue Service Form W-8 or W-9 or successor applicable form. Each such Bank also agrees to deliver to the Company and the Administrative Agent two further copies of the said Form 1001 or 4224 and Form W-8 or W-9, or successor applicable forms or other manner of certification, as the case may be, on or before the date that any such form expires or becomes obsolete or after the occurrence of any event requiring a change in the most recent form previously delivered by it to the Company, and such extensions or renewals thereof as may reasonably be requested by the Company or the Administrative Agent, unless in any such case an event (including, without

31

limitation, any change in treaty, law or regulation) has occurred prior to the date on which any such delivery would otherwise be required which renders all such forms inapplicable or which would prevent such Bank from duly completing and delivering any such form with respect to it and such Bank so advises the Company and the Administrative Agent. Such Bank shall certify (i) in the case of a Form 1001 or 4224, that it is entitled to receive payments under this Agreement without deduction or withholding of any United States federal income taxes and (ii) in the case of a Form W-8 or W-9, that it is entitled to an exemption from United States backup withholding tax.

5.13 Indemnity. Each Borrower agrees to indemnify each Bank and to hold each Bank harmless from any loss or expense which such Bank may sustain or incur as a consequence of (a) default by any Borrower in payment when due of the principal amount of or interest on any Eurodollar Loan, (b) default by any Borrower in making a borrowing of, conversion into or continuation of Eurodollar Loans after such Borrower has given a notice requesting the same in accordance with the provisions of this Agreement, (c) default by any Borrower in making any prepayment after such Borrower has given a notice thereof in accordance with the provisions of this Agreement or (d) the making of a payment or prepayment of Eurodollar Loans on a day which is not the last day of an Interest Period with respect thereto, including, without limitation, in each case, any such loss or expense arising from the reemployment of funds obtained by it or from fees payable to terminate the deposits from which such funds were obtained. This covenant shall survive the termination of this Agreement and the payment of the Loans or Notes, if any, and all other amounts payable hereunder.

SECTION 6. REPRESENTATIONS AND WARRANTIES

To induce the Banks to enter into this Agreement, each of the Company and the Parent represents and warrants to the Administrative Agent and each Bank that:

6.1 Corporate Existence; Compliance with Law. Each Loan Party (a) is a corporation duly organized, validly existing and in good standing under the laws of its jurisdiction of incorporation, (b) is duly qualified and in good standing in each jurisdiction wherein, in the opinion of the Company and the Parent, the conduct of its business or the ownership of its properties requires such qualification and (c) is in compliance with all Requirements of Law, except to the extent that the failure to comply with paragraph (a), (b) or (c) of this subsection would not, in the aggregate, reasonably be expected to have a Material Adverse Effect.

6.2 Corporate Power, Authorization; Enforceable Obligations. Each Loan Party has the corporate power and authority to make, deliver and perform its obligations under the Loan Documents to which it is or will be a party, and has taken

all necessary corporate action to authorize (i) in the case of the Borrowers, the borrowings under this Agreement and any Notes to which it is or will be a party on the terms and conditions hereof and thereof and (ii) the execution, delivery and performance of this Agreement and the Loan Documents to which it is or will be a party. This Agreement has been, and any Note and the other Loan Documents to which it is or will be a party will be, duly executed and delivered on behalf of each relevant Loan Party. This Agreement constitutes, and each of the Notes, if any, and the other Loan Documents when executed and delivered will constitute, a legal, valid and binding obligation of the Loan Party thereto, enforceable against such Loan Party in accordance with its terms, such enforceability subject to limitations under any applicable bankruptcy, insolvency, moratorium or other laws affecting creditors' rights and by general equitable principles (whether applied in a proceeding in equity or at law). No consent of any other party (including stockholders of the Parent) and no consent, license, approval or authorization of, or registration or declaration with, any Governmental Authority is required to be obtained by any Loan Party in connection with the execution, delivery, performance, validity or enforceability of this Agreement and any Notes.

6.3 No Legal Bar. The execution, delivery and performance of this Agreement, the Notes and the other Loan Documents, the borrowings hereunder and the use of the proceeds thereof, will not violate or contravene any material provision of any Requirement of Law or material Contractual Obligation of the Parent, the Company or any of its Subsidiaries and will not result in, or require, the creation or imposition of any material Lien (other than Liens permitted under subsection 9.2) on any of its or their respective properties or revenues pursuant to any such Requirement of Law or Contractual Obligation.

6.4 No Material Litigation. There is no legal action, administrative proceeding or arbitration (whether or not purportedly on behalf of the Parent or the Company or any of its Subsidiaries) presently pending, or to the knowledge of the Parent or the Company threatened, against or affecting the Parent or the Company or any of its Subsidiaries which would reasonably be expected to have a Material Adverse Effect, except that the foregoing is subject to the fact that the Parent and the Company cannot predict at this time the results and impact, if any, of (i) those investigations and litigations referred to in Item 3 of the Parent's Annual Report on Form 10-K for the fiscal year ended December 31, 1996 referred to in subsection 6.6 to the extent that inability to make such a prediction or determine liability is stated in Item 3 and (ii) the governmental investigation and related matters described in Note 6 to the Consolidated Financial Statements referred to in Item 3.

6.5 Ownership of Properties. Each of the Parent, the Company and each of its Subsidiaries is the tenant under valid leases or has good title to substantially all its properties and assets, real and personal (except defects in title and other

33

matters that would not reasonably be expected to have a Material Adverse Effect), subject to no Lien except as permitted to exist under subsection 9.2.

- 6.6 Financial Condition. The consolidated balance sheet of the Parent and its Subsidiaries as at December 31, 1996 and the related consolidated statements of operations, shareholders' equity and of cash flows (together with the related notes), included or incorporated in the Parent's Annual Report on Form 10-K filed with the SEC for the fiscal year ended December 31, 1996, present fairly in all material respects the financial position of the Parent and its Subsidiaries as at such dates and the results of their operations and their cash flows for the fiscal years then ended. The unaudited consolidated balance sheet of the Parent and its Subsidiaries as at March 31, 1997 and the related unaudited consolidated statement of operations for the three-month interim period, and the related unaudited consolidated statement of cash flows for the three-month interim period, ended on such date, included in the Parent's Quarterly Report on Form 10-Q filed with the SEC for such period, present fairly in all material respects the financial position of the Parent and its Subsidiaries as at such date and the results of their operations and their cash flows for the three-month period then ended. All of such financial statements, including the notes to such financial statements, have been prepared in conformity with GAAP (subject, in the case of interim statements, to normal year-end adjustments and to the fact that such financial statements may be abbreviated and may omit footnotes or contain incomplete footnotes) consistently applied throughout the periods involved except as stated therein.
- 6.7 Disclosure of Contingent Liabilities. To the best of the knowledge and belief of the Parent, neither the Parent nor any of its Subsidiaries has any contingent obligation, liability for taxes, long-term leases, unusual forward or other liabilities, which are material in amount in relation to the consolidated financial condition of the Parent and its Subsidiaries taken as a whole and which are not disclosed in the financial statements (including the related notes) described in subsection 6.6 above.
- 6.8 ERISA. Each Plan that is intended to qualify under Section 401(a) of the Code satisfies in all material respects the applicable requirements for qualification under that Code Section. No Reportable Event has occurred and is continuing with respect to any such Plan, and neither the Parent nor any of its Subsidiaries has incurred any liability to the PBGC under Section 4062 of ERISA with respect to any such Plan that would reasonably be expected to have a Material Adverse Effect.
- 6.9 Certain Federal Regulations. Neither the Company nor any of its Subsidiaries is engaged in or will engage in the business of extending credit for the purposes of "purchasing" or "carrying" any "margin stock" within the respective meanings of each of the quoted terms under Regulation U of the Board, and no

part of the proceeds of any Loan will be used for any purpose which violates, or which would be inconsistent with, the provisions of Regulation U or X of the Board.

- 6.10 No Default. Neither the Parent nor any of its Subsidiaries is in default under or with respect to any of its Contractual Obligations in any respect which would reasonably be expected to have a Material Adverse Effect. No Default or Event of Default has occurred and is continuing.
- 6.11 Taxes. (a) Each of the Parent and its Subsidiaries has filed or caused to be filed all tax returns which, to the knowledge of the Parent, are required to be filed and has paid all taxes shown to be due and payable on said returns or on any assessments made against it or any of its property and all other taxes, fees or other charges imposed on it or any of its property by any Governmental Authority (other than any the amount or validity of which are currently being contested in good faith by appropriate proceedings and with respect to which adequate reserves to the extent required in conformity with GAAP, have been provided on the books of the Parent or its Subsidiaries, as the case may be) except insofar as the failure to make such filings or payments would not reasonably be expected to have a Material Adverse Effect; and (b) no tax Lien (other than a Lien permitted under subsection 9.2(a)) has been filed, and, to the knowledge of the Parent, no claim is being asserted, with respect to any such tax, fee or other charge which would reasonably be expected to have a Material Adverse Effect.
- 6.12 Investment Company Act; Other Regulations. None of the Parent, the Company or any of its Subsidiaries is an "investment company", or a company "controlled" by an "investment company", within the meaning of the Investment Company Act of 1940, as amended. None of the Parent, the Company or any other Borrower is subject to regulation under any Federal or State statute or regulation which limits its ability to incur Indebtedness.
- $\,$ 6.13 Purpose of Loans. The proceeds of the Loans shall be used by the Borrowers for general corporate purposes (which may include purchases by the Parent of its capital stock).
- 6.14 Environmental Matters. To the best of the knowledge of the Parent, the operations of the Parent and its Subsidiaries and all parcels of real estate owned or operated by the Parent or its Subsidiaries are in compliance with all Environmental Laws, except where the failure to so comply would not reasonably be expected to have a Material Adverse Effect.
- $\,$ 6.15 Principal Subsidiaries. Set forth on Schedule II are all of the Principal Subsidiaries as of the date hereof.

SECTION 7. CONDITIONS PRECEDENT

7.1 Conditions to Effectiveness. The parties hereto acknowledge that the effectiveness of this Agreement is subject to the satisfaction of the following conditions precedent:

- (a) Loan Documents. The Administrative Agent shall have received (i) this Agreement, executed and delivered by a duly authorized officer of each of the Loan Parties, with a counterpart for each Bank, (ii) for the account of each Bank so requesting, a Revolving Credit Note and a Bid Loan Note conforming to the requirements hereof and executed by a duly authorized officer of the Borrowers and (iii) an incumbency certificate of each of the Loan Parties which covers such officers.
- (b) Corporate Proceedings. The Administrative Agent shall have received, with a counterpart for each Bank, a copy of the resolutions, in form and substance satisfactory to the Administrative Agent, of the Board of Directors of each of the Loan Parties authorizing (i) the execution, delivery and performance of the Loan Documents to which it is or will be a party and (ii) the borrowings contemplated hereunder (in the case of each Borrower), certified by the Secretary or an Assistant Secretary of such Loan Party as of the Closing Date, which certificate shall state that the resolutions thereby certified have not been amended, modified, revoked or rescinded and shall be in form and substance satisfactory to the Administrative Agent.
- (c) Existing Credit Agreement. All commitments of the lenders under the Existing Credit Agreement shall have been terminated, all outstanding loans thereunder shall have been repaid in full, all unpaid fees thereunder shall have been paid in full and such agreement shall have been terminated.
- (d) Legal Opinions. The Administrative Agent shall have received, with a counterpart for each Bank, the following executed legal opinions:
 - (i) the executed legal opinion of counsel to the Company and the Parent who may be the General Counsel of the Company, substantially in the form of Exhibit F-1;
 - (ii) to the extent required pursuant to subsection 13.15(a)(ii), the executed legal opinion of counsel to any other Borrower, in form and substance reasonably satisfactory to the Administrative Agent; and
 - (iii) the executed legal opinion of Simpson Thacher & Bartlett, counsel to the Administrative Agent, substantially in the form of Exhibit F-2.

Each such legal opinion shall cover such other matters incident to the transactions contemplated by this Agreement as the Administrative Agent may reasonably require.

- (e) Officer's Certificate. The Administrative Agent shall have received, with a counterpart for each Bank, a certificate respecting accuracy of representations and warranties, the absence of events having a Material Adverse Effect and the absence of Defaults and Events of Default, substantially in the form of Exhibit G hereto, signed by a Responsible Officer on behalf of each of the Company and the Parent.
- (f) Additional Matters. All corporate and other proceedings, and all documents, instruments and other legal matters in connection with the transactions contemplated by this Agreement shall be satisfactory in form and substance to the Administrative Agent, and the Administrative Agent shall have received such other documents and legal opinions in respect of any aspect or consequence of the transactions contemplated hereby or thereby as it shall reasonably request.
- 7.2 Conditions to Each Loan. The agreement of each Bank to make any Loan requested to be made by it on any date is subject to the satisfaction of the following conditions precedent:
 - (a) Representations and Warranties. Each of the representations and warranties made by each of the Loan Parties in or pursuant to subsections 6.1, 6.2, 6.3, 6.5, 6.9, 6.10, 6.11, 6.12 and 6.13 of this Agreement and in or pursuant to any other Loan Document to which it is or will be a party, shall be true and correct in all material respects on and as of such date as if made on and as of such date, and the representation and warranty made pursuant to subsection 6.6 shall be true and correct in all material respects with respect to the financial statements most recently delivered pursuant to subsection 8.1, mutatis mutandis, as if such financial statements delivered pursuant to subsection 8.1 were the financial statements referred to in subsection 6.6.
 - (b) No Default. No Default or Event of Default shall have occurred and be continuing on such date or after giving effect to the Loans requested to be made on such date.

Each borrowing by the Borrowers hereunder shall constitute a representation and warranty by the Loan Parties as of the date of such Loan that the conditions contained in this subsection 7.2 have been satisfied.

SECTION 8. AFFIRMATIVE COVENANTS

Each of the Company and the Parent hereby agrees that, so long as the Commitments remain in effect, any Note remains outstanding and unpaid or any other amount is owing to any Bank or the Administrative Agent hereunder, each of the Company and the Parent shall and the Company (except in the case of delivery of financial information, reports and notices) shall cause each of its Principal Subsidiaries to:

8.1 Financial Statements. Furnish to each Bank:

- (a) as soon as available, but in any event within 120 days after the end of each fiscal year of the Parent, a copy of the consolidated balance sheet of the Parent and its Subsidiaries as at the end of such year and the related consolidated statements of operations, shareholders' equity and cash flows for such year (as included or incorporated by reference in the Parent's Annual Report on Form 10-K or successor form filed with the SEC for each such fiscal year), setting forth in each case in comparative form the figures for the previous year, reported on without a "going concern" or like qualification or exception, or qualification arising out of the scope of the audit, by Price Waterhouse or other independent certified public accountants of nationally recognized standing not unacceptable to the Majority Banks; and
- (b) as soon as available, but in any event not later than 75 days after the end of each of the first three quarterly periods of each fiscal year of the Parent, the unaudited consolidated balance sheet of the Parent and its Subsidiaries as at the end of such quarter and the related unaudited consolidated statements of operations for such quarter and the related unaudited consolidated statements of operations and cash flows for the portion of the fiscal year through the end of such quarter (as included in the Parent's Quarterly Report on Form 10-Q or successor form filed with the SEC for each such period), setting forth in each case in comparative form the figures for the previous year, certified by a Responsible Officer as being fairly stated in all material respects when considered in relation to the consolidated financial statements of the Parent and its Subsidiaries.

All such financial statements shall be prepared in conformity with GAAP (subject, in the case of interim statements, to normal year-end adjustments and to the fact that such financial statements may be abbreviated and may omit footnotes or contain incomplete footnotes) applied consistently throughout the periods reflected therein and with prior periods (except as disclosed therein).

8.2 Certificates; Other Information. Furnish to each Bank:

- (a) concurrently with the delivery of the financial statements referred to in subsection 8.1(a), a certificate of the independent certified public accountants reporting on such financial statements stating that in making the examination necessary therefor no knowledge was obtained of any Default or Event of Default, except as specified in such certificate;
- (b) concurrently with the delivery of the financial statements referred to in subsections 8.1(a) and 8.1(b), a certificate of a Responsible Officer of the Parent in his capacity as such officer stating that, to the best of such Officer's knowledge, each of the Borrowers and the Parent during such period has observed or performed all of its covenants and other agreements, and satisfied every condition, contained in this Agreement and in the Notes and the other Loan Documents to which it is a party to be observed, performed or satisfied by it, and that such Officer has obtained no knowledge of any Default or Event of Default except as specified in such certificate and showing in detail the calculation of compliance with subsections 9.1 and 9.2;
- (c) concurrently with the delivery of the financial statements referred to in subsection 8.1(a), a list of the Principal Subsidiaries as of the corresponding fiscal year end, certified by a Responsible Officer in his capacity as such officer;
- (d) within ten Business Days after the same are sent, copies of all financial statements and reports which the Parent sends to its shareholders generally relating to the business of the Parent and its Subsidiaries, and within ten Business Days after the same are filed, copies of all reports on Forms 10-K, 10-Q, 8-K, 8 and 10, and Schedules 13D, 13E-3, 13E-4, 13-G, 14D-1 and 14D-9, or successor forms or schedules, and the final prospectus in each effective registration statement (other than registration statements on Form S-8) and each post-effective amendment to such registration statement which the Parent may make to, or file with, the SEC; and
- (e) promptly, subject to reasonable confidentiality requirements agreed to by the Company and such Bank, such additional financial and other information as any Bank may from time to time reasonably request.
- 8.3 Payment of Obligations. Pay, discharge or otherwise satisfy at or before maturity or before they become delinquent, as the case may be, all its obligations of whatever nature, except where the amount or validity thereof is currently being contested in good faith by appropriate proceedings and reserves, to the extent required in conformity with GAAP with respect thereto, have been provided on the books of the Parent or its Subsidiaries, as the case may be, and except to the extent

that the failure to so pay, discharge or otherwise satisfy such obligations would not result in a Default or Event of Default under Section 10(e)(i).

- 8.4 Conduct of Business and Maintenance of Existence. Preserve, renew and keep in full force and effect its corporate existence and take all reasonable action to maintain all corporate rights, privileges and franchises necessary or desirable in the normal conduct of its business, except as otherwise permitted pursuant to subsection 9.3; comply with all Contractual Obligations and Requirements of Law except to the extent that failure to comply therewith would not, in the aggregate, reasonably be expected to have a Material Adverse Effect.
- 8.5 Insurance. Maintain with financially sound and reputable insurance companies (which may include, without limitation, captive insurers), such insurance coverage as is reasonable for the business activities of the Parent and its Subsidiaries; and furnish to the Administrative Agent, upon written request, such information as the Administrative Agent may reasonably request as to its insurance program.
- 8.6 Inspection of Property, Books and Records; Discussions. Permit representatives of any Bank (subject to reasonable safety and confidentiality requirements) to visit and inspect any of its properties and examine and make abstracts from any of its books and records at any reasonable time and as often as may reasonably be desired and to discuss the business, operations, properties and financial and other condition of the Parent and its Subsidiaries with officers and employees of the Parent and its Subsidiaries and, provided representatives of the Parent are given an opportunity to participate, with its independent certified public accountants.
- $\,$ 8.7 Notices. Promptly give notice to the Administrative Agent and each Bank of:
 - (a) the occurrence of any Default or Event of Default;
 - (b) any (i) default or event of default under any Contractual Obligation of the Parent or any of its Subsidiaries or (ii) litigation, investigation or proceeding which may exist at any time between the Parent or any of its Subsidiaries and any Governmental Authority, which in either case, would reasonably be expected to have a Material Adverse Effect;
 - (c) any litigation or proceeding affecting the Parent or any of its Subsidiaries in which the then reasonably anticipated exposure of the Parent and its Subsidiaries is \$10,000,000 or more and not covered by insurance, or in which injunctive or similar relief is sought which is then reasonably anticipated to have an adverse economic effect on the Parent and its Subsidiaries of \$10,000,000 or more;

- (d) the following events, as soon as possible and in any event within 30 days after the Company or the Parent knows or has reason to know thereof: (i) the occurrence or expected occurrence of any Reportable Event with respect to any Plan, or any withdrawal from, or the termination, Reorganization or Insolvency of any Multiemployer Plan or (ii) the institution of proceedings or the taking of any other action by the PBGC or the Company or the Parent or any Commonly Controlled Entity or any Multiemployer Plan with respect to the withdrawal from, or the terminating, Reorganization or Insolvency of, any Plan, where in connection with any of the events described in (i) or (ii) above the liability to the Company or a Commonly Controlled Entity would reasonably be expected to be \$10,000,000 or more;
- (e) any upgrading, downgrading or cessation in the rating of the long term senior unenhanced, unsecured debt of the Company by the rating agency or agencies whose rating on such debt is then being used to determine the Applicable Margin and the Facility Fee Rate; and
- (f) a development or event which would reasonably be expected to have a Material Adverse Effect.

Each notice pursuant to this subsection shall be accompanied by a statement of a Responsible Officer setting forth details of the occurrence referred to therein and stating what action each of the Company and the Parent proposes to take with respect thereto.

8.8 Environmental Laws.

- (a) Comply with all Environmental Laws and obtain and comply with and maintain any and all licenses, approvals, registrations or permits required by Environmental Laws, except to the extent that failure to do so would not be reasonably expected to have a Material Adverse Effect; and
- (b) Defend, indemnify and hold harmless the Administrative Agent and the Banks, and their respective employees, agents, officers and directors, from and against any claims, demands, penalties, fines, liabilities, settlements, damages, costs and expenses of whatever kind or nature known or unknown, contingent or otherwise, arising out of, or in any way relating to the violation of or noncompliance with any Environmental Laws applicable to the real property owned or operated by the Company, the Parent or any of the Company's Subsidiaries, or any orders, requirements or demands of Governmental Authorities related thereto, including, without limitation, attorney's and consultant's fees, investigation and laboratory fees, court costs and litigation expenses, except to the extent that any of the foregoing arise out of the gross negligence or willful misconduct of the party seeking indemnification therefor.

SECTION 9. NEGATIVE COVENANTS

The Parent hereby agrees that, so long as the Commitments remain in effect, any Note remains outstanding and unpaid or any other amount is owing to any Bank or the Administrative Agent hereunder, it shall not, and (except with respect to subsections 9.1 and 9.5(b)) shall not permit any of its Subsidiaries to, directly or indirectly:

9.1 Financial Condition Covenants.

- (a) Consolidated Debt to EBITDA. Permit on the last day of any fiscal quarter of the Company commencing with June 30, 1997, the ratio of Consolidated Debt as at the end of such fiscal quarter to EBITDA for the period of four consecutive fiscal quarters then ended to be greater than 3.5 to 1.0.
- (b) Interest Coverage. Permit for any period of four consecutive fiscal quarters ending on the last day of any fiscal quarter of the Company commencing with June 30, 1997, the ratio of EBIT for such four fiscal quarter period to Consolidated Interest Expense for such four fiscal quarter period to be less than 2.0 to 1.0.
- 9.2 Limitation on Liens. Create, incur, assume or suffer to exist any Lien upon any of its property, assets or revenues (which property, assets or revenues are or would be reflected from time to time on the consolidated financial statements of the Parent and its Subsidiaries in accordance with GAAP), whether now owned or hereafter acquired, except for:
 - (a) Liens for taxes or other governmental charges not yet due or which are being contested in good faith by appropriate proceedings, provided that adequate reserves with respect thereto are maintained on the books of the Parent or its Subsidiaries, as the case may be, to the extent required in conformity with GAAP;
 - (b) carriers', warehousemen's, mechanics', materialmen's, repairmen's, vendors', landlords', brokers', bankers' and other like Liens arising in the ordinary course of business relating to obligations which are not overdue for a period of more than 60 days or which are being contested in good faith and Liens arising out of judgments or awards that are either discharged within 60 days after entry or execution of which has been stayed pending the outcome of appeal or review proceedings;
 - (c) pledges or deposits in connection with workers' compensation, unemployment insurance and other social security legislation and deposits securing liability to insurance carriers under insurance or self-insurance arrangements;

- (d) pledges, deposits and similar arrangements in connection with or to secure performance of bids, tenders, leases and other deposits to secure the performance of bids, trade contracts (other than for borrowed money), leases, statutory obligations, surety and appeal bonds, performance bonds and other obligations of a like nature incurred in the ordinary course of business and contractual rights of other Persons to make set-offs and to require security in connection with letters of credit, currency, commodity and interest rate contracts, surety bonds, leases, banking and brokerage agreements and other transactions in the ordinary course of business;
- (e) leases, easements, rights-of-way, restrictions and other similar encumbrances incurred in the ordinary course of business which would not reasonably be expected to have a Material Adverse Effect;
- (f) Liens on the property, assets or revenues of a Person which becomes a Subsidiary after the date hereof, to the extent that (i) such Liens existed at the time such Person became a Subsidiary and were not created in anticipation thereof, (ii) any such Lien is not extended to cover any property, assets or revenues of such Person after the time such Person becomes a Subsidiary, and (iii) the amount of Indebtedness secured thereby is not thereafter increased;
- (g) Liens arising in connection with (i) industrial development, pollution control or other tax exempt financing transactions, provided that such Liens do not at any time encumber any property other than the property financed by such transaction and other property, assets or revenues related to the property so financed on which Liens are customarily granted in connection with such transactions, or (ii) conveyances of any production payment or other obligation to make a production payment (A) which is to be made solely from production from oil, gas or other underground mineral properties dedicated thereto or (B) as to which production payment amount the obligee's sole recourse is to such properties;
- (h) Liens (including, without limitation, Liens incurred in connection with Capitalized Leases, operating leases and sale-leaseback transactions) securing Indebtedness of the Parent and its Subsidiaries incurred to finance the acquisition of fixed or capital assets, and refinancings thereof, provided that (i) such Liens do not at any time encumber any property other than the property financed by such Indebtedness and other property, assets or revenues related to the property so financed on which Liens are customarily granted in connection with such financings or refinancings, and (ii) the principal amount of Indebtedness secured by any such Lien shall at no time exceed 100% of the greater of the original purchase price of

such property at the time it was acquired and the fair market value of such property as reasonably determined by a Responsible Officer of the Company in good faith thereafter, plus fees and other costs related to the financing or refinancing thereof which have been agreed upon in an arm's length manner;

- (i) Liens incurred in connection with accounts receivable sale transactions entered into by the Parent or its Subsidiaries;
- (j) Liens securing Contractual Obligations of any Subsidiary to the Parent, the Company or any Domestic Subsidiary;
- (k) Liens on the property, assets or revenues of any Foreign Subsidiary or any Excluded Subsidiary; and
- (1) Liens (not otherwise permitted hereunder) which secure obligations in an aggregate amount at any time outstanding not exceeding an amount equal to 5% of the amount recorded opposite the caption "Properties and equipment, net" (or the equivalent caption) on the consolidated balance sheet of the Parent and its Subsidiaries most recently delivered to the Administrative Agent pursuant to subsection 8.1.
- 9.3 Limitation on Fundamental Changes. Convey, sell, lease, assign, transfer or otherwise dispose of (including by merger, consolidation, sale of stock, liquidation or dissolution) all or substantially all of the property, assets or business of the Parent and its Subsidiaries taken as a whole.
- 9.4 Limitation on Asset Transfers to Foreign Subsidiaries. With respect to the Parent or any Domestic Subsidiary, convey, sell, lease, assign, transfer or otherwise dispose of (collectively, a "transfer") any of its property, business or assets (including, without limitation leasehold interests), whether now owned or hereafter acquired, to any Foreign Subsidiary, except such transfers which, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect.
- 9.5 Limitation on Subordinated Debt. Permit any Subsidiary of the Parent (other than the Company) to create, incur, assume or suffer to exist any subordinated indebtedness other than (a) subordinated indebtedness of a Person which becomes a Subsidiary after the date hereof to the extent such indebtedness existed at the time such Person became a Subsidiary and was not incurred in anticipation thereof and any refinancings of such indebtedness after such time so long as the principal amount thereof is not increased or (b) subordinated indebtedness of such Subsidiary held by the Parent or any other Subsidiary of the Parent.

If any of the following events shall occur and be

continuing:

- (a) Any Borrower shall fail to pay any principal of any Loan or Note when due in accordance with the terms thereof or hereof; or any Borrower shall fail to pay any interest on any Loan or Note, or any other amount payable hereunder, within five Business Days after any such interest or other amount becomes due in accordance with the terms thereof or hereof; or
- (b) Any representation or warranty made, or pursuant to subsection 7.2, deemed made, by any Loan Party herein or in any other Loan Document or which is contained in any certificate, document or financial or other statement furnished at any time under or in connection with this Agreement shall prove to have been incorrect in any material adverse respect on or as of the date made or deemed made; or
- (c) The Parent or any Subsidiary shall default in the observance or performance of any agreement contained in subsection 9.1, 9.3, 9.4 or 9.5; or
- (d) Any Loan Party shall default in the observance or performance of any other agreement contained in this Agreement (other than as provided in paragraphs (a) through (c) of this Section), and such default shall continue unremedied for a period of 30 days; or
- (e) The Parent or any of its Subsidiaries (other than the Excluded Subsidiaries) shall (i) default in any payment of principal of or interest on, or any other amount payable with respect to, any (A) Domestic Indebtedness (other than the Notes and Loans) in an aggregate principal amount for all such Domestic Indebtedness of \$10,000,000 or more, or (B) Foreign Subsidiary Indebtedness (other than the Notes and Loans) in an aggregate principal amount for all such Foreign Subsidiary Indebtedness of \$20,000,000 or more, beyond the period of grace (not to exceed 30 days), if any, provided in the instrument or agreement under which such Indebtedness was created; or (ii) default in the observance or performance of any other agreement relating to any such Indebtedness in the amounts specified in clause (i) above or contained in any instrument or agreement evidencing, securing or relating thereto, or any other event shall occur or condition exist in any case which continues uncured or unwaived (and, if waived, without any change in the material terms of such Indebtedness) after the expiration of all applicable grace periods, the effect of which default or other event or condition is to cause, or to permit the holder or holders of such Indebtedness (or a trustee or agent on behalf of such holder or holders) to cause, with

the giving of notice if required, such Indebtedness to become due prior to its stated maturity; or

- (f) (i) The Parent or any Principal Subsidiary (other than the Excluded Subsidiaries) shall commence any case, proceeding or other action (A) under any existing or future law of any jurisdiction, domestic or foreign, relating to bankruptcy, insolvency, reorganization or relief of debtors, seeking to have an order for relief entered with respect to it, or seeking to adjudicate it a bankrupt or insolvent, or seeking reorganization, arrangement, adjustment, winding-up, liquidation, dissolution, composition or other relief with respect to it or its debts, or (B) seeking appointment of a receiver, trustee, custodian or other similar official for it or for all or any substantial part of its assets, or the Parent or any such Principal Subsidiary shall make a general assignment for the benefit of its creditors; or (ii) there shall be commenced against the Parent or any such Principal Subsidiary any case, proceeding or other action of a nature referred to in clause (i) above which (A) results in the entry of an order for relief or any such adjudication or appointment or (B) remains undismissed, undischarged or unbonded for a period of 60 days; or (iii) there shall be commenced against the Parent or any such Principal Subsidiary any case, proceeding or other action seeking issuance of a warrant of attachment, execution, distraint or similar process against all or any substantial part of its assets which results in the entry of an order for any such relief which shall not have been vacated, discharged, or stayed or bonded pending appeal within 60 days from the entry thereof; or (iv) the Parent or any such Principal Subsidiary shall take any action in furtherance of, or indicating its consent to, approval of, or acquiescence in, any of the acts set forth in clause (i), (ii) or (iii) above; or (v) the Parent or any such Principal Subsidiary shall generally not, or shall be unable to, or shall admit in writing its inability to, pay its debts as they become due; or
- (g) (i) Any Person shall engage in any non-exempt "prohibited transaction" (as defined in Section 406 of ERISA or Section 4975 of the Code) involving any Plan, (ii) any "accumulated funding deficiency" (as defined in Section 302 of ERISA), whether or not waived, shall exist with respect to any Plan, (iii) a Reportable Event shall occur with respect to, or judicial proceedings shall commence to have a trustee appointed, or a trustee shall be appointed, to administer or to terminate, any Single Employer Plan, which Reportable Event or commencement of judicial proceedings or appointment of a trustee is, in the reasonable opinion of the Majority Banks, likely to result in the termination of such Plan for purposes of Title IV of ERISA, (iv) any Single Employer Plan shall terminate for purposes of Title IV of ERISA, (v) the Company or any Commonly Controlled Entity shall, or in the reasonable opinion of the Majority Banks is

likely to, incur any liability in connection with a withdrawal from, or the Insolvency or Reorganization of, a Multiemployer Plan or (vi) any other event or condition shall occur or exist, with respect to a Plan; and in each case in clauses (i) through (vi) above, such event or condition, together with all other such events or conditions, if any, could reasonably be expected to subject the Company or any of its Subsidiaries to any tax, penalty or other liabilities which in the aggregate would have a Material Adverse Effect; or

- (h) One or more judgments or decrees shall be entered against the Parent or any of its Subsidiaries in aggregate amounts (not paid or fully covered by insurance) of \$10,000,000 or more and all such judgments or decrees shall not have been vacated, discharged, stayed or bonded pending appeal within 60 days from the entry thereof; or
- (i) The Parent shall cease to own directly or indirectly of record and beneficially free and clear of Liens at least 75% of the shares of the issued and outstanding capital stock of the Company;

then, and in any such event, (A) if such event is an Event of Default specified in clause (i), (ii) or (iii) of paragraph (f) above with respect to any of the Borrowers, automatically the Commitments to such Borrower shall immediately terminate and the Loans made to such Borrower hereunder (with accrued interest thereon) and all other amounts owing under this Agreement and the Notes of such Borrower shall immediately become due and payable, and (B) if such event is any other Event of Default, either or both of the following actions may be taken: (i) with the consent of the Majority Banks, the Administrative Agent may, or upon the request of the Majority Banks, the Administrative Agent shall, by notice to the Company declare the Commitments of any or all of the Borrowers to be terminated forthwith, whereupon such Commitments shall immediately terminate; and (ii) with the consent of the Majority Banks, the Administrative Agent may, or upon the request of the Majority Banks, the Administrative Agent shall, by notice of default to the Company and the Parent, declare the Loans hereunder made to any or all of the Borrowers (with accrued interest thereon) and all other amounts owing by such Borrower under this Agreement and the Notes of such Borrower to be due and payable forthwith, whereupon the same shall immediately become due and payable. Except as expressly provided above in this Section, presentment, demand, protest and all other notices of any kind are hereby expressly waived.

SECTION 11. THE ADMINISTRATIVE AGENT

and appoints Chase as the Administrative Agent of such Bank under this Agreement and the other Loan Documents, and each such Bank irrevocably authorizes Chase, as the Administrative Agent for such Bank, to take such action on its behalf under the provisions of this Agreement and the other Loan Documents and to exercise such powers and perform such duties as are expressly delegated to the Administrative Agent by the terms of this Agreement and the other Loan Documents, together with such other powers as are reasonably incidental thereto. Notwithstanding any provision to the contrary elsewhere in this Agreement, the Administrative Agent shall not have any duties or responsibilities, except those expressly set forth herein, or any fiduciary relationship with any Bank, and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or any other Loan Document or otherwise exist against the Administrative Agent.

11.2 Delegation of Duties. The Administrative Agent may execute any of its duties under this Agreement and the other Loan Documents by or through agents or attorneys-in-fact and shall be entitled to advice of counsel concerning all matters pertaining to such duties. The Administrative Agent shall not be responsible for the negligence or misconduct of any agents or attorneys-in-fact selected by it with reasonable care.

11.3 Exculpatory Provisions. Neither the Administrative Agent nor any of its officers, directors, employees, agents, attorneys-in-fact or Affiliates shall be (i) liable for any action lawfully taken or omitted to be taken by it or such Person under or in connection with this Agreement or any other Loan Document (except for its or such Person's own gross negligence or willful misconduct) or (ii) responsible in any manner to any of the Banks for any recitals, statements, representations or warranties made by any Loan Party or any officer thereof contained in this Agreement or any other Loan Document or in any certificate, report, statement or other document referred to or provided for in, or received by the Administrative Agent under or in connection with, this Agreement or any other Loan Document or for the value, validity, effectiveness, genuineness, enforceability or sufficiency of this Agreement or the Notes or any other Loan Document or for any failure of any Loan Party to perform its obligations hereunder or thereunder. The Administrative Agent shall not be under any obligation to any Bank to ascertain or to inquire as to the observance or performance of any of the agreements contained in, or conditions of, this Agreement or any other Loan Document, or to inspect the properties, books or records of any Loan Party.

11.4 Reliance by Administrative Agent. The Administrative Agent shall be entitled to rely, and shall be fully protected in relying, upon any note, writing, resolution, notice, consent, certificate, affidavit, letter, cablegram, telegram, telecopy, statement, order or other document or conversation believed by it to be genuine and correct and to have been signed, sent or made by the proper Person or Persons and upon advice and statements of legal counsel (including, without limitation, counsel to the Company, the Parent or any other Borrower), independent accountants and other experts selected by the Administrative Agent. The Administrative Agent may deem and treat the payee of any Note as the owner thereof for all purposes unless a written notice of assignment, negotiation or transfer thereof shall have been filed with the Administrative Agent. The Administrative Agent shall be fully justified in failing or refusing to take any action under this Agreement or any other Loan Document unless it shall first receive such advice or concurrence of the Majority Banks as it deems appropriate or it shall first be indemnified to its satisfaction by the Banks against any and all liability and expense which may be incurred by it by reason of taking or continuing to take any such action. The Administrative Agent shall in all cases be fully protected in acting, or in refraining from acting, under this Agreement and the Notes and the other Loan Documents in accordance with a request of the Majority Banks, and such request and any action taken or failure to act pursuant thereto shall be binding upon all the Banks and all future holders of the Notes.

11.5 Notice of Default. The Administrative Agent shall not be deemed to have knowledge or notice of the occurrence of any Default or Event of Default hereunder unless the Administrative Agent has received notice from a Bank or any Loan Party referring to this Agreement, describing such Default or Event of Default and stating that such notice is a "notice of default". In the event that the Administrative Agent receives such a notice, the Administrative Agent shall give notice thereof to the Banks. The Administrative Agent shall take such action with respect to such Default or Event of Default as shall be reasonably directed by the Majority Banks; provided that unless and until the Administrative Agent shall have received such directions, the Administrative Agent may (but shall not be obligated to) take such action, or refrain from taking such action, with respect to such Default or Event of Default as it shall deem advisable in the best interests of the Banks.

11.6 Non-Reliance on Administrative Agent and Other Banks. Each Bank expressly acknowledges that neither the Administrative Agent nor any of its officers, directors, employees, agents, attorneys-in-fact or Affiliates has made any representations or warranties to it and that no act by the Administrative Agent hereinafter taken, including any review of the affairs of the Loan Parties, shall be deemed to constitute any representation or warranty by the Administrative Agent to any Bank. Each Bank represents to the Administrative Agent that it has, independently and without reliance upon the Administrative

49

Agent or any other Bank, and based on such documents and information as it has deemed appropriate, made its own appraisal of and investigation into the business, operations, property, financial and other condition and creditworthiness of the Loan Parties and made its own decision to make its Loans hereunder and enter into this Agreement. Each Bank also represents that it will, independently and without reliance upon the Administrative Agent or any other Bank, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit analysis, appraisals and decisions in taking or not taking action under this Agreement and the other Loan Documents, and to make such investigation as it deems necessary to inform itself as to the business, operations, property, financial and other condition and creditworthiness of the Loan Parties. Except for notices, reports and other documents expressly required to be furnished to the Banks by the Administrative Agent hereunder, the Administrative Agent shall not have any duty or responsibility to provide any Bank with any credit or other information concerning the business, operations, property, condition (financial or otherwise), prospects or creditworthiness of the Loan Parties which may come into the possession of the Administrative Agent or any of its officers, directors, employees, agents, attorneys-in-fact or Affiliates.

11.7 Indemnification. The Banks agree to indemnify the Administrative Agent in its capacity as such (to the extent not reimbursed by the Loan Parties and without limiting the obligation of the Parent, the Company and any other Borrower to do so), ratably according to the respective amounts of their Commitments as in effect on the date on which the claim for indemnity by the Administrative Agent is sought, from and against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind whatsoever which may at any time (including, without limitation, at any time following the payment of the Notes) be imposed on, incurred by or asserted against the Administrative Agent in any way relating to or arising out of this Agreement, any of the other Loan Documents or any documents contemplated by or referred to herein or therein or the transactions contemplated hereby or thereby or any action taken or omitted by the Administrative Agent under or in connection with any of the foregoing; provided that no Bank shall be liable for the payment of any portion of such liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements resulting solely from the Administrative Agent's gross negligence or willful misconduct. The agreements in this subsection shall survive the payment of the Notes and all other amounts payable hereunder.

11.8 Administrative Agent in Its Individual Capacity. The Administrative Agent and its Affiliates may make loans to, accept deposits from and generally engage in any kind of business with the Parent, the Company or any other Borrower as though the Administrative Agent were not the Administrative Agent hereunder and under the other Loan Documents. With respect to its Loans

50

made or renewed by it and any Note issued to it, the Administrative Agent shall have the same rights and powers under this Agreement and the other Loan Documents as any Bank and may exercise the same as though it were not the Administrative Agent, and the terms "Bank" and "Banks" shall include the Administrative Agent in its individual capacity.

11.9 Successor Administrative Agent. The Administrative Agent may resign as Administrative Agent upon 10 days' notice to the Banks. If the Administrative Agent shall resign as Administrative Agent under this Agreement and the other Loan Documents, then the Majority Banks shall appoint from among the Banks a successor agent for the Banks, which successor agent shall be approved by the Company, whereupon such successor agent shall succeed to the rights, powers and duties of the Administrative Agent, and the term "Administrative Agent" shall mean such successor agent effective upon its appointment, and the former Administrative Agent's rights, powers and duties as Administrative Agent shall be terminated, without any other or further act or deed on the part of such former Administrative Agent or any of the parties to this Agreement or any holders of the Notes. After any retiring Administrative Agent's resignation as Administrative Agent, the provisions of this subsection shall inure to its benefit as to any actions taken or omitted to be taken by it while it was Administrative Agent under this Agreement and the other Loan Documents.

SECTION 12. GUARANTEES

12.1 Parent Guarantee. The Parent hereby unconditionally and irrevocably guarantees to the Administrative Agent and the Banks the prompt and complete payment and performance by each of the Borrowers when due (whether at the stated maturity, by acceleration or otherwise) of the Obligations owing to the Administrative Agent and the Banks by such Borrowers. This guarantee (the "Parent Guarantee") shall remain in full force and effect until the Obligations of each of the Borrowers are indefeasibly paid in full, notwithstanding that from time to time prior thereto any Borrower may be free from any Obligations. The Parent agrees that whenever, at any time, or from time to time, it shall make any payment to the Administrative Agent or any Bank on account of its liability under this Parent Guarantee, it will notify the Administrative Agent and such Bank in writing that such payment is made under this Parent Guarantee for such purpose. No payment or payments made by any Borrower or any other Person or received or collected by the Administrative Agent or any Bank from any Borrower or any other Person by virtue of any action or proceeding or any offset or appropriation or application, at any time or from time to time, in reduction of or in payment of the Obligations of such Borrower shall be deemed to modify, reduce, release or otherwise affect the liability of the Parent under this Parent Guarantee, which shall remain obligated under this Parent Guarantee,

notwithstanding any such payment or payments until the Obligations are paid in full.

12.2 Company Guarantee. The Company hereby unconditionally and irrevocably guarantees to the Administrative Agent and the Banks, the prompt and complete payment and performance by each of the other Borrowers when due (whether at the stated maturity, by acceleration or otherwise) of the Obligations owing to the Administrative Agent and the Banks by such Borrowers. This guarantee (the "Company Guarantee") shall remain in full force and effect until the Obligations of each such Borrower are indefeasibly paid in full, notwithstanding that from time to time prior thereto any such Borrower may be free from any Obligations. The Company agrees that whenever, at any time, or from time to time, it shall make any payment to the Administrative Agent or any Bank on account of its liability under this Company Guarantee, it will notify the Administrative Agent and such Bank in writing that such payment is made under this Company Guarantee for such purpose. No payment or payments made by any such Borrower or any other Person or received or collected by the Administrative Agent or any Bank from any such Borrower or any other Person by virtue of any action or proceeding or any offset or appropriation or application, at any time or from time to time, in reduction of or in payment of the Obligations of such Borrowers shall be deemed to modify, reduce, release or otherwise affect the liability of the Company under this Company Guarantee, which shall remain obligated under this Company Guarantee, notwithstanding any such payment or payments until the Obligations of such Borrowers are paid in full.

12.3 No Subrogation, Contribution, Reimbursement or Indemnity. Notwithstanding anything to the contrary in the Parent Guarantee and the Company Guarantee (together, the "Guarantees", each a "Guarantee"), the Parent and the Company (together, the "Guaranteeing Parties," each a "Guaranteeing Party") hereby irrevocably waives all rights which may have arisen in connection with its Guarantee to be subrogated to any of the rights (whether contractual, under the Bankruptcy Code, including Section 509 thereof, under common law or otherwise) of the Administrative Agent or any Bank against the Company or any other Borrowers (together, the "Guaranteed Parties", each a "Guaranteed Party") for the payment of the Obligations. Each Guaranteeing Party hereby further irrevocably waives all contractual, common law, statutory or other rights of reimbursement, contribution, exoneration or indemnity (or any similar right) from or against any Guaranteed Party or Parties or any other Person which may have arisen in connection with its Guarantee. So long as the Obligations remain outstanding, if any amount shall be paid by or on behalf of any Guaranteed Party to the Guaranteeing Party on account of any of the rights waived in this subsection, such amount shall be held by such Guaranteeing Party in trust, segregated from other funds of such Guaranteeing Party, and shall, forthwith upon receipt by such Guaranteeing Party, be turned over to the Administrative Agent in the exact form received by such Guaranteeing Party (duly endorsed by such

Guaranteeing Party to the Administrative Agent, if required), to be applied against the Obligations of such Guaranteed Party or Parties, whether matured or unmatured, in such order as the Administrative Agent may determine. The provisions of this subsection as they apply to each of the Guaranteeing Parties shall survive the payment in full of the Obligations of its Guaranteed Party or Parties.

12.4 Amendments, etc., with respect to the Obligations. Each Guaranteeing Party shall remain obligated under its Guarantee notwithstanding that, without any reservation of rights against such Guaranteeing Party, and without notice to or further assent by such Guaranteeing Party, any demand for payment of any of the Obligations made by the Administrative Agent or any Bank may be rescinded by the Administrative Agent or such Bank, and any of the Obligations continued, and the Obligations, or the liability of any other party upon or for any part thereof, or any collateral security or guarantee therefor or right of offset with respect thereto, may, from time to time, in whole or in part, be renewed, extended, amended, modified, accelerated, compromised, waived, surrendered or released by the Administrative Agent or any Bank, and this Agreement, the Notes and the other Loan Documents may be amended, modified, supplemented or terminated, in whole or in part, as the Administrative Agent or the Banks (or the Majority Banks, as the case may be) may deem advisable from time to time in accordance with the provisions of subsection 13.1(a), and any collateral security, guarantee or right of set-off at any time held by the Administrative Agent or any Bank for the payment of the Obligations may be sold, exchanged, waived, surrendered or released. Neither the Administrative Agent nor any Bank shall have any obligation to protect, secure, perfect or insure any Lien at any time held by it as security for the Obligations or for the obligations of any Guaranteeing Party under its Guarantee or any property subject thereto.

Party waives any and all notice of the creation, renewal, extension or accrual of any of the Obligations and notice of or proof of reliance by the Administrative Agent or any Bank upon its Guarantee or acceptance of its Guarantee; the Obligations, and any of them, shall conclusively be deemed to have been created, contracted or incurred in reliance upon the Guarantees; and all dealings between the Borrowers and the Parent, on the one hand, and the Administrative Agent and the Banks, on the other, shall likewise be conclusively presumed to have been had or consummated in reliance upon the Guarantees. Each Guaranteeing Party waives diligence, presentment, protest, notice of intent to accelerate, notice of acceleration, demand for payment and notice of default or nonpayment to or upon any Guaranteed Party or such Guaranteeing Party with respect to the Obligations. The Guarantee shall be construed as a continuing, absolute and unconditional guarantee of payment without regard to (a) the validity or enforceability of this Agreement, any Note, any other Loan Document, any of the Obligations or any collateral

security therefor or guarantee or right of set-off with respect thereto at any time or from time to time held by the Administrative Agent or any Bank, (b) any defense, offset or counterclaim (other than a defense of payment or performance) which may at any time be available to or be asserted by any of the Guaranteed Parties against the Administrative Agent or any Bank or (c) any other circumstance whatsoever (with or without notice to or knowledge of any of the Guaranteed Parties or such Guaranteeing Party) which constitutes, or might be construed to constitute, an equitable or legal discharge of any of the Guaranteed Parties for the Obligations of such Guaranteed Party, or of such Guaranteeing Party under its Guarantee, in bankruptcy or in any other instance. When the Administrative Agent is pursuing its rights and remedies hereunder against any Guaranteeing Party, the Administrative Agent or any Bank may, but shall be under no obligation to, pursue such rights and remedies as it may have against its Guaranteed Party or any other Person or against any collateral security or guarantee for the Obligations or any right of offset with respect thereto, and any failure by the Administrative Agent or any Bank to pursue such other rights or remedies or to collect any payments from such Guaranteed Party or such other Person or to realize upon any such collateral security or guarantee or to exercise such right of offset or any release of such Guaranteed Party or such other Person or of any such collateral security, guarantee or right of offset, shall not relieve such Guaranteeing Party of any liability under its Guarantee, and shall not impair or affect the rights and remedies, whether express, implied or available as a matter of law, of the Administrative Agent and the Banks against such Guaranteeing Party.

12.6 Reinstatement. Each Guarantee shall continue to be effective, or be reinstated, as the case may be, if at any time payment, or any part thereof, of any of the Obligations of any Guaranteed Party thereunder is rescinded or must otherwise be restored or returned by the Administrative Agent or any Bank upon the insolvency, bankruptcy, dissolution, liquidation or reorganization of such Guaranteed Party or upon or as a result of the appointment of a receiver, intervenor or conservator of, or trustee or similar officer for, such Guaranteed Party or any substantial part of any of its property, or otherwise, all as though such payments had not been made.

12.7 Payments. Each Guaranteeing Party hereby agrees that the Obligations will be paid to the Administrative Agent for the benefit of the Administrative Agent and the Banks, as the case may be, without set-off or counterclaim in Dollars or Alternative Currency, as appropriate, in immediately available funds at the office of the Administrative Agent, c/o Loan and Agency Services Group (Clearing Account No. 144810547) located at One Chase Manhattan Plaza, 8th Floor, New York, New York 10081.

54

SECTION 13. MISCELLANEOUS

13.1 Amendments and Waivers; Replacement of Banks. (a) Neither this Agreement, any Note, any other Loan Document, nor any terms hereof or thereof may be amended, supplemented or modified except in accordance with the provisions of this subsection. With the written consent of the Majority Banks, the Administrative Agent, the Parent and the Company may, from time to time, enter into written amendments, supplements or modifications hereto and to the Notes, if any, and the other Loan Documents for the purpose of adding any provisions to this Agreement or the Notes, if any, or the other Loan Documents or changing in any manner the rights of the Banks, the Parent or of the Borrowers hereunder or thereunder or waiving, on such terms and conditions as the Administrative Agent may specify in such instrument, any of the requirements of this Agreement or the Notes, if any, or the other Loan Documents or any Default or Event of Default and its consequences; provided, however, that no such waiver and no such amendment, supplement or modification shall (i) reduce the amount or extend the maturity of any Loan or Note or any installment thereof, or reduce the rate or extend the time of payment of interest thereon, or reduce any fee payable to any Bank hereunder, or change the amount of any Bank's Commitment, in each case without the consent of the Bank affected thereby, (ii) amend, modify or waive any provision of this subsection or reduce the percentage specified in the definition of Majority Banks, or consent to the assignment or transfer by the Parent or any Borrower of any of its rights and obligations under this Agreement and the other Loan Documents, or amend, modify or waive any provision of Section 12, in each case without the written consent of all the Banks, or (iii) amend, modify or waive any provision of Section 11 without the written consent of the then Administrative Agent. Any such waiver and any such amendment, supplement or modification shall apply equally to each of the Banks and shall be binding upon the Parent, the Borrowers, the Banks, the Administrative Agent, all future holders of the Notes, if any, and all future obligees under the Loans. In the case of any waiver, the Parent, the Borrowers, the Banks and the Administrative Agent shall be restored to their former position and rights hereunder and under the outstanding Loans or Notes, if any, and any other Loan Documents, and any Default or Event of Default waived shall be deemed to be cured and not continuing; but no such waiver shall extend to any subsequent or other Default or Event of Default, or impair any right consequent thereon.

(b) Notwithstanding anything to the contrary contained in subsection 13.1(a), so long as no Default or Event of Default has occurred and is continuing the Borrowers and the Parent shall be permitted in their discretion (but, if any Revolving Credit Loans are then outstanding, with the consent of the Majority Banks (which consent shall not be unreasonably withheld)) to amend this Agreement to replace one or more Banks without the consent of any Bank to be so replaced pursuant to this subsection 13.1(b) (a "Replaced Bank") and to provide for (w) the

termination of the Commitments of such Replaced Bank, (x) the addition to this Agreement of one or more other banking institutions, or an increase in the Commitments of one or more of the other Banks (with the consent of such other Banks), so that the total Commitments after giving effect to such amendment shall be in the same amount as the total Commitments immediately before giving effect to such amendment, (y) if any Loans are outstanding at the time of such amendment, the making of such additional Loans by such new financial institutions or other Bank or Banks, as the case may be, as may be necessary to repay in full the outstanding Loans of such Replaced Bank together with interest thereon and all accrued fees and indemnities with respect thereto immediately before giving effect to such amendment and (z) such other modifications to this Agreement as may be necessary to effect the replacement of such Replaced Bank.

(c) Notwithstanding anything to the contrary contained in paragraph (a) or (b) of this subsection 13.1, if as a result of a change in the Requirement of Law after the date hereof any Borrower or the Parent has become obligated to, or reasonably believes that it will become obligated to pay to any Bank any increased amount pursuant to subsection 5.11, 5.12 or 5.13, and such Bank shall not have waived payment of such increased amounts, then the Borrowers and the Parent may, if no Default or Event of Default has occurred and is continuing and payment of any such increased amounts as have become due has been made or appropriately provided for, upon five Business Days' notice to the Administrative Agent and such Bank, amend this Agreement, without the consent of any Bank or the Administrative Agent, to replace any one or more of the Banks to which such increased amounts have become payable or would become payable and to provide for the matters referred to in clauses (w), (x), (y) and (z) of subsection 13.1(b), and such replaced Bank or Banks shall be deemed to be Replaced Banks for purposes of such clauses.

13.2 Notices. All notices, requests and demands to or upon the respective parties hereto to be effective shall be in writing (including by telecopy), and, unless otherwise expressly provided herein, shall be deemed to have been duly given or made on receipt, addressed as follows in the case of the Company, the Parent and the Administrative Agent, as set forth in paragraph 5 of the Notice of Additional Borrower relating to any Borrower other than the Company, in the case of such other Borrower, and as set forth in Schedule I in the case of the other parties hereto, or to such other address as may be hereafter notified by the respective parties hereto and any future holders of the Notes, if any, or any future obligees under the Loans:

The Company: W. R. Grace & Co.-Conn

One Town Center Road Boca Raton, Florida 33486-1010

Attention: Treasurer Telecopy: (561) 362-1944 Telephone: (561) 362-1949 The Parent: W. R. Grace & Co.

One Town Center Road

Boca Raton, Florida 33486-1010 Attention: Treasurer Telecopy: (561) 362-1944 Telephone: (561) 362-1949

The

Administrative

Agent:

The Chase Manhattan Bank

270 Park Avenue

New York, New York 10017 Attention: Robert Sacks Telecopy: (212) 270-4118 Telephone: (212) 270-1355

With a copy

Loan and Agency Services Group

One Chase Manhattan Plaza, 8th Floor

New York, New York 10081 Attention: Margaret Swales Telecopy: (212) 552-5662 Telephone: (212) 552-7472

13.3 No Waiver; Cumulative Remedies. No failure to exercise and no delay in exercising, on the part of the Administrative Agent or any Bank, any right, remedy, power or privilege hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. The rights, remedies, powers and privileges herein provided are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law.

13.4 Survival of Representations and Warranties. All representations and warranties made hereunder and in any document, certificate or statement delivered pursuant hereto or in connection herewith shall survive the execution and delivery of this Agreement and the Notes, if any.

13.5 Payment of Expenses and Taxes. The Company agrees (a) to pay or reimburse the Administrative Agent for all its out-of-pocket costs and expenses incurred in connection with the development, preparation and execution of, and any amendment, supplement or modification to, this Agreement and any Notes and the other Loan Documents and any other documents prepared in connection herewith or therewith, and the consummation and administration of the transactions contemplated hereby and thereby, including, without limitation, the fees and disbursements of counsel to the Administrative Agent, (b) to pay or reimburse each Bank and the Administrative Agent for all its costs and expenses incurred in connection with the enforcement or preservation of any rights under this Agreement, any Notes, the other Loan Documents and any such other documents, including,

without limitation, fees and disbursements of counsel to the Administrative Agent and to the several Banks, and (c) to pay, indemnify, and hold each Bank and the Administrative Agent harmless from, any and all recording and filing fees and any and all liabilities with respect to, or resulting from any delay in paying, stamp, excise and other transactional taxes, if any, which may be payable or determined to be payable in connection with the execution and delivery of, or consummation or administration of any of the transactions contemplated by, or any amendment, supplement or modification of, or any waiver or consent under or in respect of, this Agreement, any Notes, the other Loan Documents and any such other documents, and (d) to pay, indemnify, and hold each Bank and the Administrative Agent harmless from and against any and all other liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind or nature whatsoever with respect to the execution, delivery and performance by the Loan Parties, and administration and enforcement by the Administrative Agent and the Banks of this Agreement, any Notes and the other Loan Documents and any such other documents (all the foregoing, collectively, the "indemnified liabilities"), provided, that the Company shall have no obligation hereunder to the Administrative Agent or any Bank with respect to indemnified liabilities arising from (i) the gross negligence or willful misconduct of the Administrative Agent or any such Bank, (ii) legal proceedings commenced against the Administrative Agent or any such Bank by any security holder or creditor thereof arising out of and based upon rights afforded any such security holder or creditor solely in its capacity as such, or (iii) legal proceedings commenced against the Administrative Agent or any such Bank by any other Bank or by any Transferee (as defined in subsection 13.6). The agreements in this subsection shall survive repayment of the Loans or Notes, if any, and all other amounts payable hereunder.

13.6 Successors and Assigns; Participations; Purchasing

Banks.

- (a) This Agreement shall be binding upon and inure to the benefit of the Parent, the Borrowers, the Banks, the Administrative Agent, all future holders of the Notes, if any, all future obligees under the Loans and their respective successors and assigns, except that neither the Parent nor any Borrower may assign or transfer any of its rights or obligations under this Agreement without the prior written consent of each Bank.
- (b) Any Bank may, in the ordinary course of its commercial banking business and in accordance with applicable law, at any time sell to one or more banks or other entities ("Participants") participating interests in any Loan owing to such Bank, any Note held by such Bank, any Commitments of such Bank or any other interest of such Bank hereunder and under the other Loan Documents. In the event of any such sale by a Bank of participating interests to a Participant, such Bank's obligations

under this Agreement to the other parties to this Agreement shall remain unchanged, such Bank shall remain solely responsible for the performance thereof, such Bank shall remain the holder of any such Note, if any, and the obligee under any such Loan for all purposes under this Agreement and the other Loan Documents, and the Parent, the Borrowers and the Administrative Agent shall continue to deal solely and directly with such Bank in connection with such Bank's rights and obligations under this Agreement and the other Loan Documents. The Parent and each of the Borrowers agrees that if amounts outstanding under this Agreement and the Loans or the Notes, if any, are due or unpaid, or shall have been declared or shall have become due and payable upon the occurrence of an Event of Default, each Participant shall be deemed to have the right of set-off in respect of its participating interest in amounts owing under this Agreement and any Loan or Note to the same extent as if the amount of its participating interest were owing directly to it as a Bank under this Agreement or any Loan or Note, provided that such Participant shall only be entitled to such right of set-off if it shall have agreed in the agreement pursuant to which it shall have acquired its participating interest to share with the Banks the proceeds thereof as provided in subsection 13.7. The Parent and each of the Borrowers also agrees that each Participant shall be entitled to the benefits of subsections 5.11, 5.12, 5.13 and 13.5 with respect to its participation in the Commitments and the Loans outstanding from time to time; provided, that no Participant shall be entitled to receive any greater amount pursuant to such subsections than the transferor Bank would have been entitled to receive in respect of the amount of the participation transferred by such transferor Bank to such Participant had no such transfer occurred.

(c) Any Bank may, in the ordinary course of its commercial banking business and in accordance with applicable law, at any time sell to any Bank or any affiliate thereof and, with the consent of the Company and upon notice to the Administrative Agent, to one or more additional banks or financial institutions ("Purchasing Banks") all or any part of its rights and obligations under this Agreement and the Loans or the Notes, if any, pursuant to a Commitment Transfer Supplement, substantially in the form of Exhibit H, executed by such Purchasing Bank, such transferor Bank (and, in the case of a Purchasing Bank that is not then a Bank or an affiliate thereof, by the Company and the Administrative Agent) and delivered to the Administrative Agent for its acceptance and recording in the Register. Upon such execution, delivery, acceptance and recording, from and after the Transfer Effective Date determined pursuant to such Commitment Transfer Supplement, (x) the Purchasing Bank thereunder shall be a party hereto and, to the extent provided in such Commitment Transfer Supplement, have the rights and obligations of a Bank hereunder with a Commitment as set forth therein, and (y) the transferor Bank thereunder shall, to the extent provided in such Commitment Transfer Supplement, be released from its obligations under this Agreement (and, in the case of a Commitment Transfer Supplement covering all or the remaining portion of a transferor Bank's rights and obligations

under this Agreement, such transferor Bank shall cease to be a party hereto). Such Commitment Transfer Supplement shall be deemed to amend this Agreement to the extent, and only to the extent, necessary to reflect the addition of such Purchasing Bank and the resulting adjustment of Commitment Percentages arising from the purchase by such Purchasing Bank of all or a portion of the rights and obligations of such transferor Bank under this Agreement and the Loan or the Notes, if any. On or prior to the Transfer Effective Date determined pursuant to such Commitment Transfer Supplement, the relevant Borrower, at its own expense, if the Purchasing Bank so requests, shall execute and deliver to the Administrative Agent in exchange for any surrendered Revolving Credit Note and Bid Loan Note a new Revolving Credit Note and Bid Loan Note to the order of such Purchasing Bank in an amount equal to the Commitment assumed by it pursuant to such Commitment Transfer Supplement and, if the transferor Bank has retained a Commitment hereunder, new Notes to the order of the transferor Bank in an amount equal to the Commitment retained by it hereunder. Such new Notes shall be dated the Closing Date and shall otherwise be in the form of the Notes replaced thereby. Any Notes surrendered by the transferor Bank shall be returned by the Administrative Agent to the Company marked "cancelled".

- (d) The Administrative Agent shall maintain at its address referred to in subsection 13.2 a copy of each Commitment Transfer Supplement delivered to it and a register (the "Register") for the recordation of the names and addresses of the Banks and the Commitment of, and principal amount of the Loans owing to, each Bank from time to time. The entries in the Register shall be conclusive, in the absence of manifest error, and the Parent, the Borrowers, the Administrative Agent and the Banks may treat each Person whose name is recorded in the Register as the owner of the Loan recorded therein for all purposes of this Agreement. The Register shall be available for inspection by the Parent, the Borrowers or any Bank at any reasonable time and from time to time upon reasonable prior notice.
- (e) Upon its receipt of a Commitment Transfer Supplement executed by a transferor Bank and Purchasing Bank (and, in the case of a Purchasing Bank that is not then a Bank or an affiliate thereof, by the Company and the Administrative Agent) together with payment to the Administrative Agent of a registration and processing fee of \$3,500, the Administrative Agent shall (i) promptly accept such Commitment Transfer Supplement (ii) on the Transfer Effective Date determined pursuant thereto record the information contained therein in the Register and give notice of such acceptance and recordation to the Banks and the Company.

(f) The Parent and the Borrowers authorizes each Bank to disclose to any Participant or Purchasing Bank (each, a "Transferee") and any prospective Transferee any and all financial information in such Bank's possession concerning such Borrower and its affiliates which has been delivered to such Bank

60

by or on behalf of the Parent, the Company or such Borrower pursuant to this Agreement or which has been delivered to such Bank by or on behalf of the Parent, the Company or such Borrower in connection with such Bank's credit evaluation of such Borrower and its affiliates prior to becoming a party to this Agreement.

(g) If, pursuant to this subsection, any interest in this Agreement or any Note is transferred to any Transferee which is organized under the laws of any jurisdiction other than the United States or any state thereof, the transferor Bank shall cause such Transferee, concurrently with the effectiveness of such transfer, (i) to represent to the transferor Bank (for the benefit of the transferor Bank, the Administrative Agent, the Parent and the Borrowers) that under applicable law and treaties no taxes will be required to be withheld by the Administrative Agent, the Parent, the Borrowers or the transferor Bank with respect to any payments to be made to such Transferee in respect of the Loans, (ii) to furnish to the transferor Bank (and, in the case of any Purchasing Bank registered in the Register, the Administrative Agent, the Parent and the Company) either U.S. Internal Revenue Service Form 4224 or U.S. Internal Revenue Service Form 1001 (wherein such Transferee claims entitlement to complete exemption from U.S. federal withholding tax on all interest payments hereunder) and (iii) to agree (for the benefit of the transferor Bank, the Administrative Agent, the Parent and the Company) to provide the transferor Bank (and, in the case of any Purchasing Bank registered in the Register, the Administrative Agent, the Parent and the Company) a new Form 4224 or Form 1001 upon the expiration or obsolescence of any previously delivered form and comparable statements in accordance with applicable U.S. laws and regulations and amendments duly executed and completed by such Transferee, and to comply from time to time with all applicable U.S. laws and regulations with regard to such withholding tax exemption.

(h) Nothing herein shall prohibit any Bank from pledging or assigning any Note to any Federal Reserve Bank in accordance with applicable law.

13.7 Adjustments; Set-off.

(a) If any Bank (a "benefitted Bank") shall at any time receive any payment of all or part of its Revolving Credit Loans, or interest thereon, or receive any collateral in respect thereof (whether voluntarily or involuntarily, by set-off, pursuant to events or proceedings of the nature referred to in Section 10(f), or otherwise), in a greater proportion than any such payment to or collateral received by any other Bank, if any, in respect of such other Bank's Revolving Credit Loans, or interest thereon, such benefitted Bank shall purchase for cash from the other Banks such portion of each such other Bank's Loan, or shall provide such other Banks with the benefits of any such collateral, or the proceeds thereof, as shall be necessary to cause such benefitted Bank to share the excess payment or benefits of such collateral or proceeds ratably with each of the

Banks; provided, however, that if all or any portion of such excess payment or benefits is thereafter recovered from such benefitted Bank, such purchase shall be rescinded, and the purchase price and benefits returned, to the extent of such recovery, but without interest. Each Borrower agrees that each Bank so purchasing a portion of another Bank's Loan may exercise all rights of payment (including, without limitation, rights of set-off) with respect to such portion as fully as if such Bank were the direct holder of such portion.

- (b) In addition to any rights and remedies of the Banks provided by law, each Bank shall have the right, without prior notice to the Parent and the Borrowers, any such notice being expressly waived by the Parent and the Borrowers, to the extent permitted by applicable law, upon any amount not being paid when due and payable by any Borrower hereunder or under the Notes (whether at the stated maturity, by acceleration or otherwise) to set-off and appropriate and apply against such amount any and all deposits (general or special, time or demand, provisional or final), in any currency, and any other credits, indebtedness or claims, in any currency, in each case whether direct or indirect, absolute or contingent, matured or unmatured, at any time held or owing by such Bank or any branch or agency thereof to or for the credit or the account of the Parent or such Borrower. Each Bank agrees promptly to notify the Parent, the Borrowers and the Administrative Agent after any such set-off and application made by such Bank, provided that the failure to give such notice shall not affect the validity of such set-off and application.
- 13.8 Counterparts. This Agreement may be executed by one or more of the parties to this Agreement on any number of separate counterparts, and all of said counterparts taken together shall be deemed to constitute one and the same instrument. A set of the copies of this Agreement signed by all the parties shall be lodged with the Parent, the Company and the Administrative Agent.
- 13.9 Severability. Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.
- 13.10 Integration. This Agreement represents the agreement of the Parent, each Borrower, the Administrative Agent and the Banks with respect to the subject matter hereof, and there are no promises, undertakings, representations or warranties by the Administrative Agent or any Bank relative to subject matter hereof not expressly set forth or referred to herein, in the other Loan Documents or in any documentation entered into pursuant to subsection 3.1(b).

- 13.11 GOVERNING LAW. THIS AGREEMENT (INCLUDING SECTION 12)
 AND THE NOTES AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES UNDER THIS
 AGREEMENT AND THE NOTES SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN
 ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK.
- 13.12 Submission to Jurisdiction; Waivers. (a) The Parent, each Borrower, the Administrative Agent and the Banks hereby irrevocably and unconditionally:
 - (i) submits for itself and its property in any legal action or proceeding relating to this Agreement and the other Loan Documents to which it is a party, or for recognition and enforcement of any judgment in respect thereof, to the non-exclusive general jurisdiction of the Courts of the State of New York sitting in New York County, the courts of the United States of America for the Southern District of New York, and the appellate courts from any thereof:
 - (ii) consents that any such action or proceeding may be brought in such courts and waives any objection that it may now or hereafter have to the venue of any such action or proceeding in any such court or that such action or proceeding was brought in an inconvenient court and agrees not to plead or claim the same;
 - (iii) agrees that service of process in any such action or proceeding may be effected by mailing a copy thereof by registered or certified mail (or any substantially similar form of mail), postage prepaid, to the Parent or such Borrower at its address set forth in subsection 13.2 or, with respect to Borrowers other than the Company, the Notice of Additional Borrower relating to such Borrower or at such other address of which the Administrative Agent shall have been notified pursuant thereto;
 - (iv) agrees that nothing herein shall affect the right to effect service of process in any other manner permitted by law or shall limit the right to sue in any other jurisdiction; and
 - (v) waives, to the maximum extent not prohibited by law, any right it may have to claim or recover in any legal action or proceeding referred to in this subsection any special, exemplary, punitive or consequential damages.
- (b) Each Borrower other than the Company hereby appoints and empowers the Parent and the Company, 1114 Avenue of the Americas, New York, New York 10036-7794, Attention: Treasurer, as its authorized agent (the "Process Administrative Agent") to receive on behalf of such Borrower service of any and all process and documents in any such legal action or proceeding brought in a New York state or federal court sitting in New York City. It is understood that a copy of such process served on the

Process Administrative Agent will be promptly hand delivered or mailed (by registered or certified airmail if available), postage prepaid, to such Borrower at its address set forth in paragraph 5 of such Borrower's Notice of Additional Borrower, but the failure of such Borrower to receive such copy shall not affect in any way the service of such process on the Process Administrative Agent. If the Process Administrative Agent shall refuse or be prevented from acting as agent, notice thereof shall immediately be given by such Borrowers to the Administrative Agent by registered or certified airmail (if available), postage prepaid, and such Borrowers agree promptly to designate another agent in New York City, satisfactory to the Administrative Agent, to serve in place of the Process Administrative Agent and deliver to the Administrative Agent written evidence of such substitute agent's acceptance of such designation.

- 13.13 Acknowledgments. The Parent, each Borrower, the Administrative Agent and the Banks hereby acknowledge that:
 - (a) it has been advised by counsel in the negotiation, execution and delivery of this Agreement and the Notes and the other Loan Documents;
 - (b) neither the Administrative Agent nor any Bank has any fiduciary relationship with or duty to the Parent or such Borrower, as the case may be, arising out of or in connection with this Agreement or any of the other Loan Documents, and the relationship between Administrative Agent and Banks, on one hand, and the Parent and the Borrowers, on the other hand, in connection herewith or therewith is solely that of debtor and creditor; and
 - (c) as to any matter relating to any Loan Documents, no joint venture exists among the Banks or among the Parent, the Borrowers and the Banks.
- 13.14 WAIVERS OF JURY TRIAL. THE PARENT, THE BORROWERS, THE ADMINISTRATIVE AGENT AND THE BANKS HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVE TRIAL BY JURY IN ANY LEGAL ACTION OR PROCEEDING RELATING TO THIS AGREEMENT OR THE NOTES OR ANY OTHER LOAN DOCUMENT AND FOR ANY COUNTERCLAIM THEREIN.
- 13.15 Additional Borrowers. (a) Any Subsidiary of the Company shall have the right to become a "Borrower" hereunder, and to borrow hereunder subject to the terms and conditions hereof applicable to a Borrower and to the following additional conditions:
 - (i) the Company shall deliver a notice in substantially the form of Exhibit I hereto (a "Notice of Additional Borrower") signed by such Subsidiary and countersigned by the Parent and the Company to the Administrative Agent and the Banks stating that such Subsidiary desires to become a "Borrower" under this Agreement and agrees to be bound by the terms hereof. From

the time of receipt of such Notice of Additional Borrower by the Administrative Agent and the Banks and subject to the satisfaction of each condition precedent contained in such Notice of Additional Borrower, such Subsidiary shall be a "Borrower" hereunder with all of the rights and obligations of a Borrower hereunder; provided, however, that the Company may revoke a Notice of Additional Borrower with respect to any Subsidiary (other than the Company) upon five Business Days' written notice to the Administrative Agent, so long as such Borrower has no Obligations outstanding. No Notice of Additional Borrower relating to a Subsidiary may be revoked as to amounts owed by such Subsidiary to the Banks under this Agreement or any Notes or when an irrevocable notice pursuant to subsection 2.3, or a notice of acceptance pursuant to subsection 3.1 or 4.2, has been given by such Subsidiary as a Borrower and is effective;

- (ii) if such Subsidiary is a Foreign Subsidiary, if reasonably requested by the Majority Banks, such Notice of Additional Borrower shall be accompanied by an opinion of counsel for such Subsidiary as specified in paragraph 4(a)(ii) of such Notice of Additional Borrower:
- (iii) and the other conditions set forth in such Notice of Additional Borrower shall have been satisfied (including the representations and warranties contained therein being true and correct as of the date thereof).
- (b) Promptly, upon receipt of any Notice of Additional Borrower by the Administrative Agent, the Administrative Agent shall notify each Bank thereof, and shall deliver to each Bank copies of each document delivered to the Administrative Agent pursuant to such Notice of Additional Borrower.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed and delivered in New York, New York by their proper and duly authorized officers as of the day and year first above

W. R. GRACE & CO.-CONN.

By: /s/ Leonard R. Byrne

Name: Leonard R. Byrne Title: President -

Authorized Representative

W. R. GRACE & CO.

By: /s/ Bruce L. Hagins

Name: Bruce L. Hagins

Title: Authorized Representative

THE CHASE MANHATTAN BANK, as Administrative Agent and as a ${\tt Bank}$

By: /s/ Robert T. Sacks

Name: Robert T. Sacks

Title: Managing Director

66

NATIONSBANK, N.A. (SOUTH)

By: /s/ Andrew M. Airheart

Name: Andrew M. Airheart

Title: Senior Vice President

ABN AMRO BANK N.V., NEW YORK BRANCH

By: /s/ David W. Stack

Name: David W. Stack Title: Vice President

By: /s/ George M. Dugan

Name: Geroge M. Dugan Title: Vice President

BANK OF AMERICA NATIONAL TRUST AND SAVINGS ASSOCIATION

By: /s/ Laurens F. Schaad, Jr.

Name: Laurens F. Schaad, Jr. Title: Vice President

BANK OF NEW YORK

By: /s/ David C. Siegel

Name: David C. Siegel

Title: Assistant Vice President

THE BANK OF NOVA SCOTIA

By: /s/ William Brown

Name: William Brown Title: Vice President

BARCLAYS BANK PLC

By: /s/ George E. McKean
Name: George E. McKean
Title: Director

CITIBANK, N.A.

By: /s/ Mary Corkran

Name: Mary Corkran Title: Vice President

COMMERZBANK AG, ATLANTA AGENCY

By: /s/ W.D. Suttles

lomo. W.D. Cuttles

Name: W.D. Suttles Title: Vice President

By: /s/ Mary B. Smith

-----Name: Mary R Smith

Name: Mary B. Smith Title: Asst. Vice President

CREDIT LYONNAIS ATLANTA AGENCY

By: /s/ David M. Cawrse

Name: David M. Cawrse Title: First Vice President

CREDIT SUISSE FIRST BOSTON

By: /s/ James P. Moran

Name: James P. Moran Title: Director

By: /s/ J. Scott Karro

Name: J. Scott Karro Title: Associate

DRESDNER BANK AG, NEW YORK AND GRAND CAYMAN BRANCHES

By: /s/ Andrew P. Nesi Name: Andrew P. Nesi Title: Vice President By: /s/ John D. Padilla Name: John D. Padilla Title: Vice President FIRST UNION NATIONAL BANK OF FLORIDA By: /s/ Mary A. Morgan Name: Mary A. Morgan Title: Vice President & Senior Portfolio Manager MARINE MIDLAND BANK By: /s/ William M. Holland Name: William M. Holland Title: Vice President MORGAN GUARANTY TRUST COMPANY OF NEW YORK By: /s/ Jeffrey Hwang -----Name: Jeffrey Hwang Title: Vice President UNION BANK OF SWITZERLAND -NEW YORK By: /s/ Hamilton W. Bullard Name: Hamilton W. Bullard Title: Assistant Treasurer By: /s/ Samuel Azizo Name: Samuel Azizo

Title: Vice President

${\tt Commitments}$

BANK	COMMITMENT
The Chase Manhattan Bank	\$21,875,000
ABN AMRO Bank, New York Branch	\$21,875,000
Bank of America National Trust and Savings Assoc.	\$21,875,000
The Bank of New York	\$21,875,000
The Bank of Nova Scotia	\$21,875,000
Barclays Bank PLC	\$21,875,000
Citibank, N.A.	\$21,875,000
Commerzbank AG, Atlanta Agency	\$21,875,000
Credit Lyonnais, Atlanta Agency	\$21,875,000
Credit Suisse First Boston	\$21,875,000
Dresdner Bank AG	\$21,875,000
First Union National Bank of Florida	\$21,875,000
Marine Midland Bank	\$21,875,000
Morgan Guaranty Trust Company of New York	\$21,875,000
NationsBank, N.A. (South)	\$21,875,000
Union Bank of Switzerland	\$21,875,000
	\$350,000,000

W. R. GRACE & CO. AND SUBSIDIARIES WEIGHTED AVERAGE NUMBER OF SHARES AND EARNINGS USED IN PER SHARE COMPUTATIONS

For the three and six months ended June 30, 1997 and 1996 (Dollars in millions, except per share; shares in thousands)

	3 Mos. 6/30/97		6 Mos. 6/30/97	6/30/96
EARNINGS PER SHARE: Weighted average shares outstanding	72,761 ======	96,634 ======	73,999 =====	97,259 =====
Net income Dividends paid on preferred stocks	\$ 117.4 	\$ 333.9 (.2)	\$ 163.8 	\$ 397.5 (.3)
Income used in per share computation of earnings	\$ 117.4 ======	\$ 333.7 ======	\$ 163.8 ======	\$ 397.2
Net income per share	\$ 1.61	\$ 3.45	\$ 2.21	\$ 4.09
PRIMARY:				
Weighted average shares outstanding Dilutive effect (as determined by the application	72,761	96,634	73,999	97,259
of the treasury stock method)	2,072	1,878	2,253	2,022
Weighted average number of shares				
outstanding - primary	74,833 ======	98,512 ======	76,252 ======	99,281 ======
Net income	\$ 117.4	\$ 333.9 (.2)	\$ 163.8	\$ 397.5 (.3)
bividends para on preferred stocks				
Income used in per share computation of earnings and in per share computation of earnings				
assuming dilutive effect	\$ 117.4 ======	\$ 333.7 ======	\$ 163.8 ======	\$ 397.2 ======
Net income per share - primary	\$ 1.57	\$ 3.39	\$ 2.15	\$ 4.00
FULLY DILUTED:				
Weighted average shares outstanding Dilutive effect (as determined by the application	72,761	96,634	73,999	97,259
of the treasury stock method)	2,272	1,878	2,357	2,022
Weighted average number of shares				
outstanding - fully diluted	75,033	98,512	76,356	99,281
Net income	====== \$ 117.4	====== \$ 333.9	====== \$ 163.8	====== \$ 397.5
Dividends paid on preferred stocks		(.2)		(.3)
Income used in per chara computation of corrises				
Income used in per share computation of earnings and in per share computation of earnings				
assuming dilutive effect	\$ 117.4 ======	\$ 333.7 ======	\$ 163.8 ======	\$ 397.2 ======
Net income per share - fully diluted	\$ 1.56	\$ 3.39	\$ 2.15	\$ 4.00

W. R. GRACE & CO. AND SUBSIDIARIES COMPUTATION OF RATIO OF EARNINGS TO FIXED CHARGES AND COMBINED FIXED CHARGES AND PREFERRED STOCK DIVIDENDS (a) (in millions, except ratios) (Unaudited)

	Years Ended December 31,(c)						
	1996 (d)		1994 (f)				1996 (c)(j)
Net income/(loss) from continuing operations	\$213.8	\$(179.6)	\$(35.1)	\$ 28.1	\$ 7.7	\$163.8	\$278.6
Provision for/(benefit from) income taxes Income taxes of 50%-owned companies Equity in unremitted losses/(earnings)		(104.5) 	(42.6) 	16.4 .1	84.1 2.1	99.6	157.4
of less than 50%-owned companies	(.4)	. 8	(.6)	(.5)	(2.0)		
including amortization of capitalized interest Estimated amount of rental expense	160.8	179.8	138.5	122.7	162.7	40.3	93.1
deemed to represent the interest factor	8.4	8.5	10.1	11.3	14.0	2.7	5.5
Income/(loss) as adjusted	\$517.4	\$ (95.0)	\$ 70.3	\$ 178.1 =======	\$268.6	\$306.4	\$534.6 ======
Combined fixed charges and preferred stock dividends: Interest expense and related financing costs, including capitalized interest	\$177.1	\$ 195.5	\$143.2	\$ 122.8	\$176.3	\$ 49.6	\$104.2
deemed to represent the interest factor	8.4	8.5	10.1	11.3	14.0	2.7	5.5
Fixed charges Preferred stock dividend requirements(b)			153.3	134.1	190.3	52.3	109.7 .4
Combined fixed charges and preferred stock dividends	\$186.1	\$ 204.5	\$153.8 =======	\$ 134.9 ======	\$191.1 ======	\$ 52.3 =======	\$110.1 ======
Ratio of earnings to fixed charges		(k)	(k)	1.33	1.41	5.86	4.87
Ratio of earnings to combined fixed charges and preferred stock dividends		(k)	(k)	1.32	1.41	5.86	4.86 ======

- (a) Grace's preferred stocks were retired in 1996; for additional information, see Note 1 to the consolidated financial statements in the 1996 Form 10-K.
- (b) For each period with an income tax provision, the preferred stock dividend requirements have been increased to an amount representing the pretax earnings required to cover such requirements based on Grace's effective tax rate.
- (c) Certain amounts have been restated to conform to the 1997 presentation.
- (d) Includes a pretax gain on sales of businesses of \$326.4, offset by pretax provisions of \$229.1 for asbestos-related liabilities and insurance coverage and \$107.5 for restructuring costs and asset impairments.
- (e) Includes pretax provisions of \$275.0 for asbestos-related liabilities and insurance coverage; \$209.5 related to restructuring costs, asset impairments and other activities; \$77.0 for environmental liabilities at former manufacturing sites; and \$30.0 for corporate governance activities.
- (f) Includes a pretax provision of \$316.0 relating to asbestos-related liabilities and insurance coverage.
- (g) Includes a pretax provision of \$159.0 relating to asbestos-related liabilities and insurance coverage.
- (h) Includes a pretax provision of \$140.0 relating to a fumed silica plant in Belgium.
- (i) Includes a pretax gain of \$103.1 on the sale of Grace's specialty polymers business and a pretax provision of \$12.4 relating to restructuring costs.
- (j) Includes a pretax gain of \$326.4 on the sale of businesses, principally the water treatment and process chemicals business, and a pretax provision of \$53.7 relating to restructuring costs.
- (k) As a result of the losses incurred for the years ended December 31, 1995 and 1994, Grace was unable to fully cover the indicated fixed charges.

W.R. GRACE & CO. FINANCIAL DATA SCHEDULE PERIOD ENDING JUNE 30, 1997

1,000

```
6-MOS
         DEC-31-1997
            JAN-01-1997
              JUN-30-1997
                         43,200
                        0
                 635,100
                        0
            1,259,400
3,293,500
              1,478,900
              4,312,300
       1,144,900
                             0
               0
                         0
                          800
                    421,100
4,312,300
                     1,627,600
            1,651,200
                         993,200
                 993,200
                    Ó
                    0
             39,400
               263,400
                   99,600
          163,800
                      0
                     0
                           0
                  163,800
                    2.15
                    2.15
```

Amount shown is net of allowances. Included within current assets and total assets are net assets of discontinued operations of \$56,800. Includes a pretax gain of \$103,100 (\$63,000 after-tax) on the sale of Grace's specialty polymer business, and a pretax provision of \$12,400 (\$8,000 after-tax) relating to restructuring costs.